04 University of Plymouth Research Theses

01 Research Theses Main Collection

2013

A comparison of carbon sequestration potential and photosynthetic efficiency in evergreen and deciduous oaks growing in contrasting environments in the Southwest UK

Carne, Demelza Jane

http://hdl.handle.net/10026.1/1598

http://dx.doi.org/10.24382/4556 University of Plymouth

All content in PEARL is protected by copyright law. Author manuscripts are made available in accordance with publisher policies. Please cite only the published version using the details provided on the item record or document. In the absence of an open licence (e.g. Creative Commons), permissions for further reuse of content should be sought from the publisher or author.

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognize that its copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without the author's prior consent.

A comparison of carbon sequestration potential and photosynthetic efficiency in evergreen and deciduous oaks growing in contrasting environments in the Southwest UK.

Demelza Carne.

Department of Bio-Medical and Biological Sciences.

University of Plymouth

A thesis submitted to the University of Plymouth in partial fulfillment of the requirements for the degree of:

D.Phil.

December 31st 2012.

Abstract

A comparison of carbon sequestration potential and photosynthetic efficiency in evergreen and deciduous oaks growing in contrasting environments in the Southwest UK.

Demelza Carne.

Global climate change is predicted to alter the weather patterns around the world, as climatic zones shift, forest carbon sequestration projects (e.g. the UK woodland carbon code) need to take into account the specific requirements of planted species. In the UK, oaks are an important charismatic group of trees favoured in recent planting programmes. The English oak (*Quercus robur* L.), has poor water conservation, but is a major component of natural forests in lowland UK. On the other hand, Holm oak (*Quercus ilex* L.) is a Mediterranean oak that has high water conservation and can also tolerate cold despite being restricted by minimum temperatures. At local scales, there may be advantages of planting either evergreen or deciduous oak species for forestry and climate mitigation. Alternatively, a comparative assessment of non native versus native productivity, may give clues to the invasiveness potential of Holm oak and its ability to out compete the deciduous oak along an urban to upland gradient. This thesis documents a series of field based experiments intended to analyse differences in carbon sequestration potential and photosynthetic efficiency between these two species and in relation to their environment within the Southwest UK.

520 one year saplings were planted, half in pots and half in nursery field beds situated on Dartmoor, the east Devon Dartmoor fringe, Totnes, and Plymouth city centre. Originally two sites were chosen for their relative 'urban' qualities, two at 'rural' localities, one upland and a control site with access to a polytunnel for comparisons with well-watered and non nutrient limited trees. However, data analyses showed that sapling characteristics were site specific with the five sites falling along an urban, rural to upland gradient. The field experiments included monthly height and diameters (ground level diameter or DAG), monthly assimilation rates and analysis of chlorophyll fluorescence to aid interpretation of photosystem II functioning and sapling 'vitality'. Further laboratory experiments analysed specific leaf area (*SLA*), mass based leaf Nitrogen (N_{leaf}) and carbon (C_{leaf}), with differences between sun and shade leaves included, to aid comparisons between species and sites. The final experiment was a destructive harvest and this was used to find total biomass estimates and carbon allocation to different root shoot fractions.

In order to quantify differences between saplings and adult trees a smaller experiment was conducted in the canopy using experienced climbers and leaf level productivity analysed. Intrinsic water use (*iWUE*), stomatal conductance (*Gs*), means net assimilation rates (A_n) and chlorophyll fluorescence parameters; Variable fluorescence over maximum fluorescence (F_v/F_m) and performance index (*PI*) were measured and relative carbon assimilation rates and productivity assessed and compared between species at one urban, rural and upland site.

Results showed that Q. ilex allocated relatively more carbon to branches and leaves as a sapling

which in turn increased growth rate and whole tree assimilation rates to larger values than the deciduous oak despite Q. *robur* being able to increase maximum assimilation rates in response to increasing temperatures. This gives Q. *ilex* the advantage and overall biomass was higher at all sites than Q. *robur* apart from the upland site where there were no differences in biomass accumulation between species. However, despite no significant difference in biomass at this site Q. *robur* had greater survival and photosystem II functioning. In mature trees Q. *ilex* was under stress and N_{leaf} and carbon sequestration potential were higher in the deciduous species at the urban site. In contrast, Q. *robur* was under stress at the upland site at Castle Drogo where thin and nutrient poor soils have made it more vulnerable to drought stress. Here, mature Q. *ilex* showed reduced photosynthetic efficiency in relation to cold and drought, but was able to recover when milder temperatures occurred. The results were site specific, with a reduction in both *SLA* and relative allocation to the leaf weight fraction (*LWF*) in *Q.robur* the only common urban related effect seen.

The potential for *Q. ilex* to perform well at sapling stage is due to its morphological plasticity and drought tolerance. This species may become more prevalent within the Southwest as local climates continue to push it northwards from its natural Mediterranean range. In contrast, if *Q. robur* continues to suffer from defoliation and fungal attack and this may leave it more vulnerable to competition throughout less fertile and drier areas of its natural range.

Contents

Al	Abstract v			
Ao	eknov	vledgements	xvii	
Aı	ithor ⁹	's declaration	xix	
1	Intr	oduction.	1	
	1.1	Climate change.	1	
	1.2	Climate and Tree Growth	3	
	1.3	Species and Environment.	12	
	1.4	Aims and Objectives:	12	
	1.5	Overview of Chapters	13	
2	Gen	eral Methodology.	17	
	2.1	Local Climate	17	
	2.2	Site Descriptions.	18	
	2.3	Sampling protocol.	21	
3	Res	ults of Environmental Parameters.	33	
	3.1	Soils.	33	
	3.2	Climate Statistics	38	
	3.3	Nitrogen Dioxide (NO_2)	40	
4	Pho	tosynthesis and Chlorophyll fluorescence.	43	
	4.1	Introduction	44	
	4.2	Respiration.	48	
	4.3	Light Saturation Curves.	48	
	4.4	Diurnal Rhythms	50	
	4.5	Species and Environment	54	
	4.6	Discussion.	66	
	4.7	Conclusion.	68	

5	Leaf	Traits.	71
	5.1	Introduction	71
	5.2	Methodology	74
	5.3	Results	75
	5.4	Discussion.	87
	5.5	Conclusion.	90
6	Proc	luctivity and Biomass Accumulation.	91
	6.1	Introduction.	92
	6.2	Methods.	97
	6.3	Results	98
	6.4	Discussion.	107
	6.5	Conclusions and limitations.	113
7	Mat	ure Trees.	115
	7.1	Introduction.	115
	7.2	Methods.	118
	7.3	Results	123
	7.4	Discussion.	129
8	Gen	eral Discussion.	143
	8.1	Comparisons with the literature.	147
	8.2	Conclusion.	151
	8.3	Limitations of this research	152
	8.4	Further Work.	153
A	Soils	Schedules	155
	A.1	TOTAL N	155
	A.2	MINERAL N	157
	A.3	AVAILABLE PHOSPHATES	158
	A.4	POTASSIUM	160
B	Resi	dual Plot Information	163
	B.1	Biomass data: Residual Information	163

С	Inte	raction Plots.	169
	C.1	Final Above Ground and Total Biomass and fractions Interaction Plots	169
D	Leaf	Trait Interactions with Factors	183
	D.1	Box and Whiskerplots for Leaf traits against factors	183
E	Bion	nass: Descriptive statistics and extra graphs	189
	E.1	Actual growth rates.	189
	E.2	Relative growth rates (RGR)	195
	E.3	Biomass descriptive statistics	201
F	Scat	tergraphs	205
	F.1	Scatter graphs for biomass and stem volume relationships	205
G	Info	rmation on Light levels and filter densities for Light Response Curves	213
Н	Mat	ure trees: extra graphs.	215
Gl	Glossary.		221
In	dex.		225
Lis	st of r	references.	225

List of Figures

1.1	Natural distribution maps for a) <i>Q. ilex</i> , and b) <i>Q. robur</i>	10
2.1	Location of study sites for saplings	18
2.2	Example of sampling design, site shown is rural DT	20
3.1	Soil moisture, pH, total organic carbon (TOC) and total nitrogen (TN) for each site	36
3.2	Soil phosphorous (P), potassium (K) and mineral Nitrogen (MN) found in soils from each site	37
3.3	Climate statistics for generic rural, urban and upland sites	38
3.4	Climate statistics for individual sites	39
3.5	Annual monthly means for NO_2	41
4.1	Daily means (± 95 % confidence intervals for a) mean dark respiration rate and b) mean daylight assimilation rate.	49
4.2	Light Response Curves for <i>Q. ilex</i> and <i>Q. robur</i> a) shade, b) sun and c) <i>Q. ilex</i> April 2011	51
4.3	Light Response Curves for <i>Q. ilex</i> , February 2011	52
4.4	Natural diurnal rhythms.	52
4.5	Diurnal rhythms.	53
4.6	a) Monthly assimilation rate (A_n) and b) Monthly stomatal conductance (Gs) for well watered control site saplings.	56
4.7	a) Monthly variable fluorescence over maximum fluorescence ratio (F_v/F_m) and b) Monthly performance index (<i>PI</i>) for well watered control site saplings	57
4.8	Differences in photosynthetic parameters between drought and well watered saplings	58
4.9	Species differences in a) Mean net carbon assimilation rate A_n and b) Mean stomatal conductance (<i>Gs</i>) for each site and season	63
4.10	Species and seasonal differences in a) variable fluorescence over maximum fluorescence ratio F_v/F_m and b) performance index <i>PI</i> for each site	64
4.11	Species and site specific differences in Upper and Lower canopy leaves	65
4.12	Species and environment differences in Upper and Lower canopy leaves	65

5.1	Interaction plot to highlight site specific nature of environmental effects, SLA	78
5.2	Interaction plot to highlight site specific nature of environmental effects, C_{leaf} .	79
5.3	Interaction plot to highlight site specific nature of environmental effects, N_{leaf} .	80
5.4	Species differences in soil treatment (field grown and potted) and effect on SLA.	82
5.5	Species differences in soil treatment (field grown and potted) effect on C_{leaf}	83
5.6	Species differences in soil treatment (field grown and potted) effects on N_{leaf} .	84
5.7	SLA: SUN and SHADE effects	85
5.8	C_{leaf} : SUN and SHADE effects	85
5.9	N_{leaf} : SUN and SHADE effects	86
6.1	Actual Growth	102
6.2	Total relative growth rate (RGR) over 33 month growth period	103
6.3	Environment and site differences in total biomass between species (means \pm 95 % CI)	104
7.1	Photosynthetic rate, effects of sun shade and canopy position	134
7.2	Stomatal conductance (Gs), effects of sun shade and canopy position. \ldots .	135
7.3	Intrinsic water use efficiency (iWUE), effects of sun shade and canopy position.	136
7.4	Variable over maximum fluorescence ratio (F_v/F_m) , effects of sun shade and canopy position.	137
7.5	Performance index (PI), effects of sun shade and canopy position	138
7.6	Canopy SLA: Sun and Shade differences	139
7.7	Canopy C_{leaf} , effects of sun shade and canopy position	140
7.8	Canopy leaf nitrogen (N_{leaf}) , Differences between species site and effects of sun shade and canopy position.	141
B.1	Residual plot for leaf fraction minus control site.	164
B.2	Residual plot for root weight fraction.	165
B.3	Residual plot for stem weight fraction	166
B.4	Residual plot for branch weight fraction	167
C.1	Interaction plot for biomass fractions; pots and field grown sapling differences.	170
C.2	Interaction plot for AGB: Species and condition	171
C.3	Interaction plot for AGB: SITE and CONDITION.	172
C.4	Interaction plot for TB:Species and Condition	173

C.5	Full interaction plot: TB	174
C.6	SPECIES:SITE interactions for Biomass Fraction	175
C.7	Interaction plot for AGB: SPECIES and SITE	176
C.8	Interaction plot for TB: SPECIES, SITE and CONDITION	177
C.9	Interaction plot: BWFt	178
C.10	Interaction plot: LWFt	179
C.11	Interaction plot: RWFt	180
C.12	Interaction plot: SWFt	181
D.1	Leaf carbon data for species, site and environment	184
D.4	Leaf N. Differences in upper and lower leaves	185
D.5	Leaf Carbon. Differences in upper and lower leaves	186
D.6	SLA. Differences in upper and lower leaves	187
E.1	Monthly growth for control site saplings over the 33 month growth period	189
E.2	Monthly growth for upland site (DPF) saplings over the 33 month growth period.	190
E.3	Monthly growth for rural site B saplings over the 33 month growth period	191
E.4	Monthly growth for rural site DT saplings over the 33 month growth period	192
E.5	Monthly growth for urban site D saplings over the 33 month growth period	193
E.6	Monthly growth for urban site L saplings over the 33 month growth period	194
E.7	Relative growth rates in control site potted saplings	195
E.8	Relative growth rates in upland site field grown saplings	196
E.9	Relative growth rates in rural site B, field grown saplings	197
E.10	Relative growth rates in rural site DT, field grown saplings	198
E.11	Relative growth rates in urban site D, field grown saplings	199
E.12	Relative growth rates in urban site L field grown saplings	200
F.1	Relationship between total biomass <i>TB</i> and stem volume HD^2 at $harvest_1 \ldots T^2$	206
F.2	Relationship between total biomass TB and stem volume HD^2	207
F.3	Relationship between above ground biomass AGB and stem volume HD^2	208
F.4	Relationship between total biomass TB and stem volume HD^2	209
F.5	Scatter: Relationship between above ground biomass AGB and stem volume HD^2	210
F.6	Scatter: RWFt and LWFt relationship.	211

H.1	Canopy iWUE; Differences between dates (means \pm 95 % CI)	216
H.2	Canopy Gs, Differences between dates	217
H.3	Canopy A_n , Differences between dates	218
H.4	Canopy F_v/F_m , Differences between dates	219
H.5	Canopy <i>PI</i> , Differences between dates	220

List of Tables

3.1	Soil and NO_2 averages in 0-20 cm of soil $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	33
3.2	Nitrogen dioxide (NO_2 levels)	41
4.1	ANOVA table for Photosynthetic parameters in drought and well watered saplings	55
4.2	ANOVA table; Differences between potted (P) and field grown (G) saplings	60
4.3	ANOVA table: Site * Leaf Position (field grown saplings)	66
5.1	Descriptive Statistics for leaf traits	77
5.2	ANOVA table: Mixed linear model for SPECIES and ENVIRONMENT (upland, rural and urban) interactions with leaf traits.	81
5.3	ANOVA table: Effect of soil treatment (POTS vs GROUND) on leaf traits	81
5.4	ANOVA: Site and species differences in sun and shade leaves.	87
6.1	Percentage of saplings lost from each factor level	99
6.2	Descriptive statistics for initial growth variables	100
6.3	ANOVA table: Site, species and soil treatment effects on total biomass	103
6.4	ANOVA table for biomass allocation to different fractions	105
6.5	Initial harvest regression: table of coefficients	107
6.6	Allometric equations: Total biomass TB & AGB for Control site and groups.	108
6.7	Allometric equations: TB & AGB for Upland data.	109
6.8	Allometric equations: AGB & TB for Rural data.	110
6.9	Allometric equations: AGB & TB for Urban data	111
7.1	Canopy site climatic information.	119
7.2	Descriptive statistics for mature tree leaf level physiological parameters	125
7.3	Mature tree canopy leaf traits: Descriptive statistics	126
7.4	ANOVA table: Physiological differences between dates.	126
7.5	ANOVA: Canopy SUN and SHADE	128
7.6	ANOVA: Canopy SLA, SUN and SHADE effects	128
7.7	Canopy leaf Carbon and Nitrogen	128
7.8	Descriptive statistics for canopy biomass.	129

E.1	Descriptive statistics for final biomass and growth variables, Ground	202
E.2	Descriptive statistics for final biomass and growth variables, POTS	203

Acknowledgements

I started this for Dad but finished it for you Mum, LOVE YOU LOADS.

I would like to thank my supervisors for all their advice over the past four years, in particular my Director of Studies, Dr Maria Donkin for her quiet patience and advice on writing up this thesis. Thanks also to Dr Steve Burchett and the Canopy research volunteers; Chris Harris, Valerie Maldeney, Sven Bakte, Matt Perry, Sophie, Rebecca and Adam Day, but especially Kiki O'Malley who I couldn't have completed the canopy work without.

Thank you to the undergraduate volunteers who helped with field work and the tedious biomass experiment, in particular Samatha Jones and Catherine Horrell. A very special thank you to Sarah Peake, who not only helped with field work but also carried out the second CN analyses for me when I was up to my ears in trees! Big thanks also to Mum, Debbie, Sharon, Norma, John, Nick, Sally, Geraldine and Rupert who helped pick leaves during the biomass experiment, I definitely couldn't have completed counting all those thousands of leaves without you.

Thanks to Moor Trees, a Devon based woodland charity and partner with Southwest OPAL (url: www.moortrees.org) for use of their tree nursery sites and to Dave Sharp at 'Digginit', a community organic allotment scheme, in Plymouth (url: www.digginit.org.uk). Paul Freeman, farm manager and volunteers at Dartmoor Prison Farm and Erica Rochester (private allotment owner) for use of allotment space for this project. Tom Wood, National Trust Ranger at Castle Drogo; Nick Butcher, Mt Edgcumbe Ranger and Gareth Harrison from Plymouth City Council for permission to use the mature tree sites.

This project was funded by the OPen Air Laboratories network, a nationwide lottery funded organisation dedicated to bringing nature to the wider community (url: www.opalexplorenature.org) and I thank them for the opportunity to conduct this research.

Most of all I am indebted to my amazing family, Frodo, Sam and Beren, for their tireless love and support.

Authors declaration

At no time during the registration for the degree of Doctor of Philosophy has the author been registered for any other University award.

Scientific seminars and conferences attended: The British Ecological Society; Forests and Global Change Symposium, University of Cambridge 28-30th March 2011. See below for conferences attended at which posters and presentations were given.

Signed:

Date: _____

Publications :

Posters and conference presentations :

2011:

SEB Joint Meeting with Biochemical Society and British Ecological Society Stress responses - molecules, organisms and environments *Charles Darwin House, London*. Poster presentation: **Photosynthesis and Efficiency in Evergreen and Deciduous oaks growing in Southwest UK.**

2010:

Climate Change Conference. *University College, Falmouth*. Oral Presentation. Carbon Sequestration Potential in Evergreen and Deciduous Oaks.

2009

SEB Annual Main Meeting 2009 Scottish Exhibition and Conference Centre (SECC), Glasgow Poster presentation: Carbon Sequestration Potential in Evergreen and Deciduous oaks.

Word count for the main body of this thesis: 42,750

Chapter 1

Introduction.

1.1 Climate change.

The last 14 years have included the eleven warmest years since 1850 (Hegerl et al. 2007) with climate forcing dominated by long-lived greenhouse gases (*GHG*). The global increase in *GHG* emissions was around 70%, between 1970 and 2004 leading to an average increase in temperature of 0.74 °C between 1906 and 2005 (Hegerl et al. 2007). Carbon dioxide CO_2 is the most important anthropogenic *GHG*; pre-industrial levels of atmospheric CO_2 have risen from 280 ppm to the 2005 average of 379 ppm (Solomon et al. 2007). In 2004 CO_2 was responsible for 77%, of all *GHG* emissions in 2004 (IPCC 2007) and data from the Earth system Laboratory shows recorded globally averaged monthly means for CO_2 of approximately 395 ppm in January 2013 (over marine surface sites) (Dlugokencky and Tans 2013).

More importantly, the global rate of growth of CO_2 emissions has increased from 0.92 gigatonnes atmospheric Carbon per year (GtC) between 1995-2004 compared to the previously recorded rate of 0.43 GtC CO_2 per year between 1970 and 1994 (IPCC 2007).

As a result of excessive *GHG* in the earth's atmosphere the next 20 years could see an increase in temperature of $0.2 \,^{\circ}$ C per decade accompanied by changes in precipitation patterns around the world (Hegerl et al. 2007).

In the northern hemisphere, northern Europe is likely to experience an increase in winter rainfall and increased flash floods while southern Europe will be subjected to increasingly high temperatures and drought (Hegerl et al. 2007). In fact these changes are already documented such as increased drought in the Mediterranean (Solomon et al. 2009).

1.1.1 Climate change in the UK

All areas of the UK are predicted to experience reduced summer precipitation, an increase in summer and winter mean temperatures and an increase in winter precipitation under medium and high emission scenarios (UKCP 2009). In fact observed changes already show that between 1961 and 2006 winter rainfall increased in all regions while summer rainfall decreased (UKCP 2009). Minimum temperatures have increased for all regions of the UK and daily mean temperature has increased between 1.05 °C (Scotland) and 1.6 °C in London and the Southeast (UKCP 2009). Daily winter mean temperatures have shown the greatest increase of 2 °C in London and the Southeast, while Southwest UK has experienced an increase in winter mean temperatures of 1.7 °C, and annual mean daily temperature increase of 1.4 °C (UKCP 2009).

The Southwest UK is predicted to experience further increases in winter mean temperatures of 1.3 °C under the medium emissions scenario and increases in summer temperature of 1.6 °C, although annual mean precipitation is not predicted to change under this scenario there is likely to be an increase in winter precipitation of 7% and a decrease in summer precipitation of 8% under high emission scenario (UKCP 2009). However by 2080s these percentage increases will become more significant; precipitation levels increasing three-fold during winter and decreasing by a similar percentage during summer (24%) (UKCP 2009).

1.1.2 Climate change and forestry.

Leading on from the 1992 Earth Summit, the subsequent 'Kyoto Protocol'¹ 1998, imposed emission targets for industrialised countries to reduce their outputs of several important *GHG*; CO_2 , Nitrous oxide N_2O , methane CH_4 , and the fluorinated gases; hydro fluorocarbons (HFCs), perfluorocarbons (PFCs) and sulphur hexafluoride (SF 6) by 2008-2012; of these CO_2 is considered the most important (Alcamo et al. 1998).

Article 3.3 of the protocol allows industrialised countries (known as 'Annex B' countries such as the UK) to use afforestation, reforestation and land management strategies as mitigation tools (IPCC 2000; Brown 2002). Although it is also stressed that such tools should only be used in

¹the Kyoto protocol became legally binding in 2005

conjunction with emission reducing tactics (van Minnen et al. 2006, 2008).

The seasonal flux of CO_2 as forests absorb CO_2 during the summer and release it over winter months, indicates the light and temperature response of vegetation in the absorption of atmospheric carbon (*C*) (Lorenz and Lal 2010). Deforestation emits 17.4% *GHG* annually and is the third largest emitter of *GHG* globally after energy use and industry (Betts 2000), having released 1.6 GtC annually during the last decade (Broadmeadow and Matthews 2003).

Despite such loses from deforestation and respiration, globally forests are a net sink (Jarvis and Linder 2007) and are estimated to remove around 2.1-3 GtC annually (Broadmeadow and Matthews 2003) nearly 10% of total terrestrial sink (Gough et al. 2008; Hegerl et al. 2007). The vast majority of carbon sequestered is stored as biomass in trees (90%) (Nowak and Crane 2002; Korner 2003). Watson et al. (2000) suggest that globally forests have the potential to sequester an extra 87 GtC by 2050, while UK forests are estimated to be sequestering carbon at a rate of 2.5 million tonnes of carbon annually (Cannell 2003) and utilise around 12% of UK landcover (Broadmeadow and Matthews 2003).

1.2 Climate and Tree Growth.

Tree growth is closely linked to climate, with tree distributions closely following climatic averages such as mean winter temperatures mean summer maximum temperatures and rainfall patterns. The effect of changing climates at regional level are already causing range shifts in broadleaved tree species such as beech (Jump et al. 2006). Evidence also suggests, evergreen broadleaved species are expanding their northern range as average minimum temperatures increase across central Europe (Berger et al. 2007). Historically tree species have expanded and contracted their range in response to changes in climate, in particular oaks have expanded from glacial refuges since the Quaternary era (Davis and Shaw 2001). Temperature and precipitation at the end of the previous growing season is important for tree growth, [e.g. for Beech (Kern and Popa 2007) , and oak (Drobyshev et al. 2008)] and changes in normal patterns at this time will exacerbate species ability to adapt to drought during the summer months.

1.2.1 Carbon Assimilation by Trees.

Trees assimilate carbon through photosynthesis and utilise 'dead' tissue to form part of their support structure (Franco 2008). The ability of trees to store carbon in such a way makes carbon sequestration by forests an important mitigation tool while other more permanent CO_2 sinks and new energies are developed (Korner 2003).

Carbon accounts for around 40% of a plants dry mass (Lambers et al. 2008) with The gross productivity of a forest ecosystem can be described as the sum of net primary productivity and autotrophic respiration as follows:- see equation 1.1 (in its simplest form).

$$GPP = NPP + Ra \tag{1.1}$$

Where GPP = Gross Primary production, NPP = Net Primary Production and Ra = autotrophic respiration ($gC m^{-2} yr^{-1}$) (DeLucia et al. 2007; Lindroth et al. 2008). However, disturbances that cause carbon release such as fire, harvest/management and pollution/transport effects should also be taken into account when measuring carbon balance in forests (Kennedy et al. 2008).

Photosynthesis

The photosynthetic apparatus of a tree is the leaf; during photosynthesis leaves convert light [Photosynthetically Active Radiation (P.A.R.)] into chemical energy and assimilate CO_2 from the atmosphere in order to synthesis carbohydrates (Scott 2008). The following equation gives the basic reaction of photosynthesis whereby CO_2 reacts with water and light energy (photons) to give carbohydrates and oxygen (O_2).

$$nCO_2 + 2nH_2O + \text{ light energy} \rightarrow (CH_2O)n + nO_2 + nH_2O$$
 (1.2)

Where *n* is the number of molecules of each given element in order to produce the general structure of a carbohydrate. The most commonly used equation is that of glucose formation where n = 6.

CO₂ enters the leaf via tiny pores known as stomata. Stomata release water vapour and can control photosynthesis as they react to different environmental factors (Lorenz and Lal 2010). Leaf photosynthetic rate can therefore depend on many interacting variables such as the absorption of light (Nobel 2005), leaf age (Harper 1989), temperature (Filella et al. 1998), water availability and climatic extremes (e.g. Boisvenue and Running 2006; Waring et al. 2008), soil nutrients (e.g. Bergh et al. 1999) and the adaptability of species to their environment (Catovsky et al. 2002; Becker et al. 1994).

Water stress is a large factor in controlling CO_2 assimilation and changes in rainfall patterns at regional levels will affect carbon fluxes and assimilation (Franco 2008; Heimann and Reichstein 2008). Primary productivity is therefore limited by water availability (Amthor 1999). Increased temperatures will exacerbate water stress as transpiration rates are expected to increase (Heimann and Reichstein 2008). However, stomatal closure in response to increasing CO_2 and water stress mean that the overall effect may depend on soil water holding capacity and drought sensitivity of plants (Heimann and Reichstein 2008).

Respiration.

Respiration releases part of the CO₂ absorbed by the leaf in order to produce energy and carbon skeletons for growth (Cannell and Thornley 2000; Lambers et al. 2008; Gratani et al. 2008; Amthor 1995). Respiration is therefore an important component when calculating carbon sequestration and can be expressed as a ratio between NPP and GPP. The literature gives average respiration rates between 33 and 60% of carbon assimilated across a number of different environmental conditions (Clark et al. 2001; DeLucia et al. 2007; Litton et al. 2007; Gratani et al. 2008). Losses have even been estimated as high as 80% of total carbon assimilated (Amthor 2000). Respiration rates are affected by a number of factors including temperature, elevated CO_2 levels and severe water stress but can also change with species (Amthor 2000; Heimann and Reichstein 2008) giving an indication of the difficulties involved in measuring carbon sequestration rates in forest ecosystems (Luyssaert et al. 2007).

The ratio between NPP and Ra tends to increase with latitude while GPP remains relatively constant across latitudes (Valentini et al. 2000). Increased temperatures caused increased net

hourly CO_2 uptake in Boreal black spruce stand, however as temperatures rose above $20^{\circ}C$ total CO_2 uptake declined (Grant et al. 2008). Despite GPP increases with rising temperatures optimum temperature for CO_2 assimilation in north temperate forests can be as low as $10^{\circ}C$ (Luyssaert et al. 2007).

1.2.2 Species Choice and Productivity.

When deciding on which species to plant for carbon sequestration, the productivity of a species and its suitability to the environment in which it will be planted is important. At smaller scales carbon fluxes are more variable and total carbon pools depend on carbon allocation to roots, leaves and stems. Allocation and partitioning of carbon also depends on the availability of water, distribution and availability of resources, temperature and management regimes (Luyssaert et al. 2007) and can also depend on pollution stress. It has been predicted that trees in Northern Europe will become more productive with climate change, with an increased allocation of carbon to foliage (Alcamo et al. 2007). Monitoring sequestration potential at stand level using biomass estimates from height and diameter allometry are key methods in forest management decision and important tools for monitoring changes in productivity (Boisvenue and Running 2006).

In the Mediterranean, typical tree species are likely to change due to increasing temperatures and drought whereas in the UK some native species such as *Q. robur* (English Oak) may become less productive in southerly and lowland areas (Boisvenue and Running 2006). Willoughby et al. (2007) assessed the suitability of a number of species for planting farm woodlands at several different lowland sites in the UK, and concluded that species choice is important because of changing climate. Further, limited land availability brings with it a need to utilise a range of both urban and rural environments (Willoughby et al. 2007). As major lowland sites become hotter and drier and rainfall patterns change, the potential of non-native species to become more productive and therefore more efficient at sequestering carbon in these areas is a key area for research. However, the dangers involved in planting exotic species should also be taken into account with many non natives considered invasive (Rackham 2008).

1.2.3 Evergreen and Deciduous leaf habit.

The dominance of deciduous forests during a warm polar past was thought to be an adaptive strategy against high respiration rates during the warm polar winter (Osborne and Beerling 2003). However, Royer et al. (2003) found that warmer winters along with high atmospheric CO_2 in a simulated ancient polar environment did not increase respiration costs in deciduous species. However, compared with evergreen respiration costs for the same environment carbon lost through leaf litter gave evergreens the advantage thus reducing overall carbon sequestration in deciduous species by nearly half that of the evergreens in the ancient polar winter (Royer et al. 2003). However, despite the winter carbon costs discussed, and that deciduous species have less time in which to assimilate carbon over the growing season, they experience boosted photosynthetic rates towards the end of the summer and beginning of autumn (August - mid October) this leads to increased annual carbon uptake to similar levels as that of evergreen trees, despite leaf losses (Royer et al. 2005). More recently, research has shown that the deciduous species Q. *rubra* experienced greater assimilation of CO_2 to biomass but also had double the photosynthetic capacity than that of the evergreen hemlock in the same woodland (Catovsky and Bazzaz 2000; Catovsky et al. 2002).

The amount of carbon allocated to biomass in forests is around 50% of total biomass (Bateman and Lovett 2000; Litton et al. 2007). Recent information shows that UK forest ecosystems are storing an estimated 790 million tonnes of atmospheric carbon (MtC) with UK forests removing approximately 15 MtC annually in 2007 (Read et al. 2009). Over a period of five years (1997-2001) in the UK, an Oak woodland removed approximately 15 tonnes of CO_2 per hectare per year ($tC ha^{-1} yr^{-1}$), compared with 24 $tC ha^{-1}yr^{-1}$ in a Sitka Spruce stand (Read et al. 2009) although the lifetime of an oak tree is longer than Sitka spruce and brings other benefits such as increased biodiversity. Recent past UK plantation forestry has relied on non native conifers such as Sitka spruce, further, woodland planting schemes have been concentrated on upland sites and other environmentally sensitive areas, although planting is now restricted to below 240 m above sea level (masl) (Cannell and Dewar 1995). Further to this as the age structure of UK woodlands change, carbon sequestration in these woodlands is likely to decline as low as 4.6 $MtCO_2 yr^{-1}$ by 2020 (Read et al. 2009) despite the Forestry Commission grant schemes for planting native broad-leaves and restoring native woodlands available since 1995 (Cannell and Dewar 1995). Therefore it is important that the carbon sequestration potential of individual species is measured and in relation to a variety of environmental conditions. Carbon sequestration projects should also be aware of the effectiveness of both prudent site and species choice not only in order to provide accurate information for evaluating the UK carbon pools, but also in order to promote the effective use of the license given by the United Nations Land Use, Land Use Change and Forestry (LULUCF) advice for afforestation and reforestation practices (IPCC 2000).

Carbon Nitrogen Ratios, Specific Leaf Area SLA.

The carbon allocated to the living part of a tree remains relatively constant after canopy closure (Franco and Kelly 1998; Franco 2008). Therefore, leaf area and mass are important components in measuring productivity, and specific leaf area (SLA) is a good representation of biomass/carbon sequestration (e.g. Lindroth et al. 2008) although climate induces variation in SLA (McCarthy et al. 2006). Larger trees are able to assimilate more CO₂ due to the relatively large leaf area compared to that of smaller trees (Gratini and Varone 2006) but photosynthetic efficiency of leaves is also important. In order to measure photosynthetic efficiency and carbon sequestration potential in species all these factors therefore need to be taken into account. Species with evergreen or long term leaf habits require greater carbon allocation to cell structure, resulting in a higher mass per unit leaf area (LMA) and lower leaf nitrogen per unit area (N_{leaf}), in turn leading to lower specific leaf area and lower gas exchange rates per unit area (Reich et al. 1999). Therefore photosynthetic nitrogen use efficiency (PNUE) is lower for evergreen species; Takashima et al. (2004), studying evergreen and deciduous oak trees, found a high correlation between photosynthetic capacity (A_{max}) and leaf nitrogen concentration (N), finding that evergreen broadleaf oaks tended to have low specific leaf area (SLA) and lower photosynthetic capacity per unit leaf N than deciduous oaks due to increased allocation of photosynthetic N to cell walls as a mechanical aid to leaf longevity (Takashima et al. 2004). In contrast, deciduous leaf habits tend to allocate greater N to photosynthetic apparatus and a higher SLA in turn

leading to greater gas exchange rates and lower root costs during winter (Takashima et al. 2004). However, this advantage is lessened by the susceptibility of deciduous leaves to herbivory and the environment (Reich et al. 1999). A review by Givnish (2002) showed longer photosynthetic season, lower construction costs, lower costs for nutrient replacement, and tougher laminae (thus better forest resistance against drought and herbivory) of evergreen leaves to be an advantage. In contrast, previous authors (e.g. Miller and Stoner 1979; Reich et al. 1999) have shown increased construction costs in evergreens in the field. Villar et al. (2006) found that construction costs were not significantly different in contrasting leaf habits but these costs differed with phylogeny, suggesting tradeoff between defense mechanisms and growth. In perennial species, typical of seasonal environments, current leaf growth is assimilated from the previous year's photosynthate (Harper 1989). Trees belonging to the genus Quercus spp. are 'ring porous'; where a ring of large pores develops early during annual growth; this porous ring is then followed by a ring of smaller pores later in the annual cycle (Morecroft et al. 2003). As a result new wood is formed around 2-3 weeks before budburst which in turn reduces the available resources for leaf development (Morecroft et al. 2003). Spring flush in European oaks also relies on the mobilisation of stored reserves from the previous year's assimilation (Both and Bruggemann 2009).

1.2.4 Evergreen Vs Deciduous Oak for Carbon Sequestration in UK.

The distribution of *Quercus spp.* is linked to water availability and soil fertility (Breda et al. 1993). In the UK there are two native oaks, *Q. petrea* (Matt) Liebl. (Sessile oak) and *Q. robur* (Pedunculate oak). Pedunculate oak preferring more fertile soils (Breda et al. 1993; Becker et al. 1994) and more tolerant of water logged soils than the Sessile oak (Morecroft et al. 2009). The Mediterranean evergreen, *Q. ilex* (Holm oak) is widely planted in urban and parkland areas and has become naturalised along the South coast of England (Peterken 2001). *Q. ilex* prefers sandy limestone soils, tolerating more clayey soils when average rainfall is more than 300mm (Barbero et al. 1992). Although in its natural range, *Q. ilex* sub species *ilex* is found in more coastal locations while the sub species *rotundfolia* is found inland, it has been suggested that *Q. ilex* represents one complex species and grows in semiarid, sub-humid, humid and per-humid

(*a*) Natural distribution map for *Q. ilex* (Figure 1.a has been removed due to Copyright restrictions)



(b) Natural distribution map for Q. robur

Figure 1.1: Natural distribution maps for a) *Q. ilex* ((Removed due to Copyright restrictions please see text for description of *Q. ilex* natural range, original figure from (Welk et al. 2013)) and b) *Q. robur* (blue area shows distribution, map from (EUFORGEN 2009).

climates but prefers the sub-humid climate where it can occur in a wide range of temperatures and soils (Barbero et al. 1992). In colder climates however competition from deciduous oaks and conifer species mean *Q. ilex* restricted to areas where competition is less such as rocky outcrops and poorer soils (Barbero et al. 1992) (see Figure 1.1).

Q. robur is native to the UK although its natural range extends throughout Europe, SW Asia and North Africa (Kleinschmidt 1993). Although *Q. robur* is thought to have suffered unfavourably in the changing climate of the last 150 years compared to *Q. petrea* (Epron and Dreyer 1993*b*; Becker et al. 1994; Beckett et al. 2000) and is considered drought sensitive (Dickson and Tomlinson 1996). However, there is evidence that *Q. robur* is becoming more productive in the upland regions of the UK due to changing regional climate (Chavana-Bryant 2006 unpublished masters thesis).

Q. *ilex* subsp. *ilex* is a Mediterranean schlerophyllus species and in its Mediterranean range tends to be restricted by water availability and winter temperatures, it has been shown that Q. *ilex* has a higher tolerance to winter freezing than to summer drought (Ogaya and Penuelas 2003). However, in contrast, others have shown that Q. *ilex* was limited more by winter freezing than by summer drought (Nardini et al. 2000). More recently Gimeno et al. (2009) showed how populations of Q. *ilex* were able to tolerate climatic extremes throughout their Iberian range, without intrinsic adaptation. Using leaf area (LAI) as a surrogate for shoot biomass Crescente et al. (2002) found a 20% reduction in photosynthetic efficiency and lower relative growth rates for Q. *ilex* during low temperatures in part of its native range in the region of Lake Garda, Italy. Net photosynthetic capacity and stomatal conductance was also reduced in Q. *ilex* subjected to severe drought, closing stomata as a defence mechanism for water stress (Filella et al. 1998). In terms of range shifts, it has been widely discussed that Q. *ilex* is becoming restricted in driers areas of its native range and that wetter sites are likely to become more productive for this species in the future (e.g. Sabaté et al. 2002).

1.3 Species and Environment.

1.3.1 Urban Forestry.

The urban environment presents challenges to plant growth. Air pollution combines with poor soils, and limitations in water availability due to restricted rooting depths, impermeable surfaces and building cover.

Plant responses to air pollution involve stomatal regulation and increases in drought sensitivity (Chappelka and Freer-Smith 1995; Calatayud et al. 2011). Increased ozone exposure can lead to stomatal dysfunction and subsequent drought sensitivity (Chappelka and Freer-Smith 1995). Increased stomatal regulation and blockage can lead to reduced photosynthesis with leaf age and reduced transpiration rates and eventually premature leaf senescence (De Nicola et al. 2011). In contrast, nitrogen dioxide (NO₂) pollution has been shown to be utilised by plants and broken down into products used in amino acid formation (Nowak and Crane 2002), thus increasing carbon assimilation in some instances.

Urban environments tend to have higher CO_2 concentrations and higher mean temperatures accompanied by a smaller diurnal temperature range (Searle et al. 2012). Trees then have the potential to sequester more carbon and can also have a cooling effect on urban environments by providing shade, while transpiration can cool air under canopies (Gratani and Varone 2006). Both deciduous and evergreen oak contributed to decreasing air temperatures in Rome through shade and transpiration effects, however, deciduous oaks had a greater effect than *Q. ilex* probably due to the larger crown area (Gratani and Varone 2006). However the same study found that *Q. ilex* had greater overall *C* sequestration due to its longer photosynthetic season.

Planting trees in urban environments is also beneficial to health and social well-being of urban dwelling humans thus having indirect affects on a country's carbon use (e.g., see Westphal 2003).

1.4 Aims and Objectives:

The overall aim of this research was to test the hypothesis; The native hardwood *Quercus robur* L. has greater carbon sequestration potential and photosynthetic efficiency than the Mediter-

ranean species *Q. ilex* L., in different rural, urban and upland environments in the Southwest UK. Experiments were conducted on one to three year saplings : Differences in height and diameter growth, relative growth rate, biomass, leaf level carbon assimilation and photosynthetic efficiency and leaf traits were assessed. In order to aid scaling to older trees, a smaller experiment was conducted on mature trees: photosynthetic rates, photosynthetic efficiency and leaf traits were measured in the canopies of mature trees at rural, urban and an upland location.

- 1. General working hypotheses
 - 1.1. There is a significant difference between *Q. ilex* and *Q. robur* in productivity and carbon sequestration potential in Southwest UK.
 - 1.2. Overall productivity, total biomass and carbon content is greater in *Q. robur* than *Q. ilex* and there are differences in these parameters between the urban, rural and upland environments.
 - 1.3. There are differences in assimilation and photosynthetic efficiency in young saplings compared to mature trees.

Further analyses on the differences between sun and shade leaves of each species and their position in the canopy were incorporated into the final discussion on productivity/carbon sequestration potential of each species in relation to their environment.

1.5 Overview of Chapters

Chapter 1 Chapter one is the introduction and gives a brief overview of anthropogenic climate change and tree growth. It then moves on to discuss the relative benefits of evergreen and deciduous leaf habits in broadleaved trees in productivity and carbon assimilation. There is also a short section discussing planting in urban environments.

Chapter 2 Chapter two is the general methods section. Where the seedling sites are introduced. Growth, photosynthesis, chlorophyll fluorescence and light response curves, biomass and environmental methodologies are also introduced. **Chapter 3** This chapter covers the measurements of the environmental parameters. Soil nutrients, moisture and pH; Nitrogen dioxide diffusion tube analyses; Climate statistics shown as monthly means for each year, 2009 - 2011.

Chapter 4 Chapter four covers direct measurements of photosynthetic rate. Experiment one gives light response curves for each species followed by an indication of both light saturated diurnal rhythms (using a dimorphic light and a steady photosynthetically active radiation (PAR) of 500-600 μ mols $m^{-2} s^{-1}$) and diurnal rhythms using natural light.

Night time respiration rates have been recorded and used to complete light response curves and as an aid to interpretation of carbon assimilation in chapter 3.

There then follows a drought experiment which gives species response to experimental drought and this is again used to interpret effects of each site environment on photosynthetic parameters. Results from the ongoing recording of sapling photosynthesis and chlorophyll fluorescence give monthly differences between species and sites and data from the control site gives annual fluctuations of each species in well watered nutrient rich environment.

Chapter 5 Chapter five analyses leaf traits in saplings of both species. N_{leaf} , C_{leaf} and SLA. N_{leaf} is closely correlated with maximum assimilation rates and can also be altered by environment through changes in light. In some cases urban pollution can cause N deposition to leaf and N accumulates, this can be used in biochemical reactions and subsequently reused in carboxylation process (see chapter for relevant references). *SLA* also varies with environment as does (although to a lesser extent) C_{leaf} . These traits can then be used to aid the prediction of productivity in relation to environment for each species and this is discussed.

Chapter 6 Chapter six is concerned with growth rates of the saplings and relative growth rates using height and diameter data

Specific leaf area and carbon nitrogen ratios are presented and finally results from the biomass study where 200 three year old saplings were destroyed and split into root, stem, branch and foliage fractions, dried and weighed for biomass.

Allometric equations were determined using total biomass (TB), above ground biomass (AGB), height (H) and diameter at ground level (DAG) data and differences between species and sites discussed.

Chapter 7 Chapter seven presents the results of the mature tree experiment includes measurement of photosynthetic rates, chlorophyll fluorescence parameters, Specific leaf area and carbon nitrogen relationships. Ontogenetic differences are left for discussion in the final chapter.

Chapter 8 Chapter eight gives the final discussion on differences in carbon assimilation and sequestration potential of the two species and compares sapling and mature tree data and discusses issues of scaling for saplings to mature trees.
Chapter 2

General Methodology.

2.1 Local Climate.

The general climate of Southwest England is described as oceanic (warm temperate, fully humid with warm summers and cool winters) (Kottek et al. 2006). Rainfall is variable throughout the region. Mean annual rainfall (1961-1990) was around 800-1000mm per annum for lowland Plymouth area (urban sites) increasing to between 1000-1200 for South Brent/Totnes area (two rural sites) and 1800-2000 for the High Moor at Princetown (third upland site) (Phillips and Denning 2007; Phillips and Mcgregor 2002). The changing climate affects precipitation and temperatures and more recent seasonal rainfall trends show Plymouth has become 50 mm a year wetter during winter and 30 mm drier during summer over the past 150 years along with a significant increase in winter average air temperatures (p < 0.05) for Plymouth (following national trends) (Wood 2004). Soils of all sites, except for the plots at Dartington (DT) and Dartmoor Prison Farm (DPF), are similarly described using the Cranfield University 'Soilscape' online tool (National Soil Resources Institute 2009); free draining, slightly acidic, base rich and with low fertility. For DT, soils are described as free draining, slightly acidic but with high fertility. Soils around the plot at DPF are described as free draining, acid and loamy soils over rock with areas of slowly permeable, wet and very acid upland soils with a peaty surface. However each site and specific plot differs in past soil management and further soil characterisation has been completed for each site (see results section for detailed, site specific assessment of soil characteristics, rainfall, and air temperatures).



Figure 2.1: Location of Study Sites: Urban sites, Saltash Rd; Laira (L) private allotment on Norfolk Rd, Laira; Digginit (D) community garden, Stoke. Rural sites, Broadley (B) 'Moortrees' allotment, near South Brent; Dartington (D), 'Moortrees' allotment near Totnes. Upland site, Dartmoor Prison Farm (DPF), Princetown. Control site, Millbrook SE Cornwall where saplings were kept in Polytunnel (Digimap 2009)

2.2 Site Descriptions.

2.2.1 Upland Site.

Dartmoor Prison Farm (DPF).

HMP Dartmoor Farm, Princetown, Yelverton, Devon (SX585740) is situated on the high moor at an altitude of approximately 400m.a.s.l. Soils are shallow upland soils and the study plot itself has a peaty top soil added. Overall acidity is pH 5 and the topsoil contains high levels of nitrogen but with relatively low mineral nitrogen.

2.2.2 Rural sites.

Broadley (B).

Moor-Trees, Broadley Nursery (B) is situated in a valley, approximately 80 m.a.s.l near Totnes on the moorland edge (SX 727 553). The nursery is the fenced corner of a field containing semipermanent grassland. The study plot at B is situated near the bottom of a shallow north-west facing slope.

Dartington (DT).

Moor Trees Hunters Moon Nursery (DT), Dartington, Totnes (SX 791 614) is approximately 30 m.a.s.l, situated above the A485, on a gentle north-north-east facing slope. The land is old agricultural land belonging to the Sharpham Estate, and once contained chickens. Past land use affects the fertility of this site and there are many 'bones' found throughout the upper soil layer.

2.2.3 Urban sites.

Digginit (D).

The community allotment at Digginit (D), Raynham road, Plymouth (SX467555) is approximately 40 m.a.s.l and is situated behind (<0.75 km) two of Plymouth Cities busiest roads (B3396 and A386). The tree saplings are planted near the top of an east-facing slope with a low mineral N content, pH 6.4 and TOC of around 5.3%.

Laira (L).

Norfolk Close council allotments near Old Laira Rd (L) (SX 502 559) is approximately 70 m.a.s.l and are situated at the top of a steep east south east facing slope approximately 60 m.a.s.l. Soils have an average soil pH of 6, a low TOC and nitrogen content.

Control Site (CS).

The control site (CS) is situated in Millbrook, SE Cornwall (SX428525). CS is a lowland rural site less than 10 m.a.s.l, flat, sheltered but with exposure to sunlight during the day. The control site also has access to a poly-tunnel during Autumn, Winter and Spring months to reduce affect of minimum temperatures (although tree saplings were taken out of poly-tunnel during peak summer 2009 due to high maximum temperatures they remained inside the polytunnel for subsequent summers due to rooting. Tree saplings at CS are potted (P); there are no planted (G) saplings and this site was mainly used as a reference site for saplings that had been watered and kept in favourable conditions, the control site was also used for respiration, light response curves and diurnal rhythm experiments.



Figure 2.2: Sampling design showing saplings grown in blocks of twenty saplings of each species in pots (P) and field grown (G) (site shown is rural DT)

2.3 Sampling protocol.

One year old saplings of the UK *Quercus robur* L. (Pedunculate oak) were obtained from the tree planting charity 'MoorTrees' ¹ as bare root stock (provenance: Hembury wood, classified as western Atlantic oakwood and situated on the SE Dartmoor fringe). The Mediterranean *Q. ilex* sub-species *ilex* (Holm oak) was purchased in one litre vermiculite/peat based compost as one year out door hardened seedlings (provenance unknown: located from SE UK, Sussex) (nomenclature for both species follows (Stace 1997)). Saplings were planted over the winter, between December 2008 and January 2009, when growth was expected to be at a minimum.

Forty saplings of each species were planted at each site in a three-factor block design, where SPECIES, SITE and SOIL CONDITION were factors and soil condition had two levels: field grown (G) and potted (P).

Each soil condition (treatment) contained 20 saplings. Twenty saplings of each species positioned side by side in 10 litre (L) pots and 20 of each species planted in the ground at 30-40cm spacing (the same distance as the 10L pots standing side by side). This was the same for all sites apart from the control site CS.

The Control site (CS) had 20 of each species in 10L pots and no planted saplings due to both space within the poly-tunnel and economic constraints.

The pots contained a similar compost/soil mix with an even distribution of slow release fertilizer (Osmocote Exact Low Start 16-18 month). Soils at all planted sites were characterised according to Cranfield University's 'Soil-Scape' and laboratory analysed for moisture content, pH and nutritional status.

Due to issues with rooting and subsequent comparisons with field grown saplings, the potted saplings were not periodically re-arranged to alleviate competitive differences in growth between interior and exterior saplings.

¹Moortrees is based in South Brent, Devon (URL: http://www.moortrees.org/

2.3.1 Sampling Protocol for Soils.

Soils at each site were sampled using a systematic sampling technique (as suggested for level II soil analysis for monitoring the effects of air pollution on forests (United Nations 2006). A soil auger was used to remove 0-10cm depth of soil from 10 places within the 2x3m plots; the soil was then mixed thoroughly, bagged and labeled for analysis. This procedure was then repeated at another depth of 10-20cm in order to give an idea of the changes in available soil nutrients through the soil profile.

Soils were then prepared for laboratory use by splitting each mixed site sample into two. Half of the soil for each site was air-dried, sieved and ground in a 2mm sieve. The other half of the soils were kept moist for analysis of water content, soil pH. and determination of total nitrogen (TN) by the Keijdhal method (Tan 2005).

Soils were tested for total nitrogen (TN), soluble inorganic Nitrates (mineral Nitrogen) analysed as (N); available potassium (AK) using exchangeable cations in soil; available phosphate (AP) (using Brays No. 1 extractant), pH using a pH meter and di-ionised water, total organic carbon (TOC) using gravimetric method, and $\% H_2 O$ in wet soil, $\% H_2 O$ in dry soil.

2.3.2 Laboratory Methods for Soils.

Soil Moisture Content.

To give a direct indication of water content of the soils at each site, the wet mass percentage of water lost was determined using the 'gravimetric' method.

Soil moisture content was found by weighing xg of fresh, moist soil in a weighed foil wrap and then drying in an oven at 70°C for 24 hrs repeatedly until two consecutive dry weights were the same [methodology follows Tan (2005)]. The following equations show how a percentage for soil moisture content was found. Eqn. 2.1 gives the amount of actual water H_2O lost during the drying process while Eqns. 2.2 and 2.3 give an indication of the H_2O content of the soil expressed as either percentage of dry mass or wet mass. For the purpose of this experiment only wet mass % H_2O was determined using eqn. 2.3.

$$H_2O \ lost = weight \ of \ moist \ soil(g) - weight \ of \ oven \ dry \ soil(g)$$
(2.1)

$$Dry mass\% H_2 O = oven dry mass - wet mass/mass oven dry soil(g) \times 100\%$$
 (2.2)

Wet mass%
$$H_2O = (wet mass - oven dry mass)/mass oven dry soil(g) \times 100\%$$
 (2.3)

(Tan 2005).

Soil Acidity/Alkalinity (pH).

pH. is determined by the Log_{10} hydrogen ion H^+ concentration in soil solution and was determined on soils direct from the field using a pH meter and de-ionized water as a reagent. pH determines nutrient availability in the soil, therefore pH was measured before determining the method of extraction for N, P and K.

Total Organic Carbon (TOC).

Total Organic Carbon (TOC) was determined using the dry combustion method. TOC was measured as an indication of organic *C* present: none of the soils were alkaline or were formed from a calcareous base rock, therefore inorganic carbon was assumed to be insignificant. Approximately 2g of oven dry soils were weighed and placed into the Skalar Primacs SLC Analyser where CO_2 is measured, by 'Non Dispersive Infra Red Detection (NDIR) for Total Carbon, at a temperature of 1043°C for 6 minutes per sample. At this temperature the carbon is completely oxidized to CO_2 in the presence of a catalyst.

Nitrogen.

Nitrogen is found in soils in both organic and inorganic forms and is an essential nutrient for plant growth chlorophyll and protein formation (Tan 2005).

Plants take nitrogen from the inorganic source while organic nitrogen is present in soil as a nutritional reserve, released after decomposition and mineralisation of organic matter (Tan 2005). Organic N is closely associated with soil organic matter with only 2-3% of organic N mineralised and released in inorganic form (Tan 2005).

Total N (TN).

TN was determined by the wet oxidation method (Kjeidhal) due to this methods applicability to samples low in N (see appendix A for full schedules on nitrogen extraction methods).

Mineral N (MN).

Mineral or Inorganic MN is composed of Ammonia $NH_3 - N$, ammonium $NH_4 - N$, Nitrates $NO_3 - N$ and Nitrites $NO_2 - N$, these types of inorganic N are continuously formed by decomposition and mineralisation of organic matter via the nitrogen cycle (Tan 2005). Most of them are soluble in water and their concentration in soil depends usually on the application of fertilizers (Tan 2005). MN was analysed using soils moist from the field and potassium chloride used to extract the soluble and exchangeable nitrate and ammonium ions (see Appendix A for detailed schedules).

Available Phosphorus (Phosphates).

Methodology for determining available phosphorus depends on the pH of soil. For the purpose of this research the Bray 1 method for pH <6 soils was used. The Bray 1 method is used for the determination of available P (phosphorous soluble in dilute Hydrochloric acid (HCl) and ammonium fluoride (NH_4F) and the extractant is designed to remove acid-soluble P and a portion of the phosphates of Al and Fe.

Available Potassium (AK) (Measuring Exchangeable Cations).

AK was analysed as exchangeable K^+ ions, using acetone as a reagent $[OC(CH_3)_2]$.

For full schedules detailing all soil analyses see appendix A.

2.3.3 Sampling Protocol for Growth, Biomass and Measurements.

Growth (Height and Diameter).

Height, measured from root collar to tip of central stem (H) and mean diameter at ground level (DAG) was measured for all saplings. Height was measured using a meter rule attached to a wooden disk with a piece cut away in order to reduce chances of error due to different volunteers sampling techniques. Diameter was taken using a metal dial caliper accurate to 0.02 mm and was taken 1cm from ground level (if saplings had root collar above ground level then DAG measurements were taken above the root collar) and again at right angle to the original measurement. Measurements were taken every four weeks between planting and harvest, August 2011.

Leaf Traits.

On one day in August 2009, one single mature, current year leaf was taken from the upper canopy from 10 trees at each factor and treatment level (no distinction between upper and lower canopy at this stage in growth).

This experiment was repeated at the end of August 2010 and again in June 2010. However this time either two or three leaves were taken from each of the 10 trees per treatment. One current year mature leaf (CYM) from the upper canopy and one CYM from the lower and if any immature (NL) leaves existed then one of these was also taken from the upper canopy.

During the initial leaf collection (2009), each leaf was drawn around using graph paper (2, 10 and 20 mm squares ruled). Length (L) from petiole to tip and width (W) at widest point were recorded and leaves divided into easily measurable shapes (see appendix A) then leaves placed on a leaf area machine, (Li-Cor) calibrated with a $25cm^2$ square accurate to $24.7cm^2$, and their true leaf area determined. Subsequent years, equations formed from initial leaf area estimates were used to find leaf area (LA).

Leaves were weighed, dried at 60°C, and reweighed. Differences in specific leaf area (SLA), leaf mass per area (LMA) were looked for between sun (upper) and shade (lower) leaves in two, three and four year old saplings. Mature current year leaves (CYL) from 2009 and 2010

were also analysed for carbon and nitrogen ratios using the Elemental Analyser EA1110 that measures carbon C, Nitrogen N and Hydrogen H percentages per 2mg of sample. Carbon and nitrogen were then analysed individually as percentages and as a C/N ratio.

Biomass and total leaf area.

On 22nd September 2009, eleven *Q. robur* and seven *Q.ilex* saplings taken from rural site B were subjected to destructive harvesting for biomass and total leaf area calculations. The saplings were cut into root, stem, branch and leaf fractions. Each fraction had initial weights taken, and were then oven dried at 60° C until two consecutive weights the same were recorded (after approximately one week).

Before weighing and oven-drying the leaf fraction, leaves were divided into mature CYL and immature NL and after initial weighing total leaf area calculated by a LiCor Leaf area machine (calibrated to 24.8cm²).

After drying, oven dry weights were calculated for each of the fractions and total biomass (TB), above ground biomass (AGB), total leaf area (TLA) and SLA were calculated to enable calibration of allometric equations using biomass, leaf area, and height and diameter data.

Allometric Equation for use with growth parameters and biomass relationships.

The literature contains a huge amount of information on allometric equations for *Quercus* species, however many of these data are only valid for mature trees with diameters 7cm or greater (Ravindranath and Ostwald 2008).

An equation in the form D_2H has been used for sapling biomass studies in various forms (Williams and McClenahen 1984 e.g.), in particular is the use of ground level diameter (DAG) as a substitute for DBH in sapling based allometry. Williams and McClenahen (1984) found it to be useful in biomass estimates for White and Black Oaks from eastern America using common logarithms. Eqn. 2.4, using natural logs to homogenise variances, was used to determine allometric biomass equations for saplings in this study.

$$Log_e Y = b_0 + b_1 Log_e(D_2 H) \tag{2.4}$$

(Williams and McClenahen 1984).

Final harvest began the last week in August 2011 where approximately 10 saplings from each site and condition were destroyed and split into the same fractions as the previous biomass experiment. New allometric equations were prepared using the same basic equation 2.4. Carbon content was also estimated from dry weights of each fraction and allometric equations derived for each species and site.

Difficulties in removing whole root systems without loss with the additional difficulty of cleaning soil stones from roots mean that often below biomass estimates are removed from final analyses. However in this instance, roots were dug out from field grown saplings to a depth of 50 cm and radius from stem base of 25cm in an attempt to reduce the error across samples. Some fine root fraction was lost during cleaning due to time involved in sieving through soil debris, however, the general aim was to retrieve 75+% of fine root fraction contained in soil waste. It is accepted that fine roots below 1mm in diameter are under sampled (Poorter et al. 2012) and total biomass TB was analysed with this error in mind.

For saplings that had over 600 leaves, total leaf area was determined in the final harvest using the individual leaf areas of the first 100-200 leaves from upper, middle and lower canopy of each tree and averages used to estimate total leaf area of each tree.

2.3.4 Sampling Protocol for Environmental Parameters.

In January 2009 rain gauges were set up at two rural sites; Broadley and Dartington and monthly averages worked out from the data collected. Rainfall data for Dartmoor prison farm and Plympton (as a reference rainfall for the South Devon sites) was received monthly (as daily averages) from Environment Agency (EA) and data between January 2009 and October 2009 were converted to monthly averages.

Rain data for Urban sites was taken from the University of Plymouth 'Metnet' weather station archive (Metnet 2009).

Temperature and relative humidity (RH) were also recorded at all sites using Thermochron 'Ibuttons' (Homechip Ltd). Thermochrons recorded hourly temperature and humidity data from

two ibuttons per site.

All the climate data collected between January 2009 and August 2011 was converted into monthly averages and converted into a climate graph (see Chapter 3 for yearly/monthly climate averages at each site).

2.3.5 Sampling Protocol for Pollution Parameters.

Nitrogen dioxide (NO_2) diffusion tubes (Gradko Environmental) were placed, one at each site, 1.5*m* from the ground as close to the research plot as possible (<10m). Tubes were sent back to Gradko Environmental Herts for analysis of monthly NO_2 levels (measured in μgNO_2). Overall average monthly NO_2 was worked out from monthly data collected between June 2009 and June 2010 (although some months were missing).

2.3.6 Sampling protocol for gas exchange and chlorophyll fluorescence.

Gas Exchange.

Gas exchange parameters were assessed throughout the sampling period using a LCi portable Infrared-Gas-Analyser (IRGA) (ADC BioScientific, Herts), accompanied with a 'broadleaved' leaf chamber.

Flow-rate was set at 200 μ mol s⁻¹ and stabilisation of sub-stomatal CO₂ was used as a guide to settling time for analysis, this taking about 20-40 seconds.

Parameters used were; mean net photosynthetic rate $(A_n) \ \mu mol \ m^{-2} \ s^{-1}$, mean stomatal conductance of CO_2 (Gs) mol $m^{-2} \ s^{-1}$ and mean transpiration rate (E) mmol $m^{-2} \ s^{-1}$.

Monthly *A_n*.

The IRGA was used with a Light-Unit attached (ADC Scientific, Herts), this was used to alleviate the effects of variation in light levels during sampling. Unless used for light saturation experiments, the light unit was used at approximately 490 - 520 *P.A.R.* for consistency in data across sites.

To obtain monthly averages for each site A_n measurements were taken at each of the seven sites from 10 trees in each species/condition. Five mature leaves were chosen per tree on one day each month (this was subject to weather therefore for some months data is missing). If there was a distinction between upper and lower canopy (which in some cases and mostly for *Q. robur* this was not obvious) then two upper and two lower mature leafs were measured.

 A_n data was downloaded and used to ascertain differences in A_n across sites, species and conditions and over time using a general linear model (R 2009). The relationship of A_n to biomass was also considered.

Light Response Curves.

In order to assess the light saturation point for each species light saturation curves were produced in the middle of February and again in April 2011 for *Q. ilex* only, and for both species in August 2010. (see results section, chapter 4 for curves).

The ADC Light-unit was used to measure A_n at a number of different light levels using three neutral density (ND) filters (optical densities, 0.1, 0.2, and 0.3) (ADC Bioscientific, Herts). The ND filters were used in combination allowing 79.4% (0.1), 63.1% (0.2) and 50.1% (0.3), 39.8% (0.1 + 0.3), 31.6% (0.2 + 0.3), 25% (0.1 + 0.2 + 0.3) of light to be transmitted onto the leaf for each combination.

Each light saturation experiment was conducted at the control site CS.

Three leaves from two healthy *Q. robur* saplings and two healthy *Q. ilex* saplings were measured at eight different light levels using the ADC light unit and natural density filters (see appendix G for specific light levels and densities used). The LCi IRGA was configured for use with a light unit at 12.5v (Hfactor = 168, TRW = 0.9).

(See results section 'Light Saturation Curves').

Diurnal Rhythms.

Diurnal rhythms for *Q. ilex* and *Q. robur* were accumulated over two days during August 2009 and 2010 (see chapter 4).

Four leafs (2 x upper, 2 x lower unless no distinction made) from each of five *Q. robur* and five *Q. ilex* trees were measured between 5:00 am and 16:30 pm. Each set of measurements took

roughly 20 minutes and was repeated every 30 minutes throughout the day.

This experiment was repeated during June 2010 using the light-unit set at 500-550 PAR (see chapter 4 on photosynthesis).

Chlorophyll Fluorescence.

Chlorophyll fluorescence was measured using the Pocket PEA (Hansatech Instruments Ltd, Norfolk).

At each site, four leaves from each of ten trees, per species and treatment were sampled.

Leaves were first covered with leaf clips and dark-adapted for 30 minutes to ensure full dark adaption. Each leaf was then subjected to a bright flash of light at $3500molss^{-1}m^{-2}$. The pocket PEA records a number of parameters over a period of a few seconds; F₀ (which is the value of fluorescence at time 0 just after full dark adaption), F_m which is the value of fluorescence at saturating light (maximum excitation), and F_v which is variable fluorescence and indicates maximum capacity for photochemical quenching, and is calculated by $F_m - F_0$ (Cavender-Bares and Bazzaz 2004).

Each of the measured parameters has meaning but the most commonly used chlorophyll fluorescence parameter used within plant research is the variable fluorescence maximum fluorescence ratio (F_v/F_m) . The ratio F_v/F_m was then used to indicate the maximum quantum efficiency of Photosystem II. F_v/F_m is a sensitive indication of plant photosynthetic performance, with 0.85 being the maximum obtained in a healthy plant, lower values are an indication of stress (Hansatech Instruments Ltd 2010).

The other parameter of interest is the performance Index (*PI*) which uses three stages of the fluorescence process to give an indication of plant vitality.

Stage one can be described in terms absorbed excitation energy; The fraction of reaction center chlorophyll molecules (excited) relative to the total number of excited chlorophyll molecules. Stage two is the maximum yield of primary photochemistry (F_v/F_m) and finally stage three which is the probability that a trapped exciton moves an electron into the electron transport chain (ETC) beyond the primary electron acceptor (Q_a^-) (Strasser et al. 2000). PI is described

by Yunus et al. (2000) as an internal force of the sample to resist external constraints. PI is more sensitive to stress than F_v/F_m (Hermans et al. 2003) and was used in the analysis to aid interpretation of plant health.

Data from the beginning and end of the sample period was analysed using general linear models (GLM), as for growth parameters and A_n and any interval plots or barcharts shown include means \pm 95 % confidecne intervals (CI).

Chapter 3

Results of Environmental Parameters.

Table 3.1: Soil characterisation showing; moisture content, total organic carbon (TOC), total Nitrogen (TN), mineral N (MN), available potassium (AK) and available phosphorous (P) of soils, and atmospheric NO_2 at each site (Upland site: Dartmoor Prison Farm (DPF), Rural sites Broadley (B) and Dartington (DT), Urban sites: Plymouth city allotments, Digginit(D) and Laira(L))

		1	=							
ENVIRONMENT	SITE	pН	wet mass H_20	TOC	TN	MN	Р	K	NO_2	
			%		•	$g kg^{-1}$		cmolc $Kg^{-1} K^+ *$	$\mu g m^3$	
UPLAND	DPF	4.77	34.39	8.12	2.79	0.01	0.15	0.59	0.16	
RURAL	В	5.63	29.35	5.05	2.95	0.02	0.04	0.96	0.18	
	DT	5.18	28.64	5.04	3.17	0.01	0.12	1.43	0.27	
URBAN	D	6.37	23.74	6.88	3.33	0.02	0.21	1.18	0.61	
	L	5.99	19.45	3.60	1.93	0.01	0.08	0.76	0.76	
ALL POTS	POTS	6.41	49.65	13.20	7.52	0.07	0.04	4.32	**	
* centimoles of exchangeable K cations per kg soil										

Soil characterisation and atmospheric NO₂ concentrations

** all pots including control site pots

3.1 Soils.

In general the pots had the highest soil moisture content and TOC. Macronutrients were also high with pots having highest TN, AP and AK present. This was expected as potted plants were potted using both soil and compost mix with added slow release fertilizer. Despite this the peaty top-layer found at the DPF site on the high moor also has relatively higher macronutrients and organic matter content, however the other sites although being allotment sites have had no compost added for this particular research which shows in the relatively low levels of TOC and macronutrients present (Table 3.1).

3.1.1 Soil Moisture.

Using equation 2.3, the wet mass percentage (%) of water was found (Figure 3.1a). As expected pots and upper layer of Dartmoor Prison Farm (DPF) had the highest wet mass % (49.7% & 47.5% respectively), due to the organic matter content present. The driest sites were the urban sites, with all sites and all layers under 30% water content. Laira, had the lowest overall water content of 19.5% (average of both layers) (see Figure 3.1a).

3.1.2 Soil pH.

All sites had slightly acidic soils as suggested by the Cranfield University Soil-scape viewer. DT had the highest acidity at pH 4.25 in the upper layer and pH 5.28 in the lower layer (Figure. 4) DPF was also acidic with lower layer at pH 4.81. The rural site B and the urban sites at D and L were slightly less acidic with upper layers of around pH 6. Overall acidity was greatest at the rural sites DT (pH 4.8), DPF (pH 5.2), B (pH 5.6), and slightly higher at the urban sites FM (pH 5.7), L (pH pH6) and D (pH 6.4) (see Figure 4 for pH info on separate layers) (see Figure 3.1b).

3.1.3 Total Organic Carbon Content (TOC).

TOC was greatest in pots (13.2%) (as expected). The top layer of soil at DPF contained higher amounts of TOC (12.47%) while all other sites were relatively lower (Figure 4). In all cases the 10-20cm soil layer contained much lower levels of TOC this was especially true for the two rural sites B and DT and one urban site at L. The % TOC was less than 4% for all rural sites in this layer only 2.94% for L. The urban sites D and FM had over 5% TOC which is around the average % for the top layer of a loam soil high in fertility (Tan 2005) (see Figure 3.1c).

3.1.4 Nitrogen.

Total Nitrogen (TN).

TN levels were highest in pots (7.52 $g kg^{-1}$) (as expected) with the urban site Digginit D and all the rural sites containing between 3.4 and 3.6 $g kg^{-1}$ of TN, only the urban sites at FM and L having less than 2 $g kg^{-1}$ of TN in upper layer of soil (see Figure 3.1d). All lower layers had smaller amounts of TN (generally less than 2 $g kg^{-1}$). Figure 3.1d shows the relationship between TOC and TN (TN relies on the presence of organic matter in order to be released) (see Figure 3.1d).

3.1.5 Mineral N (MN).

MN is usually around 2-3% of organic N (Tan 2005). However for all sites mineral N was low (less than 1% for topsoil and less than 0.5% for lower layers). Again pots had the highest mineral N however DPF and DT were the lowest (0.2% and 0.4% respectively) (see Figure 3.2c for soil *N* shown as a percentage of *TN*).

3.1.6 Available P (Bray 1 Extraction).

Phosphorous was actually low in the pots compared to the upper layer at DPF, D and DT. Lower layers were also high in P at the latter two sites perhaps a result of past management (see Figure 3.2a).

3.1.7 Available pottasium K (AK).

AK was highest in pots and between the planted sites top layer of soil contained the higher AK fraction, especially in the peaty topsoil of DPF (1.8 *mol* $kg^{-1} K^+$). Overall AK was low across soil layers and sites (see Figure 3.2b).



Figure 3.1: Soil moisture, pH, total organic carbon (TOC) and total nitrogen (TN) at each site (Upland site: Dartmoor Prison Farm (DPF), Rural sites Broadley (B) and Dartington (DT), Urban sites: Plymouth city allotments, Digginit(D) and Laira(L)), samples taken from 0-10cm and 10-20cm soil depths.



(c) Mineral N as a % of TN

Figure 3.2: Soil phosphorous (P), potassium (K) and mineral Nitrogen (MN) found in soil samples at 0-10cm and 10-20cm soil depths at each site (Upland site: Dartmoor Prison Farm (DPF), Rural sites Broadley (B) and Dartington (DT), Urban sites: Plymouth city allotments, Digginit(D) and Laira(L)).



Figure 3.3: Climate statistics for generic rural site (Mount Batten Weather Station, Plymouth), urban site (University of Plymouth, Metnet) and the Upland site (Environment Agency rain gauge and independent ibutton temperature loggers): bars show mean monthly temperatures and lines monthly rainfall totals for years 2009-2011

3.2 Climate Statistics.

Figures 3.3 and 3.4 give mean temperatures and rainfall totals for each month in 2009-2010. There are climate statistics for each site complied from data loggers and rain gauges on site and also reference data for urban sites from the University of Plymouth's 'Metnet' weather station and the Met office data from Mountbatten station (situated as a reference for rural sites).

Mean temperatures were lower in 2010 than the other two years by about $1^{\circ}C$ for all sites, with the upland site DPF having the lowest temperatures of $8^{\circ}C$ in 2010 and $10^{\circ}C$. the rural site Broadley, situated on the east Dartmoor fringe, had mean annual temperatures $1^{\circ}C$ warmer





(b) Dartington



Figure 3.4: Climate statistics for individual sites (Upland site: Dartmoor Prison Farm (DPF), Rural sites Broadley (B) and Dartington (DT), Urban sites: Plymouth city allotments, Digginit(D) and Laira(L)); bars show mean monthly temperatures and lines show monthly rainfall totals for years 2009-2011

in line with the Mt Batten Weather station with all other sites warmer on average by $2^{\circ}C$. Mean temperatures were highest in June and July with rural site DT having the highest summer temperatures (maximum monthly mean 18.8°C), followed by the urban sites, L (18.2°C) and D ($17.5^{\circ}C$). Lowest temperatures occurred between December to February the lowest being at the upland site DPF (0.51 °C) rural B (2.9 °C) and rural DT (3.4 °C) while the urban winter temperatures were between $5-6^{\circ}C$ for these months (evidence for the urban heat island effect increasing winter averages at the urban sites, despite summer averages being higher at rural DT. Rainfall was lowest at the rural site B as this site falls within the east Devon 'rain shadow' discussed by (Phillips and Mcgregor 2002; Phillips and Denning 2007), with less than 100mm of rain falling each month during 2009 and 2010 and annual totals of 500, 442 and 730 respectively for each year of the experimental period. Although upland DPF had higher overall rainfall during each year (2238, 1806 and 1090 respectively), the rural site DT seemed to have the most rainfall during July 2009 (over 800mm) and yearly totals of (2500, 1150 and 916 respective years) almost twice as much as the rural site B (see Figure 3.4 for more information). Rainfall data for the urban site L has been taken from the University Metnet data and yearly totals for this site were (980, 840 and 720) respective years.

3.3 Nitrogen Dioxide (NO_2).

The air quality objectives set out via government regulations through "Local Air Quality Management (LAQM)" and in relation to the Environment Act 1995 (Environmental Protection and Monitoring Unit 2009) give a target annual mean of 40 $\mu g m^3 NO_2$. Stoke Village has been highlighted within the report as a key area of concern for NO_2 pollution and is situated in front of the 'Digginit' urban nursery site. Tavistock road, the main route into the centre from Stoke village is also a key pollution area and the both urban mature tree sites were situated near to this road (in particular the *Q. robur* site. Tavistock road diffusion analysis by Plymouth City Council regularly exceeds the AQO of 40 $\mu g m^3$, background concentrations which should be indicative of the nursery sites were between 16.9 and 17.8 in Plymouth (2008 data) (Environmental Protection and Monitoring Unit 2009). A biomass adjustment factor (BAF) was used to correct the results, using the local adjustment factor created by Plymouth City Council from *Table 3.2:* Nitrogen dioxide (*NO*₂) levels recorded via diffusion tubes at each site and sites combined into environmental means (Upland environment: Dartmoor Prison Farm (DPF), Rural environment: Broadley (B) and Dartington (DT), Urban environment: Plymouth city allotments, Digginit(D) and Laira(L)).

		Monthly NO ₂ per SITE ($\mu g^{-1} m^2 NO_2$)							
Environment		URBAN			CONTROL	RURAL		UPLAND	
Date	Site	FM	L	D	CS	DT	В	DPF	
Jun/Jul		0.72	0.79	0.72	0.32	0.4	0.23	0.17	
Jul/Aug		0.57	0.69	0.45	0.12	0.21	0.15	0.14	
Aug/Sep		0.63	0.65	0.6	0.2	0.21	0.18	0.16	
Oct/Nov		0.27	0.9	0.67	0.22	0.19	0.2	0.14	
Annual monthly mean		0.55	0.76	0.61	0.22	0.25	0.19	0.15	
Environment mean			0.64		0.22	0.1	22	0.15	

2008 data, of 0.92 $\mu g m^3$.



Figure 3.5: Annual monthly means for *NO*₂ diffusion tubes at each site (Upland site: Dartmoor Prison Farm (DPF), Rural sites Broadley (B) and Dartington (DT), Urban sites: Plymouth city allotments, Digginit(D) and Laira(L)).

Chapter 4

Photosynthesis and Chlorophyll fluorescence.

Field based photosynthesis measurements are useful tools for analysing productivity in particular with reference to carbon sequestration potential in trees. This chapter evaluates mean assimilation rates for both species at each of the sapling sites, stomatal conductance (Gs) and intrinsic water use efficiency (iWUE) as the ratio of net photosynthetic rate (A_n) and stomatal conductance to water vapour (Gs) and photosynthetic efficiency, using variable fluorescence over maximum fluorescence (F_v/F_m) as a measure of maximum efficiency of photosystem II, photochemical quenching and a newer performance index' (PI) which incorporates three phases of the photosynthetic process and is a measure of plant vitality. Parameters were measured on ten trees per species at each urban, rural and upland site, from two leaves in the upper canopy and two leaves in the lower canopy to establish if there were differences between sun and shade leaves. Measurements were taken between 8 am and 12 pm during each monthly sampling date. Light response curves were completed for each species and diurnal photosynthesis was also measured (methods described in the general methodology chapter 2. In control site saplings, net assimilation rates were higher in the deciduous oak, however the evergreen was able to increase assimilations rates in early Spring when temperatures were cool but above 5°C despite lowered efficiency. Effects of environment were site specific; at the upland site, Q.ilex experienced lower assimilation rates and reduced F_v/F_m during winter while in both species assimilation was high during the summer at this site but PI was lower throughout the year. The evergreen oak maintained mean assimilation rates of around 7 \pm 0.4 μ mol s⁻¹ throughout the year at rural site, DT, and urban site, D where average minimum temperatures were higher. In contrast, both species experienced stress at the rural site B and urban site L, and mean assimilation rates were lower during winter months in Q. ilex. However overall assimilation rates were higher in Q. ilex at these sites. In conclusion, although the

deciduous *Q. robur* has higher carbon assimilation rates and photosynthetic efficiency in fertile soils where water availability is high alongside higher average temperatures, those sites where soil fertility and water availability are low mean that the evergreen oaks water conservation strategy gives it an advantage. This advantage is increased at sites where mean minimum temperatures are higher and consequently *Q. ilex* winter assimilation rates remain high.

4.1 Introduction.

4.1.1 Photosynthesis.

Photosynthesis is the process by which plants utilise the energy of the sun to create biochemical reactions using CO_2 and H_2O molecules. These biochemical reactions form carbon compounds such as carbohydrates used by plant cells for respiration and growth (see Chapter 1 section 1.2.1). It follows that photosynthetic rate is directly linked to the assimilation of carbon minus the loss of carbon through cellular respiration (e.g. Lambers et al. 2008; Pallardy 2008); net photosynthesis (NP) is equal to gross photosynthesis (GP) minus respiration (R) (see equation 4.1).

$$NP = GP - R \tag{4.1}$$

(Whitmarsh and Govindjee 1999).

4.1.2 Respiration.

Respiration (*R*) is closely coupled with carbon assimilation rates and is controlled by the energy demand of the plant (e.g. Lambers et al. 2008). Shade plants therefore tend to have lower respiration rates than sun adapted plants (Givnish 1988). Shade leaves of red oaks (*Q. rubra*) had lower A_n , followed by lower *R* rate the following night, in response to reduced carbohydrate availability compared to upper sun leaves (Whitehead et al. 2004). Leaf *R* is generally suppressed in the light (Villar et al. 1995) although the reduction in CO_2 respired during day-light has been attributed to its re-use during photosynthesis (e.g. Delfine et al. 1999). *R* is also known to increase with rising temperatures however, in some plants, *R* has been shown to even-

tually decrease as thermal acclimation occurs (VanOijen et al. 2010). For example, both potted and field grown red and white oaks exposed to sudden temperature increases exhibited a rapid thermal acclimation (Bolstad et al. 2003). The same study also showed that red and white oaks could maintain maximum assimilation rates between 18 and 30°C, increasing A_n with increasing temperatures while *R* decreased, and therefore increasing the amount of carbon assimilated (Bolstad et al. 2003).

4.1.3 Diurnal rhythms.

Early studies on diurnal photosynthesis have found that tree species increase their photosynthetic levels throughout the morning reaching a peak before or around noon, depending on light levels and followed by a midday depression (Kozlowski 1971). Even earlier work showed that conifers on cloudy or overcast days increased A_n to a maximum at noon, A_n remained constant for an hour before gradually reducing, while on sunny days A_n increased quickly peaking between 9am and 12pm, decreasing before a second rise in late afternoon (Hodges 1967). More recently, red oaks, along with three other deciduous species showed a decline in A_n throughout the day (Bassow and Bazzaz 1998). For deciduous species such as oaks, second flush leaves tend to have higher mean A_n than first flush leaves (Masarovicova 1991). The same study found A_{max} in oaks tended to occur between 8 and 10am during the summer with a midday depression (Masarovicova 1991).

Closely coupled with diurnal and seasonal photosynthesis is photosynthetic efficiency, with many tree species experiencing a reduction in photosynthetic efficiency as temperature and light increases throughout the morning and in mid Summer. For instance, despite a humid summer during data collection, A_n and Gs decreased during a warm summer along with a decrease in F_v/F_m (Peñuelas et al. 1998). Moreover, F_v/F_m levels increased as temperatures dropped, with the reduction in photosynthetic efficiency attributed to a down regulation of photosynthesis during warmer temperatures (Peñuelas et al. 1998).

In situations where a decrease in photosynthetic efficiency, is not followed by immediate recovery after stress relief, reductions in F_v/F_m and associated chlorophyll fluorescence parameters may indicate photo-inhibition. Photo-inhibition can occur when shade adapted plants are subjected to sudden high light environments, coupled with either high or very low temperatures depending on the bio-climatic envelope and plasticity of the species being studied (see Alves et al. 2002). In general *Quercus* spp. tend to exhibit high phenotypic plasticity, in particular, *Q. ilex* has been shown to represent a complex species (Barbero et al. 1992) and has developed phenotypes that can withstand temperature extremes of $+50^{\circ}$ C to -10° C (Gimeno et al. 2009).

4.1.4 Water use efficiency.

Water use efficiency is generally measured as the ratio between net photosynthesis and transpiration (A_n/E) , however, as *Gs* and transpiration are directly related through stomatal function, water use efficiency can also be assessed using the ratio of A_n to *Gs*, known as intrinsic water use efficiency (*iWUE*). *iWUE* is a representation of internal CO_2 concentration and its effect on water use efficiency (Linderson et al. 2012). This parameter has been used in the literature as a substitute for *WUE*, as *E* is affected by humidity and leaf temperature and subsequently leaf to air vapour pressure difference but is also directly related to *gs* (Stokes et al. 2010). High levels of leaf level internal CO_2 can be an indication of low WUE. Therefore it is possible for a plant to maintain high levels of assimilation rates while still reducing *iWUE*. *iWUE* may increase with increasing light (Ponton et al. 2002) although this effect is generally lower for less drought tolerant species and is likely to be due to genetic differences within related oaks (Ponton et al. 2002).

4.1.5 Species and environment.

Some authors suggest there will be a change from deciduous to evergreen species with future climate change (e.g. Both and Bruggemann 2009), yet others document the succession of beech (*Fagus sylvatica*) into oak dominated forests (Rohner et al. 2012) and in drier areas the replacement of Beech dominated forests to oak (Ogaya and Penuelas 2007*b*). Furthermore, *Q. ilex* is extending its range further north into central Europe and *Q. ilex* (sub sp. *ilex*) is already found along the Atlantic coast of France as far as Brittany (Barbero et al. 1992).

Research also shows that Q. *ilex* is able to tolerate areas where pollution is a problem better than Q. *robur*. One open top chamber study on ozone (O_3) effects on gas exchange and stomatal

conductance (Gs) showed that Q. *ilex* was able to maintain higher Gs levels as O_3 concentrations increased while Q. *robur*, having larger, thinner leaves, showed signs of stomatal blockage and yellowing of leaves (Calatayud et al. 2011). In contrast, both a deciduous and evergreen oak were shown to have beneficial effects on urban temperatures and carbon sequestration potential in Rome (Gratani and Varone 2006).

To summarise, *Q. robur* enjoys maximum productivity in lowland Southern England and is normally replaced by the European deciduous Sessile oak (*Q. petrea* on poorer, upland soils (Broadmeadow and Matthews 2003). However, evidence of *Q. robur's* increasing productivity has been noted in Wistmans wood, Dartmoor since mid twentieth century (e.g. Simmons 1965). Indeed, the UK Forestry Commission has suggested *Q. robur* will replace the European Sessile oak *Q. petrea* as the dominant species in the Southwest with future climate change (Broadmeadow 2004; Broadmeadow et al. 2005). On the other hand, *Q. ilex* already has established populations in mountainous regions of the Iberian peninsula, and can tolerate climatic extremes up to an altitude of around 1500m (Reille and Pones 1992) despite exposure, poorer soils and cooler temperatures reducing productivity in these environments. The Southwest UK upland environment, although only reaching altitudes of 400 - 600masl at most, is subject to poor soils, irregular frosts and long periods of wet weather. Carbon assimilation rates and photosynthetic efficiency of both species were measured on monthly basis in such an upland environment while also comparing them in urban and rural locations within the Southwest UK, to aid comparison of carbon sequestration potential in each species.

In this chapter, section 4.3 and section 4.4 introduce light saturation curves, diurnal rhythms and average dark respiration for each species at the control site where nutrient levels and water availability were good. Section 4.5 covers monthly photosynthesis and photosynthetic efficiency in well watered control trees over a year. The results of a small drought experiment taken on the three year old control site saplings is also shown in this section. Section 4.5 first compares differences between potted (P) and field grown (G) saplings at each site (section 4.5.4) then gives results on seasonal photosynthesis (section 4.5.5) and efficiency at each site and comparisons between rural, urban and upland environments are made (where differences between

soil conditions have been analysed control site data has not been included due to there being no field planted saplings at this site and consequent unbalanced design). Section 4.5.6 covers the differences between sun and shade leaves in the upper and lower canopy (not all saplings developed sun and shade leaves due to lack of foliage). Parameter units are introduced but not used throughout the chapter, please refer to the glossary for these units.

4.2 Respiration.

4.2.1 Methods.

Six saplings of each species had photosynthetic rate measured using the LCi portable IRGA. Four mature leaves were used per tree and for *Q. ilex*, two new leaves per tree were also measured. Data for daytime assimilation rates were compared with respiration rates for the same day and results analysed using ANOVAs with assumptions for normality met (Levenes test, P > 0.05). It is acknowledged that this may not be an accurate indication of dark respiration rates during the day due to the temperature control of respiration (Bolstad et al. 2003) however this is an indication of differences in respiration between species.

4.2.2 Results.

Q. robur had higher mean net photosynthetic rates than *Q. ilex* during the day. During the night net respiration was also significantly higher for *Q. robur* than *Q. ilex* $(-1.41 \pm 2.32; -0.25 \pm 2.11$ respectively) (F=13.08_{(1,190}), p <0.001). For newly developed *Q. ilex* leaves, respiration was approximately twice the value of mature *Q. ilex* leaves (see Figure 4.1).

4.3 Light Saturation Curves.

4.3.1 Methods.

Light response curves were created for both species using well watered, two year old pot grown saplings taken from the control site. Trees were removed from the polytunnel during the experiment and averages taken from three leaves per tree from the upper and lower area of foliage (this was to ascertain if there was differences between sun and shade leaves). Photosynthetically active radiation (PAR) levels were set using neutral density filters, with the ADC light



(a) Mean respiration rate



(b) Mean daylight assimilation rate

Figure 4.1: a). Mean respiration rate for each species recorded overnight June 2010 and, b) day-time assimilation rate for daylight hours of the same day.

unit (supplied with a 'dichoric' 20 watt bulb). The maximum PAR leaf level obtained by the light unit was around 550 PAR. Where possible leaves were exposed to natural light in order to obtain higher PAR levels (see methods section 2.3.6 on page 29). Curves were produced in August 2010 for both species and in February and again in April 2011 for *Q.ilex*.

4.3.2 Results.

In August 2010 when mean temperatures were approximately 15-16 °*C*, *Q. ilex* shade leaves were light saturated at around 350 PAR at 6.5 ± 0.5 and *Q. robur* shade leaves saturated at similar light levels but assimilation rate was lower at 5.8 ± 1.08 (see Figure 4.2a on page 51). *Q. ilex* sun leaves saturated at approximately 500 PAR with assimilation rate reaching 8.8 ± 1.06 and *Q. robur* reached a maximum of 7.8 ± 0.918 at the higher PAR of 600 (see Figure 4.2). Light response of *Q.ilex* in April 2011 at *Q. robur* bud burst, showed that *Q. ilex* reached light saturation at 8.5 at 700 PAR when mean temperature was around 11 °*C*, during this experiment *Q. ilex* was able to maintain maximum rates as light increased to 2000 PAR (see Figure 4.2). However, light response in February 2011 when mean temperatures were around 5°*C* was much lower (Figure 4.3).

4.4 Diurnal Rhythms.

4.4.1 methods.

Data was collected in July 2009 to find natural diurnal rhythms. photosynthesis measurements were taken in sets of two leafs per three trees per species between 5:00 and 17:00. The averages of each set of data were used for analysis. Results were then pooled according to the nearest 15 minute interval, between 5:00 and 17:00. No distinction between upper and lower canopy were made at this growth stage. For diurnal rhythms using the light unit, data was collected during July 2010 in sets of 3 trees per species on two upper and two lower leaves plus two new flush leaves where possible. Data was collected between 8:30 17:00 and analysed as before using averages of sets.



Figure 4.2: Light response curves for a) shade, b) sun, leaves for each species taken in August 2010, and c) *Q. ilex* April sun leaves


Figure 4.3: Light response curves for Q. ilex winter sun and shade leaves (February 2011).



Figure 4.4: Diurnal rhythms under natural light for *Q. ilex* (open triangles) and *Q. robur* (closed squares). (means \pm 95 % CI), data taken over two consecutive days in July 2009 on two leaves from three trees per species. Average temperatures for July 2009 were 19.1 \pm 0.39 and for the two day period July 10-11th 2009 16 \pm 1.04 and days were clear and sunny.



Figure 4.5: Diurnal rhythms at fixed PAR 550 μ mol s⁻¹ m² for, a) lower, b) upper, c) new and d) all, leaves for each species (means \pm se shown per time interval)

4.4.2 Results.

For natural diurnal rhythms, *Q. robur* had lower assimilation rates than *Q. ilex* with maximum photosynthesis(A_{max}) of around 5.5 occurring at 10:00 and again at 13:30. In contrast, *Q. ilex* reached maximum assimilation rates closer to 7.5 but these quickly fell after midday and remained low until the end of data collection (see Figure 4.4 page, 52).

Diurnal rhythms taken using the light unit however showed Q. *robur* to have higher assimilation rates and mean A_n reached close to 11 (see Figure 4.5 d). Mature upper leaves Q. *robur* obtained a maximum A_n , 5 higher than Q. *ilex* by mid morning 11-11:30 and continued to maintain these rates until the mid day decrease around 13:00 (see Figure 4.5. Although both species experienced a mid-day decrease in assimilation rates, Q. *robur* increased its rate again in the late afternoon.

A similar pattern is followed with lower leaves, but with lower mean assimilation rates (see Figure 4.5). New leaves have the greatest assimilation rate and seem to reach higher rates earlier in the day with *Q. robur* having reached maximum rates by 09:30. Mean rates did not decrease again until 13:00 but again *Q. robur* increased mean A_n to 7-8 after 16:00 (see Figure 4.5 page 53).

4.5 Species and Environment.

4.5.1 Methods.

See chapter 2, section 2.3.6 for sampling protocol for monthly photosynthetic parameters. For the drought experiment three 3 year saplings of each species were subjected to two weeks of drought and measurements of photosynthetic parameters were taken on four leaves of each tree.

Results shown are; monthly assimilation rates and stomatal conductance of CO_2 (A_n and Gs) and photosynthetic efficiency (F_v/F_m and PI) for control saplings. Results from the drought experiment show A_n , Gs, sub-stomatal CO_2 and transpiration (E) data, but unfortunately no chlorophyll fluorescence data was taken for this experiment. Sections 4.5.4 gives any significant differences between pots and ground planted saplings in order to indicate how fertility affected each species. Seasonal differences between species and site or site characteristic (rural, urban or upland) are shown, and finally section 4.5.5 gives species differences in leaf position in canopy at each site to show differences in sun and shade leaves where relevant (not all saplings developed thick enough canopies for sun and shade differentiation). Intrinsic water use efficiency was also measured using A_n / Gs and this was analysed between species, site and soil condition.

4.5.2 Results: Monthly Photosynthesis and Chlorophyll Fluorescence.

In well watered control saplings, Both species reached maximum assimilation rates in Spring (8 ± 1.6) , however *Q. ilex* attained this maximum earlier as *Q. robur* leaves were still developing (Figure 4.6). However, by the end of May, *Q. robur* reached comparable A_{max} and maintained these higher assimilation rates throughout the summer months until senescence towards the end of October when mean air temperatures dropped (Figure 4.6). As temperatures decreased *Q. ilex* A_n increased again as *Q. ilex* took advantage of the milder Autumn weather. Chlorophyll fluorescence values remained high in *Q. ilex* despite its down regulation of assimilation rates during the hotter summer months, however both species showed reduced *PI* during the colder months: *Q. robur* as leaf senescence began in Autumn and for *Q. ilex*, during the Winter months even when A_n was high (Figure 4.7).

4.5.3 Results: Drought.

Table 4.1: ANOVA table showing effects of treatment (drought or well-watered saplings) on mean photosynthetic rates $(A_n \ \mu mol \ s^{-1} \ m^{-2})$, sub-stomatal CO_2 (Ci νpm), transpiration rates (E $mmol \ m^{-2} \ s^{-1})$ and stomatal conductance of CO_2 (Gs $mol \ m^{-2} \ s^{-1})$ showing F statistics and p-values

Physiological parameters									
			A_n	C	Ci	E		Gs	
Factors	DF	F	Р	F	Р	F	Р	F	Р
SPECIES	1	8.49	0.005**	4.64	0.04*	11.78	0.001***		ns
TREATMENT	1	46.42	0.0***		ns	120.13	0***	50.65	0***
SPECIES*TREATMENT	1	6.46	0.01**	4.43	0.04*		ns		ns
Error	64		ns		ns		ns		ns
ADJ R^2 (%)		45.62		11.44		65.12		41.84	

D 14	CINE	ANOVA	4.1.1.
Results	OF LIVE	ANUVA	table

Q. ilex had around 50% lower mean A_n , in well watered saplings, than *Q. robur* (see Figure 4.8a). However, both species had similarly reduced A_n with drought (1.7 - 2.0 $\mu mol m^{-2} s^{-1}$)



Figure 4.6: a) Monthly assimilation rate (A_n) and b) Monthly stomatal conductance (Gs) for well watered control site saplings, shown with mean monthly temperatures (°C) (means \pm 95% CI).



Figure 4.7: Monthly a) Maximum photochemical quenching shown by (F_v/F_m) and b) Sapling vitality shown by performance index (*PI*) for well watered control site saplings, shown with mean monthly temperatures (°C) (means \pm 95% CI).



Figure 4.8: Differences in photosynthetic parameters for well watered saplings and saplings that had been subjected to two weeks of drought conditions), a) mean net assimilation rates (A_n) , b) mean stomatal conductance (Gs), c) mean transpiration rate (E) and d) mean sub-stomatal CO_2 .

(see 4.8 and 4.1 on pages 58 and 55.

Mean *Gs* was 10% lower for drought saplings in both species compared to well watered values. Despite similar *Gs*, *E* was higher in *Q. robur* saplings during drought and *Ci* was also lower in well watered *Q. robur* saplings despite similar decreases with drought for both species (see table 4.1, page 55 and Figure 4.8, page 58).

4.5.4 Results: Differences between potted and field grown saplings

Although there were significant effects from the urban, rural and upland environment, these effects were site specific and

Data for net assimilation rate A_n and photosynthetic efficiency parameters F_v/F_m and PI had equal variances using Levenes test for any continuous distribution (p>0.05) for all factors. However, Gs and PI data were skewed, therefore data were log transformed to equalise variances (p>0.05). Mean assimilation rates were not affected by condition (table 4.2), however an interaction with site showed that urban saplings had higher A_n in (this was more obvious in Q. ilex where A_n ranged between 3.5 - 5.5 μ mol $m^{-2} s^{-1}$ in potted saplings and 5.5 - 7.2 μ mol m^{-2} s^{-1} in field grown saplings) whereas potted were greater than field grown in rural saplings. Gs was higher in potted for Q. robur although conversely in Q. ilex field grown had higher rates of Gs. Photosynthetic efficiency was not different between potted and field grown in Q. ilex however, in Q. robur saplings potted was always higher than field grown (most obvious at the upland site where F_v/F_m varied from 0.7- 0.78 in field grown planted saplings to 0.8-0.82 in potted). All parameters measured had a three-way interaction between soil condition, species and site and showed that carbon assimilation rates and photosynthetic efficiency parameters were higher in pots than field grown saplings, however at the urban site (L) the opposite was true. F_v/F_m was higher in pots than field grown saplings for Q. robur, whereas for Q. ilex the opposite was true. (Table 4.2 shows effects and interactions between soil condition, species and site for photosynthetic parameters).

Table 4.2: ANOVA table shows significant differences between sites, species and soil condition (where Condition = potted (P) or field grown (G) saplings). (F statistics and p<0.05 '*', p<0.01 '**', p<0.001 '***' and p>0.05 'ns'). Data was analysed using application of general linear models on entire annual dataset.

ANOVA Table: Differences between field grown and potted saplings and interactions with site and species shown

Physiological parameters		A	A_n LogGs		iWUE		FV/FM		log(PI)		
FACTORS	Df	F	sig.	F	sig.	F	sig.	F	sig.	F	sig.
CONDITION	1	0.0	ns	37.7	***	25.9	***	2.4	ns	0.1	ns
SPECIES	1	12.0	***	118.5	***	113.1	***	4.3	*	560.1	***
SITE	4	7.7	***	68.4	***	43.0	***	122.5	***	123.0	***
CONDITION:SPECIES	1	0.5	ns	15.4	***	1.9	ns	5.6	*	5.9	*
CONDITION:SITE	4	6.1	***	15.4	***	10.9	***	13.7	***	19.7	***
SPECIES:SITE	4	61.8	***	34.4	***	5.2	***	56.9	***	60.1	***
CONDITION:SPECIES:SITE	4	5.3	***	12.7	***	2.9	*	5.0	***	10.5	***
Residuals IRGA	3229										
Residuals PEA	5287										

Sapling *iWUE*.

Effects on *iWUE* were site specific (Table 4.2). *iWUE* was higher in *Q. ilex* than *Q. robur*, generally higher in pots than field grown saplings except at the urban site, L, where field grown saplings had significantly higher *iWUE* than pots. For field grown saplings, *iWUE* was also significantly higher at the urban site L than all other sites in both species (74-86 μ mol CO₂ / H_2O^{-1} compared to 40-50 μ mol CO₂ / H_2O^{-1} compared for all other sites and species combinations).

4.5.5 Results: Seasonal Leaf Physiology for Species and Environment.

Phenology.

Forestry Commission phenology research gives phase 2 as bud formation and greening, phase 3 as leaf unfurling and visible leaf form and phase 4 as fully expanded leaves (Forestry Commission 2012*a*). Using this information and 20th April 2010 as a reference date (this being an average date for bud burst in Southwest as shown by the Forestry Commission bud-burst map (Forestry Commission 2012*a*) *Q. robur*), bud-burst at the different rural and urban locations in 2010 were:

Upland site: At HMP Dartmoor prison farm, *Q. robur* buds had formed but not bursting on 20th April 2010 (early phase 2), early stages of bud-burst (early phase 3) on most *Q. robur* noted on

28th April 2010. Urban sites: At Digginit bud-burst was occurring in most saplings with many new leaves (phase 3). At Laira bud-burst was occurring with some new leaves (phase 2 and early signs of 3). Rural sites: At Broadley bud-burst was occurring with one or two new leaves (phase 2 and early 3) At Dartington bud-burst with many new leaves (later stages of phase 3). Control: First signs of bud-burst recorded on 10th April (phase 3) with all new leaves by 20th April although many red (late stage phase 3 early stage phase 4).

Season and Environment.

Seasonal F_v/F_m was around 0.75-0.78 in rural and urban sites, increasing to 0.82 by July 2010 and remained high throughout the year in Q. ilex. However F_v/F_m experienced a drop in January 2011 but had already recovered by April 2011 (Figure 4.10a on page 64). The upland site followed a similar pattern with a more pronounced dip during the colder months (dropping to around 0.5 in November and January). mean PI values followed similar patterns to mean F_v/F_m although more variability was experienced (Figure 4.10b on page 64), in particular, the combined urban sites show much higher PI values than the rural sites (Figure 4.10b) and reached maximum values around 12 for Q. robur and 14 for Q. ilex. For seasonal A_n in Q. robur all combined site characters began low at around 4 μ mol m⁻² s⁻¹ in May 2010 and increased to a maximum in July 2010, although this maximum was lower at the upland site (8.5 μ mol m⁻² s^{-1}) and as high as 16 μ mol $m^{-2} s^{-1}$ for the rural sites (Figure 4.9a on page 63). The same pattern occurred during the next season although A_n was lower in general. This was especially so for the urban sites which had a combined mean of around 6 μ mol m⁻² s⁻¹ in July 2011, compared with 12 μ mol m⁻² s⁻¹ for the rural sites (Figure 4.9a A on page 63 . Q. ilex had a lower A_n in general (around 8 μ mol $m^{-2} s^{-1}$) and all sites achieved this maximum in July 2010 although for the upland site, a later bud-burst meant Spring A_n was low (Figure 4.9a B on page 63). Q. ilex maintained maximum rates throughout the winter months in combined rural and urban sites although the rural sites increased A_n to 10 μ mol $m^{-2} s^{-1}$ in July 2011 while the urban site reduced mean A_n from March to July 2011 (8 - 6 $\mu mol \ m^{-2} \ s^{-1}$). In contrast the upland site reduced A_n in march 2011 increasing steadily to another maximum in July 2011 of 8 μ *mol* $m^{-2} s^{-1}$ (Figure 4.9a B).

At the upland site, *Q. ilex* was unable to maintain high assimilation rates in winter (as low as $2 \ \mu mol \ m^{-2} \ s^{-1}$) and this corresponded with lowered *Gs*, and photosynthetic efficiency for this site (figures 4.9aA, 4.9bA, 4.10aA, 4.10bA on pages 63, 63, 64, 64). However, *Q. ilex* was able to maintain higher A_n at the rural sites for all seasons although there was a more marked increase in summer at rural site B compared to the other seasons at this site (Figure 4.9aC). Photosynthetic efficiency was also compromised during the winter months in *Q. ilex* despite its efforts to maintain assimilation rates (Figure 4.10a, 4.10b). Although both rural sites are a similar distance from Dartmoor the rural site B is located on higher ground (80masl) and experienced lower winter temperatures and rainfall than the other rural site DT (see section 3 Figure 3.4 on page 39). The urban site show *Q. ilex* A_n was uniform across the seasons except at L summer assimilation rates were lowered (Figure 4.9aE). This corresponded with reduced *Gs* at this site and although photosynthetic efficiency values were lowered during winter for the urban sites, L experienced lowered PI across all seasons (Figure 4.10bE, page 64).

In comparison, although Q. *robur* A_n is generally lowered during spring/autumn, this species enjoyed high A_n at most sites during the summer and only showed lowered A_n at the urban site L. Despite these high summer assimilation rates at the upland site, photosynthetic efficiency was also low at the upland site in Q. *robur*.

4.5.6 Results: Sun, Shade and Environment.

Generally upper leaves had higher mean A_n than lower leaves, although Q. *ilex* showed more distinction between upper and lower leaves than Q. *robur*. In fact those sites where Q. *robur* had reduced productivity, the subsequent lack in distinction between upper and lower leaves was due to less foliage for this species. This was most evident at rural site B, urban site L and the upland site (figures 4.11 and 4.12).

Mean *Gs* at the rural sites followed a similar pattern to A_n and was higher in upper than lower leaves in *Q. ilex* whereas for *Q. robur* there was no difference. At the upland site both species had low *Gs* and there was no differences between leaf position and although lower at the urban sites there was no distinction between upper and lower leaves in *Q. ilex* increased *Gs* in *Q. robur* at D was greater in upper than lower leaves (Figure 4.11).



(b) Mean Seasonal Stomatal Conductance (Gs).

Figure 4.9: Species differences in a) net assimilation rate A_n and b) stomatal conductance (*Gs*) for season and sites (means $\pm 95\% CI$). Significant differences between species and seasons at each site is shown by different letters (p<0.05).



(b) Mean seasonal performance index PI for each site

Figure 4.10: Species and seasonal differences in a) variable fluorescence over maximum fluorescence ratio F_v/F_m and b) performance index *PI* for each site (means $\pm 95\% CI$). Significant differences between species and seasons at each site is shown by different letters (p<0.05).



Figure 4.11: Species and environment differences in Upper (U) and Lower (L) canopy leaves. Individual sites used (Urban sites L and D, rural sites DT and B and upland site DPF) for, A) net assimilation rate A_n , B) stomatal conductance Gs, C) Maximum yield of primary photochemistry in photosystem II shown by the variable fluorescence, maximum fluorescence ratio F_v/F_m , and D) chlorophyll fluorescence performance index *PI* (means $\pm 95\% CI$).



Figure 4.12: Species and environment differences in Upper (U) and Lower (L) canopy leaves. Sites are pooled into urban, rural and upland for, A) net assimilation rate A_n , B) stomatal conductance Gs, C) Maximum yield of primary photochemistry in photosystem II shown by the variable fluorescence, maximum fluorescence ratio F_v/F_m , and D) chlorophyll fluorescence performance index *PI* (means $\pm 95\% CI$).

In both rural and urban sites, F_v/F_m and *PI* were reduced in upper leaves compared to lower for both species (Figure 4.12). At the upland site, F_v/F_m and *PI* were generally low for both species. Mean values were generally higher in *Q. ilex* but there was no distinction between upper and lower leaves for either species at this site (see table 4.3, page 66 for ANOVA table).

Table 4.3: ANOVA table: Effects on physiological variables of leaf position (upper U or lower L) including site and species interactions for field grown saplings showing F statistics and p<0.05 '*', p<0.01 '**' and p<0.001 '***' and p<0.05 not significant 'ns'.

Analysis of Variance Table												
Physiological variables		A	n	G	s	iW	UE	$F_{v}/$	F_v/F_m		PI	
FACTORS	Df	F	p<	F	p<	F	p<	F	p<	F	p<	
SPECIES	1	13.2	***	46.2	***	95.3	***	4.5	*	255.8	***	
SITE	4	7.4	***	28.8	***	44.8	***	125.7	***	114.0	***	
LEAF POSITION	1	80.8	***			8.4	**	159.0	***	105.2	***	
SPECIES:SITE	4	66.6	***	15.8	***	3.6	**	57.3	***	40.5	***	
SPECIES:LEAF POSITION	1	0.7	ns			4.1	*	4.6	*	4.8	*	
SITE:LEAF POSITION	4	3.1	*			1.3	ns	2.8	*	1.5	ns	
SPECIES:SITE:LEAF POSITION	4	5.9	***			2.7	*	4.1	**	4.1	**	
Residuals	3195							5281				

4.6 Discussion.

Differences in assimilation rates and photosynthetic efficiency indicate Q. roburs preference for fertile soils (Boisvenue and Running 2006) as assimilation rates and efficiency tended to be higher in potted plants. In contrast, Q. *ilex* thrives on poorer, drier soils (Barbero et al. 1992). Dark respiration was lower in Q. *ilex* than Q. *robur* and tends to be lower in Mediterranean evergreens compared to deciduous species in general (Gratani et al. 2008), however although extensive experiments were not undertaken to deduce changes in respiration with temperature recent experimental evidence suggests that although respiration initially increases with increasing temperatures, acclimation occurs and respiration returns to pre temperature increase values (VanOijen et al. 2010). Additionally, daytime respiration is thought to be similar to that of night time respiration as any extra CO_2 used in respiration is taken back into the photosynthetic process (Bolstad et al. 2003).

Both oaks studied here are late successional species, however my research indicates that Q. *robur* is more shade tolerant than the Mediterranean Q. *ilex*, deciduous oak saplings have been shown to be less tolerant to high light than Q. *ilex*, in a Mediterranean environment where de-

ciduous oaks have to establish under thick shade (Gomez-Aparicio et al. 2006). Additionally, during April when temperatures were mild *Q. ilex* made the most of light levels to increase assimilation rates at higher light levels than it did during the summer. This strategy enables it to make the most of winter light levels, which it can't do during the summer when drought and high temperatures are more likely to occur. However, this increase in light saturation level during early spring led to a reduction in photosynthetic efficiency during the winter months and leaves showed signs of reversible photo-inhibition during this season. Another study monitoring photosynthetic traits in *Q. ilex*, growing in two different climatic conditions in Italy, showed that light response increased with temperature, maximum assimilation rates were lower during winter (at light saturation) and photosynthetic efficiency was reduced, shown by F_v/F_m in leaves during winter at the colder site (Gratani, Pesoli, Crescente, Aichner and Larcher 2000). However earlier laboratory experiments proved that at optimum temperatures light saturation in *Q. ilex* was around 1000 $\mu mol s^{-2}$ (e.g. Larcher 1969) cited in (Gratani, Pesoli, Crescente, Aichner and Larcher 2000).

In this thesis *Q. ilex* saplings showed a much more conservative response to summer temperatures and light; closing stomata and down regulating photosynthesis regardless of water availability as protection to the photosynthetic apparatus occurs. This is a conservative water use strategy and has also been shown in the Mediterranean deciduous Turkey oak (Tognetti et al. 2007). However, the deciduous species does not down regulate photosynthesis in response to higher temperatures and therefore its drought response is slower than *Q. ilex*. The ability of *Q. robur* to increase assimilation in warmer weather leaves this species more vulnerable to drought. This strategy even competes within its own sympatric species, *Q. petrea*, which had 10-15% higher intrinsic water use efficiency than *Q. robur* when grown in full sun despite *Q. robur* net assimilation rates being higher (Ponton et al. 2002).

In order to reduce the trade off in assimilation rate during the summer Q. *ilex*, increases its photosynthetic light response during milder weather in the cooler months. This is evidence of its plasticity towards sun/shade adaption where carbon fixation is is favoured in high light environments and light harvesting in low (Friedrichs et al. 2009) (this is also shown in LMA

responses shown in chapter 5). However lower photosynthetic efficiency at lower temperatures makes this species more vulnerable to photoinhibition (Camarero et al. 2012).

The urban environment exerts stresses on oak growth through limited water availability as well as heat and pollution. Saplings at the urban sites exhibited different responses. It is hard to separate effects of pollution and water stress through field studies alone although further work on C_{leaf} and N_{leaf} showed higher levels of N_{leaf} in Q. robur at the urban versus rural environments, evidence that Q. robur accumulates N in leaves through N deposition in urban environments (Searle et al. 2012). This may mean that Q. robur is able to make use of N, and not be inhibited by it. In fact, one study has shown that Q. robur was more affected by water availability than particulate pollution in an urban environment (Ferretti et al. 2002), although another showed that Q. robur leaves were more susceptible to urban pollution than Q. ilex (Calatayud et al. 2011).

4.7 Conclusion.

My research showed that Q. robur had higher mean assimilation rates than Q. *ilex*. However, Q. *ilex* had higher assimilation rates at the urban site L where soils were drier and saplings were more exposed, despite photosynthetic efficiency being lower in Q. *ilex* at this site. Net carbon assimilation A_n and the more sensitive chlorophyll fluorescence parameter PI were significantly lower in Q. *robur* than Q. *ilex* at rural site B. This site was situated on the eastern Dartmoor 'fringe', having the second lowest air temperatures after the upland site and second driest site after the urban site L.

Higher iWUE, in both species, at the exposed urban site L indicated water stress at this site. Struve et al. (2009) found that *Q. robur* grew the tallest and used more water when compared with *Q. ilex* growing in an urban environment, indicating that the conservative water use of the evergreen *Q. ilex* would give this species an advantage in urban environments, while the colder, wetter upland environment should give the deciduous *Q. robur* the advantage as its deciduous habit gives it protection during winter cold stress.

Although photochemical efficiency is lower in young Q. ilex in response to cold stress. Tem-

peratures above 5 $^{\circ}$ C tend to enable increased assimilation in this species. *Q. ilex* has been shown to represent a complex species (Barbero et al. 1992) and phenotypic plasticity within this species enable it to withstand climatic extremes (Gimeno et al. 2009).

In conclusion, the response of each species was site specific with Q. *ilex* net assimilation and photosynthetic efficiency higher at those sites that were drier compared with Q. *robur* and colder average temperatures reducing photochemical efficiency in both species although Q. *ilex* was able to increase assimilation rates when mean temperature were above 5 °C.

Chapter 5

Leaf Traits.

Analyses of leaf traits can be a useful addition when measuring productivity and carbon sequestration potential in relation to environment. Specific leaf area (SLA), leaf nitrogen content (N_{leaf}) and leaf carbon (C_{leaf}) are closely coupled with net photosynthesis and relative growth rates, where the inverse of SLA, leaf mass per area (LMA), increases along with N_{leaf} and net assimilation rates and low SLA is generally combined with higher Cleaf concentrations and faster growing species. Leaf traits were measured in two year old saplings of Q. robur L. and Q. ilex subs. ilex.L growing at two urban, two rural and one upland site in and around Plymouth SW England. Leaf traits were analysed for differences between species and sites in relation to an urban, rural and upland site using mixed general linear models. SLA and N_{leaf} were higher in Q. robur than Q. ilex in common with generalised leaf traits for evergreen and deciduous broadleaved species. However, Cleaf and Nleaf were higher in Q. ilex sun versus shade leaves. Mean SLA tended to decrease with increasing light in Q. robur while SLA increased at sites where mean temperatures were lower and this effect was greater in potted saplings. Mean SLA was less variable in Q. ilex although a similar pattern was seen at the upland site. This chapter then discusses the relationship with SLA and environment. However, further evidence for the observed trends is linked to biomass allocation and general growth strategies employed by deciduous and evergreen species.

5.1 Introduction.

Commonly, leaf traits are used to measure growth and productivity in trees. Leaves as the main site of carbon fixation and used alongside more direct growth and biomass experiments, can give realistic estimates of carbon sequestration potential (e.g., Cornelissen et al. 2003). A leaf trait of particular importance, is specific leaf area SLA (leaf area / leaf weight, $cm^2 g^{-1}$), or

its reciprocal *LMA* a much used parameter that is related directly to relative growth rate (RGR) through assimilation rates (A_n) and leaf area ratio $(LAR)^1$. Other commonly used leaf traits are leaf carbon and nitrogen concentrations which can be used to infer construction costs in relation to leaf habit and environment. It follows that leaf traits are known to change with environmental pressures such as drought (Ogaya and Penuelas 2006), pollution and average temperatures (Cornelissen et al. 2003) and are considered as both morphological and physiological adaptations to their environment (e.g. Gratani and Bonmbelli 2001). As such leaf traits can aid interpretation when comparing functional differences in evergreen and deciduous trees.

5.1.1 Specific Leaf Area.

Generally *SLA* increases with decreasing irradiance, and leaves developed in high light tend to be smaller and thicker with correspondingly low *SLA* (Evans and Poorter 2001). This is due to an increase in palisade cells and photosynthetic enzymes that are necessary to maximise the carboxylation process (Lichtenthaler et al. 2007). On the other hand, shade adapted leaves are larger and thinner and contain more light harvesting pigments, arranged to avoid self shading, and therefore increasing the area to mass relationship of the leaf (Lichtenthaler et al. 2007).

5.1.2 Leaf Nitrogen.

Leaf nitrogen (N_{leaf}) tends to be associated with concentrations of the enzyme, Ribulose-1 5-Biphosphate Carboxylase Oxygenase (RuBisCO) and thus increased N leads to greater carbon fixation in leaves (Grime 2002; Koike et al. 2004; Lorenz and Lal 2010). However, light, temperature and drought modify the effect of N on photosynthesis (Lorenz and Lal 2010), in particular N_{leaf} tends to increase through the canopy along with increasing light availability (Niinemets and Tenhunen 1997).

The foliar deposition of N is an important issue (Kennedy 2003) and has been shown to increase assimilation at low concentrations while having a detrimental effect at higher concentrations (WHO 2011). Despite this, N deposition in the UK does not seem to affect oaks (Kennedy 2003).

¹leaf weight ratio *LWR X SLA*, where *LWR* is the ratio of leaf weight to plant weight see equation 6.7 (Fitter and Hay 2002)

A substantial amount of N_{leaf} is utilised by photosynthetic organs within the leaf, and thus increasing N_{leaf} is commonly associated with increasing carbon assimilation rate (Takashima et al. 2004). Leaf level net assimilation and leaf nitrogen ratio (A_n/N_{leaf}) is known as photosynthetic nitrogen use efficiency (*PNUE*) as 50% of N_{leaf} generally being used for photosynthesis (Takashima et al. 2004). Evergreens generally have lower *PNUE*, as do species growing at higher altitudes where an increase in nitrogen associated with cell wall formation may occur, or, in other areas where heat and drought stress induce stomatal control of photosynthesis and leaf cuticles are thickened to reduce water loss (Takashima et al. 2004).

5.1.3 Leaf Traits and Environment.

Changes in leaf traits such as *LMA* are an indication of adaptation to stress, and species adapted to drier environments such as *Q. ilex* are known as 'schlerophylls'. The degree of adaptation to xeric environments is measured on an index of 'schlerophily' which starts at *LMA* > 7.5 mg cm^{-2} (Filippo 2008).

Q. ilex has an average leaf life span (LL) of 36 months (Gratani and Bonmbelli 2001) while *Q. robur* LL can average from 160 to over 200 days (this research), depending on site and environmental conditions (M. et al. 2006; Bobinac et al. 2012). Both species are episodic (Kozlowski 1971) and produce a second flush in August. *Q. ilex* can continue to produce new leaves periodically throughout the year if conditions are mild (personal observation). *Q. robur* retains leaves until late October (again depending on environment) although they can remain on a young tree much longer despite being photosynthetically inactive (marcescence²) (Morin et al. 2010).

In my research, *Q. ilex* saplings first year leaves were still important contributors to photosynthetic gain as second year leaves but as third year leaves their photosynthetic rates were halved. For this reason mature current year leaves (CYL) were used for analyses of leaf traits although some data on developing current year leaves is presented separately.

- 1. General working hypotheses for Leaf traits are:
 - 1.1. There is a significant difference in leaf traits between species and leaf position (up-

²Marcescence: The retention of dead plant anatomy, here referring to leaves.

per and lower).

- 1.2. There are significant differences in leaf traits between species and sites (with interactions).
- 1.3. Are there any significant effects of site character (SC: URBAN RURAL or UP-LAND) on leaf traits and is this effect different for upper and lower leaves.

5.2 Methodology.

Mature leaves from the previous years growth were still important contributors to photosynthetic gain and assimilation rates were only reduced in three year old leaves, mainly through self shading of foliage (data not shown). With this in mind mature current year leaves (CYL) were used for analyses of leaf traits although, for *Q. ilex*, as the year progressed the sampled leaves became previous years foliage as second flush leaves matured. In some cases where saplings were under stress and leaves matured and died quickly, some data on developing current year leaves is also presented as new leaves (NL).

For each site and condition two leaves were taken from 10 saplings: One leaf from upper and one from lower canopy were measured in order to see if there were differences in 'sun' and 'shade' leaves. Where there was no distinction between upper and lower foliage it was expected for there to be no difference in results due to lack of self shading and therefore 'shade' leaves not present.

Fresh leaves were sealed in bags and transported straight to the laboratory in a cold box for immediate measurement of fresh weight. After weighing, leaf area (LA) was determined using the LiCor Leaf area Analyser and each leaf dried in the oven at 70°C until dry (two consecutive weights the same in a sub sample of each set). Dry weights were then recorded and *SLA* and dry matter content (*LDMC*) determined (*LDMC* data not shown). Mass based leaf carbon C_{leaf} and N_{leaf} were analysed as percentage concentrate in 2 mg of leaf, using dried and ground leaf samples from each treatment level (see chapter 2 section 2.3.3 for complete methodology). Leaf traits quantified are; *SLA* (sometimes discussed as *LMA* within the wider literature), mass based N_{leaf} and C_{leaf} . N_{leaf} was determined on a mass basis and N_{leaf} on an area basis was

determined by multiplying the mass based N_{leaf} by LMA (Wright et al. 2005).

5.2.1 Statistical Analyses.

2010 leaf traits were analysed using a mixed general linear model (MGLM) after making sure the data met the assumptions for ANOVA (Levenes test, p>0.05). All mixed linear models were performed using the lm(model) command in R version 2.10 (R 2009). To avoid over complication of interactions first the model was run using ENVIRONMENT and its interaction with SPECIES (SP) to look for species differences within the urban, rural and upland environments; a second model was produced using SITE (S) nested in ENVIRONMENT and analysed with planting TREATMENT (field grown (G) or potted (P)) and LEAF POSITION (LP) plus interactions. Interval plots show means \pm 95% confidence intervals and descriptive statistics within text give means, standard deviation (StD) minimum, maximum, skew and sample number (N) (Table 5.1). Table 5.1 also gives tukeys honest significance test (HSD) results for differences between sites for each species.

5.3 Results.

SLA and N_{leaf} were significantly higher in Q. robur than Q. ilex, (SLA, 16.9 ± 6.6 : 7.6 ± 2 . (cm² g^{-1}), N=539 and N_{leaf} 2.4 ± 0.32: 1.5 ± 0.23 (%), N=60). C_{leaf} was generally around 47 % in both species, however there were also significant effects of site and species site interactions where for Q. ilex rural site DT was significantly higher C_{leaf} and in Q. robur urban site D had signicantly lower C_{leaf} (48.4 ± 0.95: 46 ± 1.35 (%), N=12 respectively)(see figures 5.4 to 5.7, pages 82 to 85). To summarise Q. robur SLA was highest at upland DPF (150.4 ± 54.43) and lowest at L (124.2 ± 21.43). Q. ilex SLA was also lowest at urban L (59.06 ± 12.4) but highest mean values were found at B for this species (76.27 ± 32.08) (see Table 5.1).

5.3.1 Effect of the Urban, Rural, Upland Environmental Gradient.

Leaf traits were affected by ENVIRONMENT, however, the upland environment has a greater affect on *SLA* and N_{leaf} , while differences in environment in C_{leaf} are also site specific (see figures 5.1, 5.2 and 5.3. In *Q. robur* the higher *SLA* at the upland site (Figure 5.1) accounts for the difference in *SLA* with SC while for *Q. robur* C_{leaf} was lower than in *Q. ilex* although at the

upland site *Q. robur* had greater C_{leaf} than rural and urban sites. The higher C_{leaf} at rural DT and lower at urban D account for the significant effects of ENVIRONMENT on *Q. ilex* C_{leaf} (Figure 5.2). N_{leaf} is also site specific and although higher in *Q. robur* the difference between sites mean there is little difference between the rural and urban environments and again it is the upland site that gives the significant effect of this factor (see Table 5.1 for descriptive statistics and Table 5.2 for model results).

5.3.2 Effect of Condition: Differences between Potted and Field Grown Saplings.

Although *Q. robur SLA* was higher in DPF potted saplings (Figure 5.7), there was no significant effect of TREATMENT or interactions with TREATMENT and other factors on C_{leaf} or *SLA* (ANOVA results not shown). However, N_{leaf} was significantly affected by TREATMENT and this interacted with all other factors in the model (Table 5.3). N_{leaf} was significantly lower in field grown saplings G at the upland site DPF for both species (Figure 5.6 a and b) although this effect was stronger for *Q. robur*. Pots were less variable for *Q. robur* N_{leaf} although the rural site B was significantly lower than urban D, and rural DT (Figure 5.6).

5.3.3 Sun and Shade Effects.

No interactions with TREATMENT were observed therefore a simplified model was produced using full interaction plot with SP*S*LP (Levenes test, P>0.05 for all leaf traits) (table 5.4).

SLA.

The effect of LP interacted with SP and S for *SLA* (Table 5.4 and Figure 5.7). In *Q. robur* mean *SLA* was significantly lower in upper leaves and interacted with SITE to show more constrained and lower *SLA* at rural DT and urban L especially in upper leaves (Figure 5.7), however, at the upland site, DPF, there is no difference between upper and lower leaves and indeed there was no distinction between lower and upper canopy at this site in *Q. robur* and therefore no 'shade' leaves. *SLA* was less variable in *Q. ilex* although higher in both upper and lower leaves at rural B, there was no difference between upper and lower at this site. There were however, significant differences between upper and lower leaves at urban D and rural DT for this species an indication of the dense foliage see at this site giving a clear distinction between sun and

			Des	criptive s	statistics	for leaf traits			
Variable		SITE	Ν	Mean	StDev	Minimum	Maximum	Skewness	Tukey
					Q. ilex				
SLA	Rural	В	38	71.4	33.3	29.6	161	1.56	А
		DT	40	60.8	15.12	39.45	97.6	0.74	AB
	Urban	D	39	56.8	10.5	39.17	82.3	0.56	В
		L	38	58.0	10.25	34.76	86.7	0.37	В
	Upland	DPF	17	63.3	11.81	47.93	86.6	0.32	AB
		С	18	52.6	9.64	35.34	69.0	0.19	В
N	Dural	р	12	1 25	0 155	1.07	1 55	0.42	C
INleaf	Kurai	D DT	12	1.55	0.133	1.07	1.55	-0.42	
	TT-1.		12	1.03	0.179	1.35	1.93	0	AB
	Urban		12	1.48	0.137	1.28	1.00	-0.19	BC
	TT1	L	12	1.70	0.208	1.35	2.10	-0.39	A
	Upland	DPF	12	1.39	0.193	1.17	1./1	0.74	C
	Control Site	CS	6	1.53	0.126	1.30	1.67	-1.39	ABC
C_{leaf}	Rural	В	12	47.5	0.687	46.5	48.5	0.12	AB
		DT	12	48.4	0.948	46.6	49.9	-0.17	А
	Urban	D	12	47.3	1.21	44.2	48.4	-1.61	В
		L	12	48.0	0.605	46.8	48.9	-0.68	AB
	Upland	DPF	12	47.8	0.896	46.7	49.6	0.53	AB
	Control Site	CS	6	46.9	0.61	46.4	48.0	1.26	В
					Q. robur				
SLA	Rural	В	40	160.6	42.7	96.8	295	1.22	В
		DT	39	138	29.4	72.6	220	0.24	В
	Urban	D	36	153	51.2	53.2	265	0.43	В
		L	32	145	27.1	101	200	0.57	В
	Upland	DPF	38	190	53.2	54.3	282	-0.4	А
	Control site	CS	20	154	37.7	81.3	245	0.52	В
N.	Dural	B	12	2 21	0 231	1.83	2 70	0.47	C
INleaf	Kulai	DT	12	2.21	0.251	1.05	2.70	1.44	1
	Urbon		12	2.08	0.10	2.32	2.03	-1.44	
	UIDall	D I	12	2.55	0.190	2.23	2.07	0.07	
	Unland		11	2.51	0.19	2.23	2.19	0.09	AD C
	Optanu Control Site		12	2.09	0.302	1.035	2.37	0.13	
	Control Site	CS	0	2.20	0.20	1.97	2.00	0.02	ЪС
C_{leaf}	Rural	В	12	47.2	0.69	46.3	48.9	0.9	А
5		DT	12	47.7	1.18	45.7	49.3	-0.04	А
	Urban	D	12	46.5	1.36	44.7	49.2	0.41	AB
		L	11	47.7	1.15	46.2	50.7	1.94	А
	Upland	DPF	12	47.6	1.20	46.3	49.8	0.92	А
	Control Site	CS	6	45.5	0.274	45.0	45.7	-1.47	В

Table 5.1: Leaf traits: Specific leaf area SLA $(cm^2 g^{-1})$, carbon and nitrogen % concentrations in 2mg leaf. Capital letters denote significant differences between sites for each species using Tukey's HSD which looks for differences in all pairwise means



Figure 5.1: Interaction between SPECIES and both SITE (DARTMOOR is UPLAND(DPF), BROADLEY is rural, DARTINGTON is rural, DIGGINIT is urban and LAIRA is urban) and ENVIRONMENT (upland rural, urban). Means analysed with mixed linear models to highlight the site specific nature of environmental effects on leaf trait, *SLA* (cm^2g^{-1}).



Figure 5.2: Interaction between SPECIES and both SITE (DARTMOOR is UPLAND(DPF), BROADLEY is rural, DARTINGTON is rural, DIGGINIT is urban and LAIRA is urban) and ENVIRONMENT (upland rural, urban). Means analysed with mixed linear models to highlight the site specific nature of environmental effects on leaf trait, CResult= C_{leaf} (mean % concentration).



Figure 5.3: Interaction between SPECIES and both SITE (DARTMOOR is UPLAND(DPF), BROADLEY is rural, DARTINGTON is rural, DIGGINIT is urban and LAIRA is urban) and ENVIRONMENT (upland rural, urban). Means analysed with mixed linear models to highlight the site specific nature of environmental effects on leaf trait, NResult= N_{leaf} (mean % concentration).

Table 5.2: Species and environment (Urban, Rural and Upland) effects and interactions on leaf traits.

-		~ ~				~			
Response:		SL	\mathcal{A}	N_{le}	eaf	C_{leaf}			
	Df	F	sig.	F	sig.	F	sig.		
SPECIES	1	411	***	408	***	9.38	**		
ENVIRONMENT	3	12	***	10.1	***	7.6	***		
SPECIES:ENVIRONMENT	3	6.86	***	1.82	ns	0.962	ns		
Residuals SLA	338								
Residuals CN_{leaf}^{1}	138								
$^{1}CN_{leaf} = C_{leaf}$ and N_{leaf} residual error									

ANOVA table: Leaf Traits

Table 5.3: Effect of soil treatment (POTS vs GROUND) on leaf traits (Species = SP, soil treatment = TREATMENT, Site = S (S nested within ENVIRONMENT). Significant levels, $p < 0.001 = * * *, p < 0.01 = * *, p < 0.05 = *, p \ge 0.05$ = not significant (ns) (C_{leaf} ns, data not shown).

ANOVA table: Soil effects on leaf traits.									
Response:		N _{leaf}							
	Df	F	Sig.						
SP	1	794	***						
S	5	23.8	***						
TREATMENT	1	14.3	***						
SP:S	5	4.65	***						
S:TREATMENT	4	6.24	***						
SP:S:TREATMENT	4	4.69	**						
Residuals	109								

|--|



(a) Q. ilex



(b) Q. robur

Figure 5.4: SLA Species differences in SITE (Rural site BROADLEY (B),CONTROL SITE (C), urban site DIGGINIT (D), upland site DARTMOOR (DPF), rural site DARTINGTON (DT) and urban site LAIRA (L) and soil treatment (field grown (GROUND) and potted (POT)) plus interactions (means $\pm 95\%$ CI).



(a) Q. ilex



(b) Q. robur

Figure 5.5: C_{leaf} : Species differences in SITE (Rural site BROADLEY (B),CONTROL SITE (C), urban site DIGGINIT (D), upland site DARTMOOR (DPF), rural site DARTINGTON (DT) and urban site LAIRA (L) and soil treatment (field grown (GROUND) and potted (POT)) plus interactions (means \pm 95% CI).



(a) Q. ilex



(b) Q. robur

Figure 5.6: N_{leaf} : Species differences in SITE (Rural site BROADLEY (B),CONTROL SITE (C), urban site DIGGINIT (D), upland site DARTMOOR (DPF), rural site DARTINGTON (DT) and urban site LAIRA (L) and soil treatment (field grown (GROUND) and potted (POT)) plus interactions (means \pm 95% CI).



Figure 5.7: SLA ($cm^{-2} g^{-1}$): Species differences with S and LEAF POSITION (LP) interactions to quantify sun/shade effects (means \pm 95% CI)



Figure 5.8: C_{leaf} (%): Species differences with SITE and LEAF POSITION (LP) interactions to quantify sun/shade effects (means \pm 95% CI)



Figure 5.9: N_{leaf} (%): Species differences with SITE and LEAF POSITION (LP) (lower, upper) interactions to quantify sun/shade effects (means ±95% CI)

shade leaves (Figure 5.9).

C_{leaf} .

There were significant effects of each factor on C_{leaf} , however no interactions occurred (Table 5.4). C_{leaf} was higher in *Q. ilex* than *Q. robur* and higher in upper leaves than lower (Figure 5.8). In *Q.robur* upper leaves at CS had significantly lower C_{leaf} than upper leaves at the other sites (although there were no differences between upper and lower leaves at this site. In *Q. ilex* C_{leaf} was higher in upper leaves at DT although this was not significant (Figure 5.8).

N_{leaf} .

 N_{leaf} was low in both species at the upland DPF in both upper and lower leaves (Figure 5.9). In *Q. robur* highest N_{leaf} was at rural DT in both upper and lower leaves and although N_{leaf} was reduced in lower leaves at L for this species upper leaves were comparable to DT values in *Q. robur* (2.6 ± 1.5 and 2.6 ± 1.6 (%)) (Figure 5.9). There were no significant differences between upper and lower leaves in *Q. robur* at any of the sites although upper leaves at rural B had the lowest N_{leaf} values for this species (2.09 ± 0.20). However, at the urban site D, N_{leaf} was significantly greater in upper than lower leaves for *Q.ilex* (1.36 ± 0.06 L, 1.61 ± 0.03 U

Table 5.4: Site and species differences in sun and shade leaves using leaf position (UPPER and LOWER) to ascertain sun and shade (includes interaction terms). Significant levels are p < 0.001 = ***, p < 0.01 = **, p < 0.05 = *, $p \ge 0.05 =$ not significant (ns).

Response:		SLA		N _{lea}	f	C_{leaf}		
	Df	F	Р	F	Р	F	Р	
SP	1	1189.75	***	883.85	***	18.23	***	
S	5	12.34	***	20.18	***	10.79	***	
LP	1	83.70	***	10.11	***	58.18	***	
SP:S	5	5.54	***	5.33	***			
SP:LP	1	40.62	***	4.31	*			
S:LP	5	4.41	**	2.11	ns			
SP:S:LP	5	4.61	**					
Residuals	338			146				

ANOVA table: Effect of Leaf Position on leaf traits

(%)(Figure 5.9).

 N_{leaf} was significantly higher in *Q. robur* than *Q. ilex*, higher N_{leaf} in upper leaves for *Q. ilex* compared to lower (Table 5.4).

5.4 Discussion.

Although both *SLA* and N_{leaf} were greater in *Q. robur* than *Q. ilex*, C_{leaf} was greater in *Q. ilex*. Differences between species and environment were very site specific, thus highlighting the difficulties in determining species responses when generalising urban environments. The upland site had lowest N_{leaf} in both species and highest *SLA*. In general *SLA* increased from urban to upland environments while N_{leaf} followed the opposite response and generally declined along the same gradient. However, for *Q. ilex*, the site with the highest N_{leaf} was the rural site DT, situated on the outskirts of Totnes town. Other studies also show that *SLA* is higher in sun than shade leaves, and in *Q. ilex* trees, *SLA* decreased with drought (Ogaya and Penuelas 2006).

It has been generalised that deciduous species tend to have higher N_{leaf} than evergreens, coupled with high *SLA* values (e.g. Evans and Poorter 2001; Villar et al. 2006) this has a negative correlation with *RGR* where faster growing species have lower N_{leaf} an higher *SLA* (Villar et al. 2006). However, for phylogenetically related species such as evergreen and deciduous oaks, this relationship does not always hold true and increased leaf mass ratios (*LWF_t*) in *Q. ilex* can cause this (Antunez et al. 2001). Environmental factors also disrupt this, such as soil fertility,
light availability and the effects of drought (Laureano et al. 2008).

 N_{leaf} is tightly coupled with light levels and A_n as much of N within a leaf is responsible for the formation of proteins (around 30%) (Meir et al. 2002) and the enzyme RuBisCO which is responsible for the carboxylation process (Fitter and Hay 2002). Further, $leafN_{mass}$ has been shown to decrease with increasing mean annual temperatures (MAT) coupled with irradiance. Shade leaves therefore tend to have lower mass based N_{leaf} and higher *SLA*. *Q. ilex* saplings tended to have lower mass based N_{leaf} in lower canopy leaves (shade), however, Meir et al. (2002) showed this trend in mature *Q. petrea* therefore the lack of foliage is likely to be responsible for this in *Q. robur*. *Q. ilex* is a Mediterranean evergreen and as such has a conservative growth pattern with regards to leaf traits (Valledares et al. 2000).

Q. ilex growing in the Mediterranean basin had higher N_{leaf} in relation to light despite no effect of nutrient status (Valledares et al. 2000) and also (Ogaya and Penuelas 2007*a*). This was true for my research at the upland site (where mean temperatures were 1-2 °*C* lower than the other sites), and to a lesser extent on the Dartmoor fringe where temperatures were slightly lower. Interestingly this site lies within the east Dartmoor rain shadow (Phillips and Mcgregor 2002) and here both species have higher *SLA* (lower *LMA*) in field grown saplings suggesting that in *Q. ilex SLA* responds to colder temperatures when associated with drought (Ogaya and Penuelas 2007*a*). In fact, *Q. ilex* has more leaves with higher *SLA* in order to maximise photosynthetic gain, when lower temperatures correspond with drought (Ogaya and Penuelas 2007*b*). *SLA* has greater correlation with A_{max} at lower canopy levels as within forest stands this part of the canopy is generally shaded and therefore leaves invest in light harvesting as opposed to photosynthetic enzymes (Meir et al. 2002). Ogaya and Penuelas (2006) also found that *LMA* (1/*SLA*) was higher in sun leaves as *Q. ilex* leaves were thicker with a smaller surface area than a deciduous oak, *LMA* also decreased with drought. Morphologically there is also a gradient from temperate to Mediterranean in *LMA* increase (Ogaya and Penuelas 2006).

In contrast leaves developed in high light environments lead to thick leaves with low *SLA* as leaves develop an extra layer of palisade cells in order to increase the number of chloroplasts and photosynthetic enzymes in order to enhance assimilation rates, of course this comes with a trade

off and light harvesting capabilities are reduced (Evans and Poorter 2001). This coupling means that changes in SLA affect RGR more in low light than high, and although N_{leaf} increased as *SLA* decreased and acclimation to low light results in a twofold increase in *SLA* compared with a 50% reduction in N_{area} . Therefore increasing *SLA* in response to low light is more effective for increasing carbon assimilation than re-allocating N_{leaf} (Evans and Poorter 2001).

5.4.1 Urban Environments.

Adaptations of plants to stress can include changes in morphology such as *SLA*. Decreasing *SLA* is a signature of 'schlerophllous' plants such as *Q. ilex* that are adapted for high temperatures accompany by periods of drought (Filippo 2008).

Generally C_{leaf} was lower in relation to the urban environment while N_{leaf} was higher. *Q. ilex* growing in Naples were found to increase concentrations of both leaf C_{leaf} and N_{leaf} over time even in control trees (Alfani et al. 2000). N_{leaf} from deposited NO_2 is taken up by leaves and rapidly incorporated into nitrogen compounds such as amino acids as a means of removing this gas from the atmosphere (Sparks 2009).

However the stimulation in growth caused by increased N deposition is mediated by water availability and a threshold usually occurs where N accumulation starts to have adverse effects on leaf performance and stomatal damage can occur (Abrams 2011). Exposure in my research saplings may not be long enough at the levels measured to induce adverse effects despite increased N_{leaf} at urban sites.

SLA increases with decreasing rainfall (Wright et al. 2004) and *SLA* can decrease with decreasing mean annual temperatures in evergreen broadleaves compared to a more conservative response in deciduous species (Wright et al. 2005). This thesis showed that decreasing *SLA* was more related to lower temperatures and higher nutrient supply, evidenced by the significantly higher *SLA* at the upland site pots. Similarly, Ordpnez et al. (2009) when analysing a global dataset, found low N_{leaf} and low *SLA* corresponded with low soil fertility and vice versa and was related to MAT (Ordpnez et al. 2009). Therefore response in leaf traits of the saplings measured within my research show evidence of complex intrinsic and extrinsic responses in both species, the Mediterranean evergreen increases N_{leaf} at urban locations however this is accented when

nutrient supply is good despite this species in generally having a more conservative response to nutrient supply (Valledares et al. 2000).

5.5 Conclusion.

Q. robur has greater *SLA* and N_{leaf} than *Q. ilex*. This is the general trend in deciduous and evergreen species. Usually greater *SLA* and N_{leaf} corresponds to faster growth in deciduous species. However, young saplings here have shown that this is not always the case. *SLA* seemed to decrease with increasing light in *Q. robur* while, in both species *SLA* increase with decreasing temperatures and in *Q. robur* higher soil nutrients. Low *SLA* is related to increased allocation to photosynthetic process in order to maximise photosynthesis. Therefore decreasing *SLA* with lower N_{leaf} may be a result of photo inhibition in low temperatures and high light environment often seen in Mediterranean oaks and manifest here in deciduous oaks on high nutrient soils.

Chapter 6

Productivity and Biomass Accumulation.

Whole tree allometry is a key area of climate change science historically rooted in commercial forestry science. Essentially, Kyoto Protocol requirements mean Countries need to routinely monitor their carbon stocks therefore providing a renewed interest in biomass determination. The literature contains information on specific allometric equations for species and locations, however, more general equations are also available although their usefulness is debated. This chapter looks at differences in total biomass (TB) and allocation to different root and shoot fractions between the two oak species studied. Survival, and relative growth rates of all saplings remaining until the end of the experiment were determined by analysing monthly growth data taken as height (h) and diameter at ground level (DAG) measurements and expressed as stem volume with the equation $V = HD^2$. An initial harvest was taken of a selection of one year saplings from the control site, these were harvested in September 2009 and initial allometic equations created using TB and HD² regression equations. A second harvest was completed at the end of the experimental period September 2011. Ten saplings for each site, species and condition were cut into relative foliage, branch, stem and root fractions and dry weight determined. Differences in TB between species and sites was determined using general linear models and allocation to different fractions was also analysed between species and sites.

Results showed that the evergreen oak not only had faster relative growth rates it also allocated a greater proportion of biomass to leaf and branch fractions in comparison to the deciduous species that allocated more to root and stem.

The evergreen *Q. ilex* competed better at sites where the English oak suffered from drought stress although at the upland site the deciduous leaf habit had the advantage and was able to increase overall biomass at this site to values comparable to *Q. ilex.*

However survival was also higher in the deciduous species at this site. Urban influences were also evident with the deciduous species having lower leaf weight fractions at the urban sites.

Evergreen broadleaved trees are known to increase allocation to foliage when young and this is a competitive strategy that seems to pay off in favourable environments. However as trees mature their relative allocation strategies change and relative leaf weight fraction can decrease to just 6 % of whole tree mass. The urban environment poses many stresses on oak productivity, water limitations and atmospheric pollutants combine to lower water use efficiency and leave them susceptible to long term stress.

6.1 Introduction.

Evidence suggests that European, temperate forests will maintain or increase in productivity with different climate change scenarios; in particular, UK forests have the potential to increase carbon sequestration (Broadmeadow and Matthews 2003). Current carbon sequestering rates for UK forests are estimated at up to 1.1 *GtC* yr^{-1} , however, this figure is variable and can depend on how data is collected (Nabuurs et al. 2007). In fact, although more recent eddy covariance¹ methods are useful for analysing ecosystem scale carbon assimilation across biomes, these methods are expensive and large errors can occur (Ohtsuka et al. 2009). Carbon inventory methods using tree allometry are routinely used to determine stem biomass for use in the forestry industry, where stem volume estimates, wood specific gravity, and root shoot ratios or biomass expansion factors (BEF) are used alongside regression to ascertain biomass (Matthews and Mackie 2006; West 2009). Although historically, allometry for different biomass fractions did not play an important part in forestry science, the growing importance of using branch and brash for commercial biomass and the National requirement of carbon accounting for climate change mitigation, mean the science of whole tree allometry is now an important study area (Zianis and Mencuccini 2004).

Further, in the wake of the Kyoto Protocol, Marrakech Accords and subsequent Land-use, Land Use Change and Forestry Guidance (LULUCFG) (IPCC 2003), the UK along with other Coun-

¹Eddy covariance technique uses infra red gas analyser (IRGA) situated on a tower. This constantly measures the flux of CO_2 between the atmosphere and the canopy. Wind-speed and direction are also monitored and correction factors involving atmospheric conditions used to find CO_2 assimilation rates

tries ratified by the Kyoto Protocol are obliged to monitor carbon stocks and changes at regular intervals (Broadmeadow and Matthews 2004). The Guidance gives advice on sampling protocols to ensure quality reporting of source/sink information including that for(de)(re)(a)fforestation (IPCC 2003).

6.1.1 Carbon Allocation and Relative growth rate.

Direct measurement of biomass alleviates the problem of determining respiration rates when assessing productivity at the smaller scale and can be used alongside NPP studies to ascertain relationships between carbon assimilation and subsequent sequestration. More information on below ground biomass is also desirable: the retrieval of root systems is difficult and time consuming and tends to introduces errors into carbon sequestration modeling (Clark et al. 2001). The destructive harvesting of mature trees is also extremely time consuming and undesirable, yet can be the only way of determining accurate biomass equations for a specific site as more general equations used for simplicity can introduce a wide error range (Clark et al. 2001). In addition to finding whole tree biomass, the relative allocation of carbon to roots, stems, branches and foliage can play an important part in overall carbon sequestration potential of a tree. Evergreen trees are associated with less fertile and drier sites than deciduous and are therefore expected to allocate more carbon to roots than foliage and for deciduous the opposite is true (Litton et al. 2007).

It is also important to note that allocation of assimilated carbon is tightly coupled with mycorrhizal status of a tree (Lunt and Hedger 2003). An exchange of nutrients occurs between plant and fungus and as well as enhancing mineral status within the plant (e.g. Lunt and Hedger 2003) and in some cases improving water use efficiency within the plant (Futai et al. 2008), however the cost to the plant is around 20% of assimilated carbon (Futai et al. 2008).

6.1.2 Allometry and Biomass Fractions.

The ratios of biomass fractions (branch, leaf and root) to stem volume are known as biomass expansion factors (BEF) and multiplication of stem volume by the BEF can provide an estimate for the biomass fraction (West 2009).

Another important aspect of allocation and productivity is ontogeny: For trees, allocation to leaf mass fraction(LMF) is higher in young saplings but as a tree matures its RGR in the stem mass fraction SMF increases as LMF decreases (Poorter et al. 2012). This is logical as a tree grows it stores non living carbon as structural support (Franco 1985) and this quantity is included in biomass estimates. In tree seedlings, the relative allocation patterns of biomass to different aboveground and below ground fractions can change with water, nutrient and temperature stresses (Poorter et al. 2012).

Standard biomass equations for different tree species are often used and there is a plethora of information in the literature on stand level biomass estimates using allometry (e.g. Muukkonen 2007). In the UK, at the national and regional scales the National Inventory of Woodland and Trees (NIWT) has been modified in order to honour the monitoring agreement. The latest Inventory, is due for completion in 2015 (Forestry Commission 2012*b*) and involves an extensive monitoring network. This network includes mapped information on soils, species composition, productivity, structure and management of UK forests. This is being 'backed up' by three more monitoring networks at various detail and scales (Forestry Commission 2012*a*). Crucially, assessment of productivity and carbon sequestration potential in species in relation to environment is important not only to help monitor stocks already established but to assess productivity of non-native species that may become more frequent with future climate change. Further, the increased occurrence of damaging pathogens in UK woodlands mean that native species are becoming increasingly at risk and may not have the ability to out-compete those species moving in.

Generally biomass equations take the form of the power function:

$$M = aD^b \tag{6.1}$$

where M is mass and D, the diameter at breast height, while a and b are scaling coefficients usually determined via least squares linear regression on log transformed data (to avoid issues with normality and misuse of power functions) (Zianis and Mencuccini 2004). Although information regarding different scaling coefficients and variations of the above equation is abundant in the literature, the values are very much species and site specific, despite attempts to simplify and generalise equations for use with different forest types (Zianis and Mencuccini 2004).

Volume estimates, using height and diameter at breast height (*DBH*), for trees with DBH > 7cmare the normal method of estimating biomass content in commercial forestry (Matthews and Mackie 2006). For seedlings and young saplings diameter at ground level (DAG) can be used as an alternative to DBH and, incorporated into a stem volume equation as HD^2 , has been proven to be useful for measuring biomass in young hardwood species in Central America (Williams and McClenahen 1984) (see eqn 6.2). Although Williams and McClenahen (1984) and others (e.g. Zianis and Mencuccini 2004) suggest removal of H from the equation to simplify fieldwork, other research suggests errors involved in estimating biomass without H can be too large to warrant its removal (e.g. Ketterings et al. 2001). Additionally, for seedlings and young saplings, height may be an important variable when measuring growth as height growth tends to be more rapid during early growth stages compared to that of a mature tree (Grime 2002), indeed, Canadell et al. (1988) suggest it is useful addition in biomass estimation. Therefore for the purpose of finding allometric equations for Q. ilex and Q. robur growing at six different sites at various rural, urban and one upland location, the logged form of the power function D^2H was used (in order to homogenise variances) and using DAG as a substitute for DBH). The natural Log_e (*Ln*) was used and once allometric equations determined, a correction factor (CF) added in order to increase reliability of the equation when back transforming data into the power function. When mean square errors of the regression are small, equation 6.5 has been proven to be a reliable CF for biomass regression (Baskerville 1972; Mascaro et al. 2011).

$$V_i = D^2 H \tag{6.2}$$

$$Log_e Y = a + bLog_e(D^2H) \tag{6.3}$$

$$Y = a(D^2H)xCF \tag{6.4}$$

$$CF = e^{(MSE/2)} \tag{6.5}$$

where Y is biomass and *a* and *b* are scaling coefficients and CF is the correction factor and MSE is the mean square error of the regression.

The experiments in this chapter were designed to assess productivity and biomass accumulation of each species in relation to the relative urban, rural and upland environments. Actual growth and relative growth rates (RGR) have been determined for growing season and deciduous dormant season, overall growth rate for the 30 month period and also a the smaller scale of approximately growth every two months. Growth was measured as HD^2 and \overline{RGR} has been determined using the standard \overline{RGR} equation presented in Hunt (1982) (see eqn. 6.6, page 97, this chapter). To standardize this equation and reduce error, data were logged before means were taken and not after (see Hoffman and Poorter (2002)). Allometric equations for biomass estimates were determined using final HD^2 and final harvest (TB and AGB) with least squared regression on Log_e transformed data. Specific density was not found and therefore has not been used as an additive parameter due to the young age of the trees; average specific density in the literature may not be accurate. Specific gravity for mature oaks can range from 0.4-0.8 with an average found for white oak growing in the Southern Appalachian mountains of 0.6 (Clark III and Schroeder 1986).

General working hypotheses for this chapter are:

- 1. Working hypotheses
 - 1.1. For *Q. ilex*, mean \overline{RGR} and actual growth is significantly faster than *Q. robur*.
 - 1.2. For *Q. ilex* growth is faster than *Q. robur* at each of the six sites.
 - 1.3. Is final total biomass and carbon content greater in *Q. ilex* than *Q. robur* and is this true for each site.
 - 1.4. Finally is there any significant affect of urban and rural as a site character or are any differences site specific.

6.2 Methods.

Initial height (H) and ground level diameters (DAG) were recorded at each site after planting in February 2009, then height and diameters recorded every 1 - 2 months until final harvest in August/September 2011 (see chapter 2 section 2.3 on page 21 for detailed sampling protocol). Sapling deaths within the first (Feb 09- Nov 09) second (Nov 09- Nov 10) and third (Nov 10-Aug 11) season were recorded and shown as percentage losses. Dead saplings were deemed those showing no evidence of green shoots at the end of the favourable season (1st year losses were determined at the beginning of the second favourable season in case of re-growth). Actual and *RGRs* are shown using data from those saplings surviving the entire 30 month growth period (see 6.3).

6.2.1 Relative growth Rates.

The natural $log_e(Ln)$ was used to normalise data as $Ln(HD^2)$. Subsequently, $Ln(HD^2)$ was used to signify growth from all sites and used to compare growth rates using the standard equation for \overline{RGR} but revised to substitute $ln(HD^2)$ in place of weight (W) (see equation 6.6). In addition to this, Ln values of HD^2 were used to compare RGR between Feb 2009 - Sep 2009, Sep 2009 - Feb 2010, Feb 2010 - Sep 2010, Sep 2010 - Feb 2011, Feb 2011 - Aug 2011 (final harvest).

$$\overline{RGR} = (\overline{Ln(HD^2)_2} - \overline{Ln(HD^2)_1})/(2t - t)$$
(6.6)

During September 2009 (7 months after planting) a small number of saplings were harvested (*harvest*₁). Due to the number of losses in the first growing season this sample was taken from a sub-sample of original planted saplings at the control site and not from the individual sites (again see chapter 2 section 2.3 on page 21 for detailed sampling protocol). Due to this, *harvest*₁ data was used to provide base line allometric equations for each species but was not included in any *RGR* analyses (HD^2 was used instead - see above). Approximately 10 saplings per treatment (depending on surviving saplings at each site) were harvested at the end of the experiment September 2011 and total biomass of each fraction determined using the methods described in chapter 2. Final harvest data was used to determine overall carbon sequestered over the 30

month period and also to see if there were any differences between species in allocation between the fractions at each site. Finally, least squares regression was used to compare relationships between HD^2 with AGB and TB and site specific allometric equations established.

Statistical analyses were only performed on those saplings that had survived throughout the entire experiment. Descriptive statistics showing initial H and DAG measurements and statistical differences between sites for each species and parameter are shown (under Group), followed by final biomass as separate fractions and above ground biomass (AGB) and total biomass (TB), H and DAG measurements. Comparisons between sites for final biomass are shown using naturally logged data (ln) data, for homogenised variances. General linear models (GLM) were performed on the logged data to look for the effects of site and species plus interactions on growth parameters. Allocation to the different biomass fractions were analysed for effects of different sites and species to ascertain the importance of each fraction to overall biomass at each site (see appendix B for residual plots).

Statistics software used was 'R' for GLM's and simple least squared regression for allometric equations (R 2009), descriptive statistics were generated with Minitab16 (Minitab (R) 2010).

6.3 Results.

6.3.1 Survival.

The effect of the sampling design meant that there were more losses at the outer edge of the blocked saplings, although this was most evident in *Q. ilex* saplings at the exposed sites at rural, B on the Dartmoor fringe and urban L, on the south facing slope near the city center. Most losses occurred in the first growing season while saplings were still acclimatising and *Q. ilex* lost more saplings from the upland site than *Q. robur* in both soil conditions and also lost more saplings over the three seasons at the rural site B and the urban site D in potted trees (Table 6.1). Although *Q. robur* had greater survival than *Q. ilex* in general, at the exposed urban site L, field grown *Q. robur* suffered the greatest loses, due to exposure and increased drainage at the top of the slope increasing the occurrence of drought (Table 6.1).

Table 6.1: Cumulative percentage of lost saplings over 3 growing seasons for Upland site (DPF), rural sites at Dartington (DT) and Broadley (B) and urban sites at Digginit (D) and Laira (L).

Character	Site	Condition	Sep-09	Sep-10	Sep-11	Sep-09	Sep-10	Sep-11
			Q	. <i>ilex</i> % lo	SS	<i>Q</i> .	robur % l	oss
Gl	ROUNE)	%	%	%	%	%	%
Upland	DPF		15	35	50	10	15	15
Rural	В		25	25	30	35	35	35
	DT		15	20	20	40	40	40
Urban	D		35	40	40	30	40	45
	L		45	45	45	80	80	80
	POTS							
Upland	DPF		50	55	65	30	40	40
Rural	В		40	50	50	15	25	35
	DT		25	25	25	15	20	20
Urban	D		45	50	50	25	30	35
	L		50	50	55	0	0	5
Control site	CS		15	20	20	30	30	30

Accumulative percentage losses (%) of Saplings over three years

6.3.2 Initial height and diameters.

One year saplings were checked for significant differences, in *H* and *DAG*, between sites at start of growth period (February 2009). Data was log transformed (using the natural log *ln*)and Levenes test was used to determine homogeneous variances. Levenes test showed that variances in both ln(H) and ln(DAG) datasets were homogeneous (T = 0.65p = 0.751; 1.12, p = 0.344). One-way ANOVA showed no significant differences between sites for mean *DAG* or mean *H* for *Q. robur* (p > 0.05). However for *Q. ilex*, initial heights and diameters were more variable and *Q. ilex*, mean *DAG* was significantly lower at the rural site B ($F_{(4,77)} = 8.64, p < 0.001$). Subsequently, Tukeys test showing significant differences in *H* between sites B and CS ($F_{(4,77)} = 3.04, p < 0.05$) (CS saplings were generally smaller in *H*) (see Table 6.2).

6.3.3 Actual Growth and Relative Growth rates.

Figure 6.1 shows actual growth for saplings over the entire experimental period, using stem volume HD^2 as the measure of growth. At the control site (CS) *Q. ilex* had the greatest overall growth (Figure 6.1 but see appendix E, section E.1 for individual growth graphs) (around 2000g

Table 6.2: Descriptive statistics for height (H) and diameter at ground level (DAG) measurements taken at planting in February 2009 means, stadnard deviations (StDev, minimum and maximums given. Different letters denote significant differences between sites; Rural sites, Broadley B, Dartington DT, control site CS, upland site DPF and urban sites Digginit D and Laira L.

Descriptiv	ve statistics for	initial h	eight	(H) and di	iameter a	t ground leve	(DAG)meas	urements
Variable		SITE	Ν	Mean	StDev	Minimum	Maximum	Group
				Q. rob	pur			
Н	RURAL	В	19	209.05	37.91	159	293	А
		DT	19	203.21	28.19	167	260	А
	CONTROL	CS	7	162.3	46.9	105	223	А
	UPLAND	DPF	29	217	42.55	168	360	А
	URBAN	L	15	223.6	54.4	175	390	А
		D	20	201.9	66.4	105	418	А
DAG	σισλι	D	10	1 3 1 3	0.01	2 815	634	٨
DAU	KUKAL	D DT	19	4.313	0.91	2.015	0.34 5.86	A
	CONTROL		19	4.579	0.700	2.42	5.05	A
			20	4.551	0.970	5.42 2.19	J.95 7 1	A
	UPLAND	DPF т	29 15	4.005	0.945	5.16	/.1	A
	UKBAN		15	4.303	0.889	3.04	0.18	A
		D	20	4.751	1.034	2.04	10.09	A
				Q. ile	2 <i>x</i>			
Н	RURAL	В	19	261.47	39.01	187	350	А
		DT	17	248.4	42.9	185	330	AB
	CONTROL	CS	8	205.5	17.25	186	230	В
	UPLAND	DPF	17	259.94	37.56	200	335	А
	URBAN	L	15	255.4	30.86	200	305	AB
		D	14	235.4	52.3	129	280	AB
DAG	RURAL	В	19	3.692	0.753	2.62	5.08	С
		DT	17	4.419	0.818	3.3	6.49	AB
	CONTROL	CS	8	5.158	0.45	4.54	5.76	А
	UPLAND	DPF	17	4.691	0.77	3.29	6.34	AB
	URBAN	L	15	4.225	0.587	3.17	5.45	BC
		D	14	5.048	0.501	4.37	5.76	А

higher average stem volume than CS Q. *robur*) saplings. Rural site B on the Dartmoor fringe showed the greatest growth in Q. *ilex* in both potted and field grown saplings, however at the more favourable rural site DT on the outskirts of Dartington, both species have higher HD^2 in field grown saplings than pots (Figure 6.1D and E.4). In fact, when actual height and diameters were used to model growth, Q. *ilex* grew faster at all sites except the upland site on the high moor and in planted saplings at DT where Q. *robur* was able to maintain similar growth (Figure 6.1).

Relative growth rates show that for potted saplings, *Q. ilex* maintained higher total *RGR* at rural site B and at CS (Figure 6.21). However this advantage is greater in field grown saplings (Figure 6.22). total \overline{RGR} is only higher in *Q. robur* at the upland site DPF (Figure 6.2). Appendix E also gives \overline{RGR} data for seasonal growth and shows that *Q. ilex* has greater \overline{RGR} during the first growing season (Feb 09 - Oct 09), however, after establishment (one year), *Q. robur* increased its growth rate at all sites until \overline{RGR} was similar to *Q. ilex* during the final growth period (Apr 09 - Aug 11) at all sites (Figures E.7 to E.12 in appendix E). In fact field grown saplings generally had faster \overline{RGR} than potted saplings although for *Q. robur* planted at the upland site this was not true. Overall, \overline{RGR} were lower at the rural site B and upland DPF for *Q. robur* and higher at D and DT. \overline{RGR} in *Q. ilex* remained constant across sites apart from the upland site where \overline{RGR} was as low as 0.004 cm^3t^{-1} (Figure 6.2) (Appendix E gives figures for monthly and seasonal \overline{RGR} s).

6.3.4 Final growth and Biomass.

TB was greater in *Q. ilex* containing over twice as much biomass than than *Q. robur* (284.2 \pm 259.85; 119.5 \pm 134.3, mean \pm SD) and thus proving the first hypotheses. There were no differences between potted (P) and field grown (G) saplings for the majority of sites, however, P had higher TB than G at the upland site and lower TB at the rural site DT (see tables 6.3, E.1 and E.2). *Q. robur* maintained similar TB at the upland site, where both species had reduced biomass (33.62 \pm 53.6; 39.7 \pm 59.5 for *Q. ilex* and *Q. robur* respectively). The highest TB was found at DT and D, both species maintained their highest growth and biomass at these sites and *Q. ilex*, P experiencing the highest TB at urban D site (632.72 \pm 374.96), twice as much as the



Figure 6.1: Actual growth; A: Control site CS, B: Upland site DPF, C: Rural site B, D Rural site DT, E Urban site D, and F Urban site L (monthly means ± 95 % CI).



Figure 6.2: Relative growth rate (RGR) for all saplings between planting in Feb 09 and final harvest Aug 11 using eqn 6.6. Rural sites, Broadley B, Dartington DT, control site CS, upland site DPF and urban sites Digginit D and Laira L (means ± 95 % CI).

Table 6.3: Species, site and soil treatment (pots (P) or field grown (G) saplings) effects on total biomass, Degrees of freedom for factors and residual errors (Df), F statistic and significance levels given as (sig. p<0.05 '*', p<0.01 '**' p<0.001 '***' and p ≥ 0.05 not significant 'ns')

ANOVA table: Total Biomass (T	TB)		
Variable		TB (g	g)
Factors	Df	F	sig.
SITE	4	84.8514	***
SPECIES	1	45.0447	***
TREATMENT	1	7.3847	**
SITE:SPECIES	4	6.8617	***
SITE:TREATMENT	4	7.2169	***
SPECIES:TREATMENT	1	1.0208	ns
SITE:SPECIES:TREATMENT	4	0.6284	ns
Residuals	162		

G saplings for this species (Tables E.1 and E.2). Interestingly, TB in *Q. robur* at this site was similar for both P and G saplings and along with DT, D was *Q. roburs* better site. However, DT field grown saplings had the greatest TB (327.842 ± 186.524) in *Q. robur*.

Both species had lower biomass at rural B and urban L although again *Q. ilex* is still over twice as much as *Q. robur* (Table E.1 and E.2). There is a huge amount of variation in the data, and differences in biomass is very site specific, to show this, when data was analysed according to site character where rural and urban sites were analysed together. There was a significant effect on TB from character and species (p < 0.05) (data not shown). However, this was due to the reduced biomass at the upland site and there were no significant differences between rural and urban sites. Therefore data shown is for SITE, SPECIES and interactions for G and P saplings (Tables E.1 and E.2).



(a) Environmental differences in total biomass (TB g) for upland, rural and urban environments.



(b) Site differences in total biomass (TB g)

Figure 6.3: Environmental and site specific differences in total biomass for 1) potted and 2) field grown saplings (means ± 95 % CI). Rural sites, Broadley B, Dartington DT, control site CS, upland site DPF and urban sites Digginit D and Laira L.

Table 6.4: ANOVA table for biomass and allocation, all variables analysed in *ln* form; Total biomass *TB* (g); root weight fraction (*RWF_t*) leaf weight fraction (*LWF_t*), stem weight fraction (*SWF_t*) and branch weight fraction (*BWF_t*) analysed as percentage of total biomass. Factors are, SITE: upland DPF, rural B and DT and urban D and L; TREATMENT: soil treatment, POT (G) or field grown (G). Significance values; sig. p<0.001 '***', p<0.01 '**', p<0.05 '*', p \geq 0.05 not significant 'ns'.

Response variable		ln(T	'B)	ln(RV	VF_t)	ln(LW	VF_t)	ln(SW	(F_t)	ln(BV	VF_t)
FACTOR	Df	F	sig.	F	sig.	F	sig.	F	sig.	F	sig.
SITE	4	84.85	***	17.14	***	3.15	**	10.12	***	37.75	***
SPECIES	1	45.04	***	38.10	***	20.83	***	0.27	ns	40.73	***
TREATMENT	1	7.38	**	12.02	***	2.13	ns	16.60	***	3.81	ns
SITE:SPECIES	4	6.86	***	2.47	ns	2.03	ns	3.33	**	5.02	***
SITE:TREATMENT	4	7.22	***	1.53	ns	0.60	ns	0.81	ns	5.73	***
Residual error	162										

6.3.5 Allocation to Biomass Fractions.

Q. robur allocated a greater percentage of its total biomass (TB) to the root fraction (*RWF_t*) (37.31 ± 8.91; 47.62± 5.94 g) (p < 0.05) and also allocated more to stem (*SWF_t*) than *Q. ilex* (Table E.1 and Table 6.4). However, *Q. ilex* branch and foliage allocation (*BWF_t* and *LWF_t*) increased overall biomass in this species (Tables E.1 and E.2) (*Q. ilex* allocated 24.8 ± 8.91 g compared to 18.03 ± 5.94 in *Q. robur*) to *LWF_t* and *BWF_t* combined. *Q. robur* allocated just over half the amount of its *LWF_t* to *BWF_t* whereas, *Q. ilex* allocated 2/3*rd* of its *LWF_t* to *BWF_t*. Again site specific differences in allocated greater biomass to the *RWF_t* at the urban site L. As expected greater *RWF_t* occurred in P, however, no interactions occurred with this factor (Tables E.1 and E.2). *LWF_t* was higher in *Q. ilex* (See appendix E, tables E.1 and E.2 for descriptive statistics and for more information on differences see Table 6.4 and appendix C for interaction plots).

6.3.6 Allometric Equations for Biomass Estimation.

The following tables 6.5 to 6.9 give regression coefficients, associated errors and adjusted R^2 and include the CF for reverting estimated biomass from logged to original form. Table 6.5 gives information for *harvest*₁ regressions, and Table 6.6 on page 108 gives information on equations generated on control site and grouped datasets for *harvest*₂. Table 6.7 on page 109

gives equations for upland site *TB* and *AGB* and tables 6.8 and 6.9 on pages 110 and 111 give information from regression at rural and urban sites for *harvest*₂.

Appendix E, Figures F.1, F.2, F.3 and F.4 on pages 206,207, 208 and 209 show HD^2 TB and AGB scatter for individual factors and for species.

6.3.7 Initial Harvest.

Using ANCOVA in R to determine the added effects of species to the linear regression in initial harvest data, no extra significance was brought by analysing the species separately, (p>0.05), therefore only regression equations for combined species are given (Table 6.5).

Significant correlations are evident between coefficient of the slope of the regression and the coefficient of the independent variables for AGB however for TB the coefficient of the independent variable dose not meet the assumptions for a significant correlation at the 95% confidence interval. However the overall regression is significant and has some predictive power despite a low adjusted R^2 (69%) (Table 6.5).

6.3.8 Final Harvest.

For the final *harvest*₂, allometric equations for total dataset and species datasets show significant slope coefficients (*a*) and also independent variable coefficient *b*, adjusted R^2 for grouped data was between 84 and 89 % for TB and AGB data (Table 6.6).

When analysed separately for each factor level, predictive power of the equations is reduced for control site saplings (Table 6.6). *Q. robur ad jR*² only 49.3%. The sites where trees were the most stressed, equations seemed to lose predictive power with *Q. ilex* saplings at DPF upland site showing very little individual allometry (Table 6.7). Interestingly despite HD^2 being generally better at predicting *AGB*, at the upland site it was better at predicting *TB*, probably due to the increased *RWFt* at this site. In some cases reduced sapling numbers led to high adjusted R^2 values, however, due to the reduced sample numbers at these sites these equations are not robust enough (e.g. LRG, Table 6.9). In most cases *Q. robur* pots showed lowest prediction (adj R^2 values less than 60%), however, contrary to this upland and DT had good predictive power (*ad jR*² > 90%). HD^2 was better at predicting TB in field grown G saplings, although for *Q*.

Table 6.5: Multiple regression on initial harvest data in September 2009 for aboveground biomass (AGB) total biomass (TB), scaling coefficients *a* and *b*, variable SE, MSE and CF shown. Significant levels are; p≥0.05 'ns', p<0.05 '*', p<0.01 '**', p<0.001 '***'

Growth parameter	Coefficient symbol	value	Variable SE	sig	MSE ResSE ²	$CF e^{(MSE/2)}$	<i>ad jR</i> ² (%)	overall sig.
AGB	a b	-2.360 0.860	0.320 0.061	*** ***	0.133	1.069	91.40	***
ТВ	a b	-1.063 0.740	0.570 0.109	ns ***	0.428	1.239	69.40	***

robur DPFRP had better predicitive power than DPFRG reduced R^2 values (Tables 6.7, and 6.8, for scatterplots see Figures F.1 and F.2).

6.4 Discussion.

6.4.1 Relative growth rates and biomass allocation.

Relative growth rate.

My research has shown that in well watered and mild conditions such as that found at the control site. The evergreen oak Q. *ilex* has the advantage over the deciduous Q. *robur* in terms of faster \overline{RGR} . However, this took into account the deciduous winter period when Q. *robur* was not in leaf. Looking at \overline{RGR} on a two monthly basis the English oak increased its \overline{RGR} during spring, summer and in the final year in early spring while not in leaf. This growth strategy is common in ring porous oaks, where radial stem growth is added in preparation for new leaves and early wood formation of large transport vessels (Barabaroux and Beda 2002; Gieger and Thomas 2002). In the past it has been generalised that deciduous species have faster RGR than evergreens due to generally higher carbon assimilation rates and higher *SLA* (e.g. Wright et al. 2004). However, when taking phylogenetic relationships into account Antunez et al. (2001) found that this was not always true, and in closely related evergreen and deciduous oak species, the evergreen Cork oak (*Q. suber*) had similar *RGR* as deciduous *Q. robur*. This was attributed to the greater *LMF* in relation to its lower *SLA* which in turn masked differences in the ratio between leaf area and plant mass; leaf area ratio (*LAR*); In fact, in agreement with the general trend, the same study showed *Q. ilex* subsp. *rotundifolia* had lower *RGR* than *Q. robur* (13.1).

Table 6.6: Allometric relationship between HD^2 and TB/AGB. Variables shown, Complete dataset, individual species complete set and Control site groups Showing; Variable Std error, Mean Sq Error (residual) (MSE) and Correction factor CF for allometric regressions (significance levels shown are p<0.001 '***', p<0.01 '**', p<0.05 '*', p>0.05 'ns'.). All control site saplings = 'CS', *Q. ilex* control site saplings = 'CIP' and *Q. robur* control site saplings = 'CRP'.

	C	ONTRO	L AND GR	OUPS			
Sample	Coefficient		Variable	sig	MSE	CF	ad jR ²
	symbol	value	SE		$ResSE^2$	$e^{(MSE/2)}$	(%)
		TOTA	L BIOMAS	S			
All sites/species	a	-0.911	0.164	***	0.223	1.118	85.98
	b	0.761	0.022	***			
All Q. ilex	а	-0.888	0.281	**	0.236	1.125	84.15
	b	0.776	0.036	***			
All Q. robur	а	-0.662	0.190	***	0.176	1.092	86.90
	b	0.707	0.027	***			
CONTROL Site							
CS	а	2.891	1.252	*	0.077	1.04	86.5
	b	0.687	0.072	***			
CIP	а	8.603	3.981	ns	0.047	1.02	21.21
	b	0.372	0.219	ns			
CRP	а	0.979	3.795	ns	0.118	1.06	64.8
	b	0.803	0.232	*			
	AI	BOVEGR	OUND BIC	OMASS	5		
All sites/species	а	-2.230	0.177	***	0.258	1.138	87.12
	b	0.860	0.024	***			
All Q. ilex	a	-2.061	0.298	***	0.265	1.142	85.50
	b	0.865	0.038	***			
All Q. robur	а	-1.958	0.191	***	0.177	1.093	89.30
	b	0.795	0.027	***			
CONTROL Site							
S	а	-0.626	1.739	ns	0.149	1.077	83.57
	b	0.851	0.100	***			
CIP	a	5.009	2.884	ns	0.025	1.012	60.21
	b	0.541	0.159	*			
CRP	a	-3.163	6.313	ns	0.327	1.178	49.29
	b	1.006	0.385	*			

Regression coefficients: ln(M) = a + bln(HD2)

 \pm 2.7; 19.2 \pm 2.2 mg g⁻¹) (Antunez et al. 2001). A later study, also confirmed higher *RGR* in deciduous versus evergreen oaks, where *LAR* was higher for *Q. ilex*, and *SLA* was higher in the deciduous *Q. faginea*, the deciduous species having twice as much biomass at end of the growth period (Ruiz-Roberto and Villar 2005).

In favourable environments Q. robur was able to increase \overline{RGR} to rates comparable and in some

Table 6.7: Allometric relationship between HD^2 and TB/AGB for Upland data.Showing; Variable Std error, Mean Sq Error (residual) (MSE) and Correction factor CF for allometric regression equations (significance levels shown are p<0.001 '***', p<0.01 '**', p<0.05 '*', p>0.05 'ns'). All upland site saplings = 'DPF', all *Q. ilex* upland site saplings = 'DPFI', all upland site *Q. robur* saplings = 'DPFR', for field grown saplings a 'G' is added to the sample code and for potted saplings 'P'.

Sample	Coefficient		Variable	sig	MSE	CF	$ad jR^2$
	symbol	value	SE		$ResSE^2$	$e^{(MSE/2)}$	(%)
	-	TC	TAL BION	IASS			
DPF	а	2.70	1.39	ns			
	b	0.670	0.089	***	0.395	1.22	55.2
DPFI	а	10.37	3.758	*			
	b	0.179	0.245	ns	0.671	1.40	-2.99
DPFR	а	-0.146	0.819	***			
	b	0.845	0.052	***	0.101	1.05	90.3
DPFIP	а	13.9	12.178	ns			
	b	-0.032	0.802	ns	1.13	1.76	-16.6
DPFIG	а	8.349	2.324	**			
	b	0.291	0.150	ns	0.21	1.11	25.6
DPFRG	а	4.027	1.839	*			
	b	0.566	0.122	***	0.09	1.05	58.0
DPFRP	а	-1.184	1.175	ns			
	b	0.910	0.071	***	0.08	1.04	93.7
			~~~~~				
		ABOVE	GROUND	BIOM	ASS		
DPF	a	1.50	1.507	ns	0.466	1.262	53.30
	b	0.701	0.097	***			
DPFI	а	9.56	4.05	*	0.781	1.478	-3.23
	b	0.187	0.264	ns			
DPFR	a	-1.543	0.927	ns	0.129	1.067	88.94
	b	0.887	0.059	***			
DPFIP	а	14.957	12.906	ns	1.268	1.885	-16.07
	b	-0.150	0.850	ns			
DPFIG	a	7.319	3.020	ns	0.350	1.191	16.66
	b	0.315	0.195	ns			
DPFRG	а	2.578	2.169	ns	0.127	1.066	51.52
	b	0.611	0.144	**			
DPFRP	а	-2.256	1.343	ns	0.102	1.052	92.30
	b	0.934	0.081	***			

Regression coefficients: ln(M) = a+bln(HD2) for Upland site (DPF)

cases higher than Q. *ilex* where temperatures and soil nutrients were favourable (Figures E.7 to E.12 in appendix E give  $\overline{RGR}$ s for saplings growing in two monthly growth increments). The increased  $\overline{RGR}$  seen during the last growth phase, includes the most favourable season (April 2011 - August 2011) and confirms that growth in deciduous oak was faster than Q. *ilex* between early spring and summer at sites where conditions were favourable. However, as Q. *ilex* was

48.3	1.10	0.184	*	3.28 0.176	5.225 0.538	56.5	1.06	0.123	ns ***	2.69 0.144	6.09 0.513	a b	DTRG
92.12	1.030	0.060	* * * *	1.182 0.065	3.053 0.633	84.49	1.04	0.085	* * * * *	1.41 0.078	5.65 0.520	a b	DTRP
83.61	1.020	0.040	ns ***	2.461 0.129	-1.020 0.885	84.49	1.01	0.029	ns ***	2.099 0.110	1.257 0.779	a b	DTIG
97.40	1.024	0.047	* * * *	1.224 0.067	-3.324 1.004	95.41	1.03	0.067	ns ***	1.459 0.080	-0.871 0.894	а b	DTIP
76.55	1.081	0.157	ns ***	1.570 0.085	2.802 0.658	75.94	1.06	0.116	* * * * * *	1.349 0.073	5.14 0.557	а b	DTR
96.01	1.020	0.040	* * * * *	0.944 0.050	-3.040 0.990	94.75	1.02	0.040	ns ***	0.951 0.051	-0.352 0.864	а b	DTI
82.85	1.079	0.151	ns ***	1.171 0.063	-0.058 0.822	82.37	1.06	0.115	* * * * * *	1.02 0.055	2.543 0.703	a b	DT
58.96	1.114	0.216	ns ***	2.511 0.154	4.055 0.574	52.31	1.09	0.167	* *	2.21 0.136	6.897 0.447	a b	BRP
82.08	1.05	0.092	ns ***	2.047 0.128	0.248 0.787	75.93	1.05	0.096	ns **	2.080 0.130	2.95 0.668	a b	BRG
77.52	1.03	0.067	ns ***	2.94 0.166	-1.76 0.938	78.79	1.03	0.055	ns ***	2.66 0.150	-0.258 0.879	а b	BIG
91.11	1.028	0.056	ns ***	1.785 0.098	-1.237 0.893	78.73	1.05	0.08976	ns ***	2.267 0.125	3.07 0.689	а b	BIP
68.17	1.099	0.189	ns ***	1.793 0.111	1.861 0.698	61.45	1.09	0.171	* * * *	1.705 0.106	4.626 0.575	а b	BR
82.75	1.038	0.075	ns ***	1.611 0.090	-0.109 0.839	80.12	1.03	0.067	ns ***	1.530 0.085	2.346 0.731	а b	BI
88.8	1.08	0.151	ns ***	0.887 0.052	-1.15 0.891	86.5	1.07	0.135	* * * *	0.837 0.049	1.78 0.757	a b	в
							RURAI						
ad jR ² (%)	MASS CF e ^(MSE/2)	MSE <i>ResSE</i> ²	sig	ABOV Variable SE	value	ad jR ² (%)	CF e ^(MSE/2)	MSE ResSE ²	sig	Variable SE	value	Coefficient symbol	Sample
	2								$HD^2$ )	= a + bln(H)	S: $ln(M)$ =	on coefficient	Regressi
				AL. Vari- llometric ). All ru- all rural ed to the	<ul> <li>CF for al</li> <li>P &gt; 0.05</li> <li>or 'BI',</li> <li>G' is add</li> </ul>	bles show ion factor < 0.05, <i>ns</i> gs = 'DTI gs a ' plings a '	/ <i>TB</i> . Varia and Correct < 0.01, * <i>p</i> < site sapling d grown sa	² and AGB (al) (MSE); (001, ** p ). <i>ilex</i> rural 3R', for fiel 'P'.	een $HL$ (residu :* $p < 0$ 3', all $\zeta$ R' or 'I aplings	an Sq Error ons (sig: ** ons (sig: ** ong = 'DT' ings = 'DT' for potted s	tric relatic l error, Me on equatic saplings = <i>robur</i> sapl code and	6.8: Allomet able Std regressia ral site site 2. 1 sample	Table

# DEMELZA CARNE

$6.9$ : Allometric relationship between $HD^2$ and $AGB/TB$ . Variables shown: URBAN. Vari-	able Std error, Mean Sq Error (residual) (MSE) and Correction factor CF for allometric	regression equations (sig: $***p < 0.001, **p < 0.01, *p < 0.05, nsp > 0.05$ ). All ur-	ban site saplings = 'D' or 'L', all Q. <i>ilex</i> urban site saplings = 'DI' or 'LI', all urban	site Q. robur saplings = 'DR' or 'LR', for field grown saplings a 'G' is added to the	sample code and for potted saplings 'P'.
Table 6.9: All	abl	reg	baı	site	sar

Regression coefficients:  $ln(M) = a + bln(HD^2)$ 

		).L	UIAL BIU	MASS					ABOV	EGRO	UND BIUI	MASS	
Sample	Coefficient		Variable	sig	MSE	CF	ad $jR^2$		Variable	sig.	MSE	CF	$adjR^2$
	symbol	value	SE		$ResSE^2$	$e^{(MSE/2)}$	(%)	value	SE		$ResSE^2$	$e^{(MSE/2)}$	$(0_{0}^{\prime\prime})$
						URBAD	7						
D	a	0.787	1.666	ns	0.170	1.09	69.96	-1.844	1.968	su	0.237	1.126	68.57
	p	0.789	0.089	* * *				0.903	0.106	* * *			
DI	a	-1.760	2.751	ns	0.089	1.05	75.62	-3.315	2.953	su	0.103	1.053	75.35
	p	0.941	0.146	* * *				1.003	0.157	* * *			
DR	a	2.694	1.481	ns	0.105	1.05	78.69	0.245	1.543	us	0.114	1.058	81.78
	p	0.675	0.080	* * *				0.774	0.083	* * *			
DIG	a	1.555	3.492	ns	0.082	1.04	69.07	-0.754	3.973	ns	0.106	1.054	68.92
	p	0.762	0.187	* * *				0.864	0.213	*			
DIP	a	-5.834	4.654	ns	0.101	1.05	80.9	-6.160	5.080	ns	0.120	1.062	77.81
	p	1.159	0.246	*				1.157	0.269	*			
DRG	a	2.176	1.679	ns	0.106	1.05	86.73	0.328	1.740	su	0.114	1.059	88.13
	p	0.698	0.090	* * *				0.770	0.094	* * *			
DRP	a	3.635	3.390	ns	0.107	1.05	54.21	-0.042	3.905	su	0.141	1.073	58.84
	p	0.629	0.184	*				0.790	0.212	*			
L	a	1.905	1.611	ns	0.192	1.10	70.53	-2.351	1.490	us	0.164	1.086	82.21
	p	0.743	0.092	* * *				0.951	0.085	* * *			
LI	a	0.445	2.442	su	0.195	1.10	72.2	-0.631	2.100	su	0.145	1.075	79.47
	p	0.827	0.135	* * *				0.865	0.116	* * *			
LR	a	8.264	2.981	* * *	0.145	1.08	21.72	3.205	2.261	ns	0.084	1.043	62.79
	p	0.365	0.175	ns				0.613	0.133	*			
LRG	a	6.014	0.135	*	0.000	1.00	99.94	2.718	0.123	*	0.000	1.000	99.97
	p	0.478	0.008	*				0.637	0.007	*			
LRP	a	11.4	4.223	*	0.131	1.07	-5.07	3.895	3.882	su	0.111	1.057	37.19
	p	0.187	0.248	ns				0.574	0.228	*			
LIG	a	0.240	3.428	su	0.214	1.11	64.45	0.136	3.154	su	0.181	1.095	67.64
	p	0.834	0.191	* *				0.821	0.175	*			
LIP	a	1.563	3.023	su	0.129	1.07	87.5	-1.368	2.682	su	0.102	1.052	92.5
	p	0.780	0.166	* * *				0.909	0.148	*			

at an advantage at planting (possibly due to increased nutrients and mychorrizal association in germination compost compared to bare root stock *Q. robur*), this has masked the increasing *RGR* of *Q. robur* (Figures 6.2 but see E.10 in appendix E).

### **Biomass allocation.**

Allocation to RWFt and SWFt was greater in Q. robur, along with greater SLA (see chapter 5), while LWFt and BWFt were higher in Q. ilex. Despite greater SWFt in the deciduous species this was not enough to increase  $\overline{RGR}$  to rates higher than Q. ilex in this research, thus agreeing with (Struve et al. 2009). However the greater foliage and branch allocation in Q. ilex attributed to the greater overall biomass at final harvest in the evergreen saplings. Higher RWF allocation and lower LWF seem to be related not just to the deciduous leaf habit but also to reproductive strategy; the fire adapted oak, Q. rubra had the lowest LWFt of four species measured and slowest RGR compared with two early and one late successional species, Q. rubra is a gap coloniser and the authors suggest allocation to roots helped its survival and ability to resprout after fire (Walters et al. 1993).

Nutrient availability tends to increase allocation to foliage and *AGB* although *LWFt* also decreases with light (Poorter et al. 2012), this was not relevant to my research as all saplings were planted in open situations. Water availability can also effect allocation patterns with greater *RWF* allocated when water is scarce (Kuster et al. 2012), while stem and branch fractions can increase without water stress (Litton et al. 2007). Hamerlynck and Knapp (1996) found that red maple and Pine consistently gave greater allocation to roots when soil resources were low, whereas red oak increased *RWF* in response to decreased soil *N* but this increase was significant only when accompanied by increased water. Allocation patterns in this research show *Q. robur* did increase allocation to roots at nutrient limited sites but this was also linked to sites that were also water limited (L and B).

### Urban influences.

*Q. robur* had reduced allocation to foliage at urban sites, although the urban site L was water limited, water stress was less obvious at D which may suggest other urban related factors

were reducing *LWFt* at these sites. Other research shows a reduction in *LWF* for *Q. petrea* due to drought stress (Gieger and Thomas 2002) although it my also be a result of *Q. roburs* increased susceptibility to pollutants compared with *Q. ilex* (e.g. Calatayud et al. 2011). Further, Searle et al. (2012) found *Q. rubra* seedlings accumulated biomass at double the rate from rural and remote sites along a rural to urban gradient, however this research shows that despite higher  $CO_2$  concentrations and warmer night time temperatures in urban situations, oak sapling growth is site specific and not always explained by strict 'urban' or 'rural' environments, at least within Plymouth where background  $NO_2$  levels are relatively low compared to other cities (Anon 2008).

### 6.5 Conclusions and limitations.

At the sapling stage, increased allocation to foliage in the evergreen Q. *ilex* enables this species to acquire greater biomass and higher  $\overline{RGR}$  over time than the deciduous Q. *robur* over the 30 month experimental period shown here. However, mean temperatures below 5°C put Q. *ilex* under stress and overall biomass is reduced due to lower allocation to foliage at this site in this species. In contrast, in the urban environment and at the drier rural site, Broadley, Q. *ilex* has greater total biomass than Q. *robur* and this was regardless of soil fertility (apart from at the urban site L where RWFt was greater in pots than field grown saplings in this species). At the most favourable site, rural DT, both species were able to accumulate similar amounts of biomass over the experimental period.

The experimental period was short in the relative lifetime of a tree and growth strategies change with ontogeny. In general, young trees invest more carbon to foliage in an attempt to increase productivity (Poorter et al. 2012). The studies cited above analysed young seedlings from germination to 100 days and therefore are only able to infer *RGR* at seedling stage. My research gives a better idea of relative variability of *RGR* over the lifetime of a young tree after the initial seedling stage. However, relative growth rate was analysed using stem growth therefore further analyses of *RGR* in terms of *SLA* and *ULR* and *LAR* would be useful (see equation 6.7). Further, only 10-30% of *C* fixed during photosynthesis is using for wood production and further, overall biomass is poorly related to carbon assimilation as retention of past assimilation contained in

biomass (Litton et al. 2007). See also (Franco 1985) who suggested a model for tree growth based on living components of biomass.

$$RGR = ULRXSLAXLWF \tag{6.7}$$

# 6.5.1 Allometric Equations.

The search for simplicity in monitoring forest carbon stocks has led to a vast bank of information. From the theory of metabolic scaling (Enquist et al. 1998; Muller-Landau et al. 2006; Enquist, Allen, Brown, Gillooly, Kerkhoff, Niklas, Price and West 2007; Enquist, Tiffney and Niklas 2007) to the generalisation of equations for forest biomass at large (Zianis and Mencuccini 2004). Many authors have looked at the usefulness of additive parameters such as specific density (Muller-Landau 2004) and BEFs (Gracia and Sabate 2002).

Further analyses of the equations added here are necessary to elucidate the usefulness of site specific equations although it is need that population size at some of the more stressed locations will be unlikely to be useful. However, further modeling of the data is required. Specifically, adding density formulae and using the Akaike information criterion (AIC) where:

$$AIC = -2ln(L) + 2p \tag{6.8}$$

and L is the probability of the fitted model, p is the total number of parameters in the model and the best fit model reduces AIC (Chave et al. 2005).

# **Chapter 7**

# Mature Trees.

In order to aid the interpretation of ontogenetic¹ changes in photosynthesis and water use efficiency in Q. ilex and Q. robur a small canopy experiment was carried out and mean assimilation rates, photosynthetic efficiency and stomatal conductance were recorded in three mature trees of each species at one rural, urban and upland location within the Southwest. Results showed that although Q. robur had greater assimilation rates overall, this was only significant at the urban site, where low assimilation rates in Q. ilex coupled with closed stomata led to high water use efficiency but low carbon uptake at the urban site. Both species experienced a reduction in photosynthetic efficiency in sun leaves compared to shade. Performance index (PI ratio) was reduced in Q. ilex in upper canopy leaves due to the high light levels at the top of the canopy  $(7.08\pm0.08$  upper sun leaves compared to 6.67  $\pm$  0.08 lower sun leaves) and this effect was more prominent at the urban site. So, although Q. ilex had higher photosynthetic rates in March when the deciduous species was not in leaf, and recovered better from drought, evidenced by an increase in photosynthetic efficiency in the wetter June 2011 compared to June 2010, the down regulation of photosynthesis and stomatal control during higher temperatures interfered with carbon gain at the urban site for this species. However, milder winters and the ability of Q. ilex to withstand cold periods mean it is likely to become more common throughout the rural Southwest. Comments on ontogenetic differences are presented in the general discussion.

### 7.1 Introduction.

It is recognised that forests and forestry can have a substantial influence on the reduction of atmospheric carbon dioxide  $(CO_2)$  concentrations and are important for climate change mitiga-

¹Relating to growth and developmental stages of an individual

tion (Anon 2003; Hegerl et al. 2007). In the UK, forest area has more than doubled over the past 80 years, and in 2007 was removing about 15  $MtCO_2$  per year (Read et al. 2009). However, Britains existing forests only cover around 12% of available land area, which is among the lowest of any country in Europe (Read et al. 2009). UK Forestry policy involves expansion of forest and short rotation forestry as well as the increased use of timber for construction (West and Gawith 2005). However, species choice is an important consideration and climate change mitigation using (a/re)fforestation has to take into account food production and national biodiversity targets (Morecroft et al. 2009).

Locally changing weather patterns mean that species productivity in relation to local environments also needs to be addressed. The potential of changing climate to interfere with a species range means the incorporation of non-native species for carbon sequestration may become more important in the future and the Forestry Commission conducts trials using non-native species (Willoughby et al. 2007). Range changes also mean that even without planting schemes, species composition in British woodlands is likely to change. Already there is an increasing presence of understory evergreen shrubs such as holly (Peterken 2001; Berger et al. 2007) and risk of invasion from non-native species is more likely. *Rhododendron ponticum* is considered invasive and reduces biodiversity, while Sycamore (*Acer pseudoplatanus*), a non-native gap species, has become more common in disturbed woodland (Epron and Dreyer 1993*a*).

There may be no advantage to either a deciduous or evergreen leaf habit (Royer et al. 2005; Dungan and Whitehead 2006). An evergreen tree may have a longer photosynthetic season, while the deciduous tree may make better use of its available growing season, or there may be trade-offs in relation to their environment: long lived leaves being an advantage in milder climates turn to a disadvantage when temperatures become very cold. To illustrate, *Q. robur* was hardier than *Q. ilex* with mean cold hardiness values in January of -56 and -27 °C for *Q. robur* and *Q. ilex*, respectively (Morin et al. 2007).

Further, the evergreen habit may be an adaptation for areas where soils are poor, an example of this was shown using *Quercus* spp.; the deciduous oak *Q. lobata* had higher photosynthesis and carbon sequestration potential than the evergreen *Q. agrifolia* when resources were plentiful but

### CHAPTER 7. MATURE TREES.

lost this advantage when resources were scarce (Hollinger 1992).

Deciduous oaks are important forestry species throughout Europe. Q. robur is native to the UK and its natural range extends throughout Europe, SW Asia and North Africa (Kleinschmidt 1993). In the UK Q. robur is considered more drought sensitive compared to Q. petrea, and has suffered in areas that have become drier over the last 150 years (Epron and Dreyer 1993a). However, there is evidence that Q. robur is becoming more productive in upland areas of the Southwest UK with a more favourable climate (e.g., Wistmans wood, Dartmoor (Chavana-Bryant 2006)). Indeed, research in America has found differences in drought sensitivity between cooccurring deciduous oaks in the US and that tolerance to high temperatures determines species distribution at range limits (e.g. Hamerlynck and Knapp 1996). Non-native evergreen and deciduous oaks including Q. ilex and Q. alba have also been suggested as potentially useful for UK forestry if large climate changes occur (Read et al. 2009). Q. ilex, a schlerophyllus evergreen, is one of the dominant Mediterranean forest species and in its Mediterranean range tends to be restricted by water availability and winter temperatures (Ogaya and Penuelas 2007a). Its leaves may suffer from thermal stress above 35 °C (Gratani, Pesoli, Crescente, Aichner and Larcher 2000) but Q. ilex still grows on sites where the maximum air temperatures reach 40 - 50 °C (Penuelas and Ogaya 2003). Q.ilex also shows large phenotypic plasticity throughout its range (Gimeno et al. 2009). Q. ilex is naturalised in Southern UK and is frequently found along the coast, in parks, and in urban areas throughout the Southwest. Although Q. petrea is the most widely found oak in the Southwest, Q. robur is still an important species in this area and can cope with water logging better than Q. petrea with increased productivity in this species found in lowland areas (Boisvenue and Running 2006). Given these considerations, and in order to aid evaluation of sapling data, we investigated net assimilation rates, water use efficiency, photosynthetic efficiency and leaf level carbon nitrogen patterns as well as specific leaf area (SLA) in mature trees of these two species at three contrasting sites.

- 1. General working hypotheses for mature trees:
  - 1.1. Are *Q. ilex* carbon assimilation rates and photosynthetic efficiency higher or lower than *Q. robur* for mature trees?

- 1.2. Are *Q. ilex* carbon assimilation rates and photosynthetic efficiency higher or lower than *Q. robur* in relation to urban, rural and upland environments?
- 1.3. At leaf level, do leaf state (sun and shade) and position in the canopy profile (lower middle and upper); affect assimilation rates and photosynthetic efficiency in either species?
- 1.4. Are leaf nitrogen  $(N_{leaf})$  and leaf carbon  $(C_{leaf})$  and SLA affected by any of the factors discussed and how does this relate to overall productivity for each species.

### 7.2 Methods.

### 7.2.1 Study sites.

Three locations, representative of upland, rural and urban areas, were chosen. The upland site was chosen as the highest site in SW England containing mature evergreen *Q. ilex*. This upland site is at Castle Drogo, [SX723901, 260 metres above sea level (m.a.s.l)], soils are thin, freely draining, acid and loamy soils over granite rock, typical upland soils of heath and moor (National Soil Resources Institute 2009). Both *Q. ilex* and *Q. robur* at this site are approximately 100 years old. The rural site is at Mount Edgcumbe, SE Cornwall. Here, *Q.ilex* is found at the 'top park', (SX457519, 60 m.a.s.l) and *Q. robur* is found at the Barrow car park area near Mount Edgcumbe House (SX451526). Soils at Mt Edgcumbe are freely draining, acid loams and can shallow near the top of the slope (National Soil Resources Institute 2009). Tree age at the rural site is between 100-150 years old, although these trees have possibly been pollarded in the past (Park Ranger pers. Comm.). Finally, the urban sites are Devonport Park (SX457550), Plymouth, for *Q. ilex* and Central Park (SX469560), Plymouth, for *Q. robur*. At 35 m.a.s.l. both sites are situated next to busy roads. Using historical evidence (e.g. Digimap 2009), tree age at both urban sites was estimated to be 120 years, again freely draining acid loams predominate but developed by the urban environment.

# 7.2.2 Climate Information.

In general climate in the Southwest UK is Oceanic with South-westerly dominating winds. Therefore this part of the UK experiences mild winters and warm wet summers with annual

*Table 7.1:* Climate statistics for the study sites (maximum, minimum, mean average temperate (MAT) and total rainfall). The first column provides both the distance between study site and the nearest meteorological station (m) from which the climatic information was obtained and their difference in height above sea-level.

m/m			$^{\circ}C$		mm
Site-station	Period	Max Temp	Min Temp	MAT	Total rainfall
Upland	Jun-10	19.7	9.8	14.6	39.1
3000/0	Mar-11	10.4	3	6.2	23.4
	Jun-11	17.1	9.1	12.7	74.9
	2010	11.8	5.2	8.4	946.2
	2004-2010	12.7	6.4	9.5	1101.3
Rural	Jun-10	22.4	8	15.6	26.2
3000/0	Mar-11	17.4	-1.7	8.5	36.4
	Jun-11	26.9	5.9	14.6	82.8
	2010	14.7	7.7	11.2	714.8
	2004-2010	14.3	8.2	11.3	962.6
Urban	Jun-10	23.5	9.6	15.9	37.5
1000/0	Mar-11	17.6	-0.1	8.8	29.4
	Jun-11	26.6	7.5	14.5	68.8
	2010	*	5.9	10.1	884.7
	2004-2010	*	8.2	11.1	989.4

Climatic data station information:

Rural, UK met office historical data for Mountbatten, Plymouth (50.35N, 04.12W) Upland, Bridford, East Dartmoor Independent Weather station, (50.42N, 03.42W) Urban, University of Plymouth Metnet archive, (50.22N, 04.08W). temperatures averaging around 11  $^{\circ}$  C, although the upland environment such as found on Dartmoor can experience average annual temperatures up to 2 $^{\circ}$  lower (Jebson 2007). Annual rainfall averages for the South West region are around 900-1000mm in the lowland coastal areas with Princetown, 400masl having around twice as much (Jebson 2007). Climate records were obtained from the UK Met Office Southwest, and from Independent Weather stations at the University of Plymouth (situated at Drakes Circus, Plymouth) and Bridford (situated in East Dartmoor) (Table 7.1. See general methods section for complete climatic summary (chapter 2).

### Nitrogen dioxide concentrations within Plymouth.

At the time of data collection the 2010 annual mean  $NO_2$  concentrations for Tavistock road (situated next to Central park and *Q. robur* trees) were between 40 and 43  $\mu mol m^3$ , close to and slightly above the target annual mean set out in the Environment Act 1995 of 40  $\mu mol m^3$ (Environmental Protection and Monitoring Unit 2009). However more recent data for the automatic monitoring site at Tavistock Road put annual means at 37.7 and 30.9  $\mu mol m^3$  in 2010 and 2011 respectively (Environmental Protection and Monitoring Unit 2012). The closest monitoring site in Alma road recorded annual mean of 36.1 and 33.1  $\mu mol m^3$  in 2010 and 2011 respectively (Environmental Protection and Monitoring Unit 2012). Background levels were 16.9  $\mu mol m^3$  in 2008, 36.5  $\mu mol m^3$  in 2010 and 29.6  $\mu mol m^3$  in 2011 (Environmental Protection and Monitoring Unit 2012). Therefore it can be assumed that both species are receiving some *N* deposition as both species are situated by the roadside at the urban sites.

### 7.2.3 Sampling Protocol.

During the initial data collection in June 2010, only two trees per species at each site were measured due to the time constraints imposed by using newly qualified canopy climbers and the need to complete measurement before 12pm when assimilation rates tend to drop. The following year it was possible to measure three *Q. ilex* per site during March 2011, before the deciduous oak came into leaf and the same three sampled again with three *Q. robur* in June 2011 at each site. The different levels of the canopy were accessed via ropes: Specifically using the double rope, 'Basic Canopy Access Procedure' (BCAP) to gain access to the canopy, followed by single rope technique to enable movement around the canopy.

Sampling was undertaken in the tree by trained volunteers from the Universal Canopy Exploration (UCE) group (see acknowledgments) and where access was difficult, branches cut and sent down to be measured at the base of the tree. All cut branches were placed in water to avoid water loss during sampling time and a small experiment showed no significant differences between cut and uncut branches for any of the parameters used (p>0.05). Parameters were measured on ten sun leaves and ten shade leaves from each canopy position (upper, middle and lower canopy). Lower canopy was accessed from the ground using a crook to bring branches closer if necessary, middle canopy was taken to be the area of the crown at its widest point ( $\pm$  2.5m) and the upper canopy the area as close to the top as possible but within 3 m of the top for difficult trees. For more open canopies where there was less distinction between sun and shade leaves, leaves were measured according to lightest (outer canopy) and shadiest (inner canopy) parts of the canopy.

All leaf level measurements were taken on 10 leaves per canopy level (LOWER/MIDDLE/UPPER) and leaf state (SUN/SHADE). Parameters measured were; Leaf level net photosynthetic rate (An  $\mu mol \ m^2 \ s^{-1}$ ) and stomatal conductance (*Gs mol \ m^2 \ s^{-1}*) was measured using the LCi portable IRGA (ADC Bioscientific 2007). Intrinsic Water Use Efficiency (iWUE) measured as net photosynthetic rate (*A_n*) and stomatal conductance of water (*Gs*) ratio *A_n/Gs* ( $\mu mol \ CO_2 \ / \ mol \ H_2 O^{-1}$ ). The IRGA was used with a ADC light unit and irradiance set to 550-600 P.A.R. to enable comparison between sites and species at a consistent light level. Therefore any differences between sun and shade leaves are likely to be intrinsic due to the homogeneous light environment used.

Photosynthetic efficiency was measured using the plant efficiency analyser "Pocket PEA" (Hansatech Instruments Ltd 2010). The pocket PEA is a continuous excitation fluorometer that takes advantage of the red/far-red light energy re-emitted during the process of photosynthesis, and gives an indication of photosynthetic performance and plant health. Parameters used for analysis were; variable fluorescence over maximum fluorescence ( $F_v/F_m$ ), a measure of maximum quantum efficiency of Photosystem II.  $F_v/F_m$  is a sensitive indication of plant photosynthetic performance, with 0.85 being the maximum obtained in a healthy plant, lower values are an indication of stress (Hansatech Instruments Ltd 2007). Performance index (*PI*) developed by Strasser et al. (2000). *PI* incorporates three different stages of the fluorescence process to give an indication of plant vitality (Hermans et al. 2003). *PI* is more sensitive to stress than  $F_v/F_m$ and can exhibit a loss in performance during the energy transduction process in PSII (Clark et al. 2001). Both parameters are a measure of photosynthetic efficiency. Photosynthetic efficiency can be an indication of a plants reaction to drought and/or light stress.  $F_v/F_m$  itself is related to the potential for photochemical quenching or quantum yield of PSII (Strasser et al. 2000). Values less than 0.8 may indicate stress responses from the sample, and such reductions have been correlated with high light levels during the day (Cavender-Bares and Bazzaz 2004) and is often associated with photo-inhibition where leaves reduce their capacity for photochemical quenching as a protection from high light (Clark et al. 2000).

During June 2011 data collection, twenty leaves were taken from each factor level from each tree and taken to the lab in a cold storage box. Ten leaves from each factor level and tree were used to estimate specific leaf area *SLA*. this was done by first measuring fresh weight, then taking the leaf area using the LiCor Leaf Area Instrument. The leaves were then dried at 70° C for one week, or until two weights were the same. Dry weight was then taken and *SLA* and its inverse leaf mass per area *LMA* derived (*LMA* data not shown). The other half of the leaf samples were dried and ground to a fine powder using a burr grinder. The ground leaves were then weighed to 2 *mg* into small tin cups and then analysed in the Autoanalyser (see chapter 2 for more details on this procedure) for *N* and *C* concentrations (Leaf CN was measured on two separate occasions for seedlings and mature trees. The mature tree samples were kindly prepared and measured by Sarah Peake, Environmental Science graduate and volunteer).

Height and DBH were also taken of each tree sampled using a clinometer and DBH tape. For Q. *ilex* that had multiple stems DAG was taken. An estimate of biomass was taken using the general volume equation  $(HD^2)$  in  $cm^3$  and this was used in a full factorial GLM using site and species as fixed factors. Mature tree biomass estimates are treated with caution due to differences in tree age between sites.

### 7.2.4 Analysis of data.

Statistical tests were performed on pooled leaf data (i.e. ten leaves for each leaf state and canopy position averaged) on three trees per species; this was to avoid pseudo-replication within the canopy layers. Data was analysed with general mixed linear models (GMLM) using R (R 2009). Model one looked for differences between sites and species for each parameter, models two and three analysed differences between dates for each species separately and model four analysed species (SP), site (S: Urban, Rural, Upland) leaf state (STATE: shade, Sun) and leaf position (LP: Lower, Middle, Upper) in a full factoral design just using the June 2011 data to simplify interpretation of the results.

All model parameters met conditions for normality (equal variances using Levenes test, p>0.05). Values are presented with means and std deviations when shown as descriptive statistics and means  $\pm$  95% confidence intervals when used in graphs to aid interpretation of significant differences.

### 7.3 Results.

### 7.3.1 Differences between species and sites.

### Water use efficiency and photosynthetic parameters

Mean  $A_n$  was higher in Q. robur than Q. ilex at all sites, mean  $A_n$  was significantly lower at the urban site in Q. ilex than at the other sites for this species while also significantly lower than Q. robur at this site (Tables 7.2 and 7.5). Q. ilex Gs was more variable than in Q. robur and was significantly lower at the urban site than all other factor levels (Table 7.2). iWUE was lower in Q. ilex than Q. robur and the combination of increased  $A_n$  and moderate Gs in Q. robur has lowered its iWUE at the urban site where in fact both species have lower iWUE (Tables 7.2 and 7.5).  $F_v/F_m$  and PI were lower in Q. robur than Q. ilex at the upland site, where mean PI was nearly twice as high in Q. ilex (Tables 7.2 and 7.5).
#### leaf traits.

*SLA* was significantly higher in *Q. robur* than *Q. ilex*, the same was true for  $N_{leaf}$ , however, for  $C_{leaf}$  the opposite was true.  $N_{leaf}$  was higher at the urban site than at the other sites in *Q. robur* and  $C_{leaf}$  was significantly lower at the urban site in both species (Table 7.3).

#### 7.3.2 Differences between dates.

#### Water use efficiency and photosynthetic parameters.

There were significant differences between dates for both species in *iWUE*, however although both species also showed an interaction between date and site, there was no significant effect of site on *Q. robur iWUE* in mature trees (Table 7.4 for ANOVA results and appendix H for graphs). *iWUE* was significantly higher in June 2011 than June 2010 for both upland and rural sites but not the urban site in *Q. robur*, whereas in *Q. ilex*, *iWUE* was significantly higher in March 2011 and June 2011 than the previous June except at the urban site where June 2011 is lower than the previous March (Table 7.4 for ANOVA results and appendix H for graphs). This corresponds with lower monthly rainfall in June 2010 than in June 2011.

There was an effect of all factors and interaction in both species for mean  $A_n$  (Table 7.4). Closer inspection of the data showed that in *Q. ilex*  $A_n$  was higher in March (when *Q. robur* was not in leaf) than June (both dates). For *Q. robur*, there was a slight increase in mean  $A_n$  rates in June 2011 at the uplands site although the opposite was true at the urban site (appendix H Figure H.3). However, in *Q. ilex* photosynthetic efficiency was low in March compared to June dates  $(F_v/F_m$  mean values  $0.77 \pm 0.04$  in March compared to  $0.82 \pm 0.03$  in June and PI mean values of  $6.1 \pm 1.88$  and  $9.2 \pm 2.68$  respectively). There was also an increase in *PI* values between June 2010 and 2011 at the urban and uplands sites for *Q. robur* and for all sites in *Q. ilex* (See appendix H for figures).

.(e			Aac	Aab	Bb		Aac	Aac	Ac
appulse	St		1.86	2.39	1.13		2.44	3.48	2.45
	$A_n$		5.91	4.61	3.12		6.17	5.55	5.11
	D		Aa	$\mathbf{A}\mathbf{b}$	$\mathbf{A}\mathbf{b}$		Aab	Aac	$\operatorname{Bab}$
	St		30.6	36.8	14.0		24.2	30.9	16.5
/ argumea	iWUE		55.0	49.9	44.0		64.9	75.2	42.6
			Aa	Ac	$\mathbf{Bc}$		Aac	Abc	Aac
משר ורוור	StI		0.146	0.560	0.030		0.069	0.102	0.053
	$G_S$	. ilex	0.18	0.32	0.08	robur	0.12	0.10	0.12
ics allu		б	Аа	Ac	Aa	Ó	Bb	Abc	Aac
	St		2.26	1.94	3.59		1.39	2.34	1.99
	Ιd		9.44	8.57	9.56		4.21	7.10	8.07
			Aa	Аа	Aa		Ba	Аа	Aa
n enroade	StI		0.016	0.015	0.027		0.043	0.035	0.012
MOITE ST	$F_v/F_m$		0.82	0.82	0.82		0.79	0.81	0.83
זרוור	No.		18	18	18		18	18	18
			UPLAND	RURAL	URBAN		UPLAND	RURAL	URBAN

*Table 7.3:* Mature tree leaf traits: specific leaf area (SLA)  $cm^2 g^{-1}$ , leaf nitrogen  $N_{leaf}$  %, and leaf carbon  $C_{leaf}$  %. Sample size (N), means, standard deviations (StD) and significant differences between sites (group) for each species (capital letters) and between sites (lower case letters)

Descriptive statistics for canopy leaf traits

Leaf trait	No.	SLA	StD	group	N _{leaf}	StD	group	$C_{leaf}$	StD	group
					Q. ilex					
	10	6.92	1 (05	Α.	1 070	0.107	<b>A</b> .	47.000	0.069	
UPLAND	18	6.82	1.605	Aa	1.272	0.197	Aa	47.999	0.968	ABD
RURAL	18	8.38	1.553	Ab	1.363	0.278	Ac	48.133	0.982	ABb
URBAN	18	7.53	1.855	Ac	1.361	0.279	Ae	47.220	0.803	Ba
					Q. robur					
UPLAND	18	16.84	6.681	Ad	2.128	0.406	Ab	47.336	0.930	Aa
RURAL	18	18.34	5.473	Ad	2.221	0.389	Ad	47.245	0.777	Aa
URBAN	18	16.23	4.467	Ad	2.468	0.345	Af	46.565	0.779	Aa

*Table 7.4:* Differences in physiological parameters between dates (June 2010, March 2011 (Q. *ilex* only) and Jun 2011). Differences in intrinsic water use efficiency (*iWUE*) A/Gs  $\mu mol \ s^{-1} \ m^{-2} \ / \ mol \ H_2O^{-1}$ , net assimilation rate ( $A_n$ )  $\mu mol \ s^{-1} \ m^{-2}$ , Variable fluorescence over maximum fluorescence ratio  $F_{\nu}/F_m$  and performance index *PI* parameters shown for sites and dates for **A** *Q*. *ilex* and **B** *Q*. *robur*.

#### A) Q. ilex

ANOVA table: Canopy physiological variables

		iWUE		GS		An		$F_{v}F_{m}$		PI	
	Df	F	sig	F	sig	F	sig	F	sig	F	sig
DATE	2	6.54	**	0.77	ns	27.62	***	35.70	***	38.01	***
SITE	2	2.12	ns	5.96	**	7.39	***	1.57	ns	1.63	ns
DATE:SITE	4	2.77	*	1.88	ns	2.50	*	1.28	ns	1.10	ns
Residuals	135										

#### B) Q. robur

		iWU	UE	$F_v/I$	$F_m$	Pl	!
	Df	F	sig	F	sig	F	sig
DATE	1	8.31	**	1.23	ns	10.87	**
SITE	2	4.62	*	5.53	**	28.95	***
DATE:SITE	2	8.37	***	6.69	**	8.89	***
Residuals	84						

#### 7.3.3 Differences in sun and shade leaves and position in the canopy.

#### Water-use efficiency and photosynthetic parameters.

Mean  $A_n$  was significantly higher in sun leaves than shade leave for both species (Figure 7.1) and although there was no effect of LP on this parameter, a complex interaction existed where Q. *robur* mean  $A_n$  was significantly lower in the upper canopy sun leaves at the rural site (2.9  $\pm$  0.12 SUN;  $3.7 \pm 0.49$  SHADE) whereas the opposite is true for Q. *ilex* (Table 7.5, Figure 7.1). Mean Gs and *iWUE* were not affected by LP or STATE but *iWUE* did share an interaction between Species, LP and site where Q. *robur* had significantly greater *iWUE* in the lower canopy at the upland site compared with Q. *ilex* (22.6  $\pm$  12.88 I; 78.7  $\pm$  6.12 R) (Figure 7.3).  $F_v/F_m$  and *PI* were significantly lower in sun than shade leaves for both species (SUN 0.81  $\pm$  0.02; SHADE 0.83  $\pm$  0.01 for Q. *ilex* and SUN 0.79  $\pm$  0.04; SHADE 0.83  $\pm$  0.01 for Q. *robur*) (Table 7.5 and Figure 7.4).  $F_v/F_m$  was reduced in upper canopy in Q. *ilex* leaves although this effect was not significant 7.4.

#### leaf traits.

SLA was only significantly affected by species and leaf state and there were significant differences at all levels of the interaction between species and state. Sun leaves are significantly lower *SLA* than shade in both species (SUN: 67.69  $\pm$  19.76; SHADE: 83.77  $\pm$  10.50 in *Q. ilex* compared to SUN 142.69  $\pm$  50.30; SHADE: 200  $\pm$  45.87 in *Q. robur*) (see Table 7.6 and Figure 7.6 different letters show significant differences using Tukeys HSD). There was an effect of leaf state (STATE) on *C*_{*leaf*}, although on average SUN leaves had 0.4 % higher *C*_{*leaf*}, however in *Q. robur* mean difference was 0.6 % higher in SUN leaves (Figure 7.8). In contrast *N*_{*leaf*} had an effect of leaf state, while leaf position interacted with species and site (Table 7.5, Figure 7.8). *N*_{*leaf*} is significantly lower in sun than shade leaves (SUN 1.4  $\pm$  0.99; SHADE 1.2  $\pm$  0.96 *Q. ilex* and SUN 2.4  $\pm$  0.69; SHADE 2.18  $\pm$  0.93 *Q. robur*) and this distinction is more prominent in the lower canopy in *Q. ilex. Q. robur* upper canopy leaves had higher *N*_{*leaf*} at the urban and rural sites (Figure 7.8).

1 7 87											
Response:	Gs		iWUE		$A_n$		$F_v/F_m$		PI		
	Df	F	sig	F	sig	F	sig	F	sig	F	sig
SPECIES	1	2.61	ns	5.63	*	6.07	*	7.24	**	41.4	***
SITE	2	1.91	ns	6.49	**	6.61	**	7.48	***	7.30	**
STATE	1	4.53	*	0.15	ns	15.8	***	54.3	***	0.08	ns
LP	2	0.03	ns	0.16	ns	1.34	ns			2.05	ns
SPECIES:SITE	2			2.67	ns			9.12	***	8.65	***
SPECIES:LP	2			1.94	ns			9.82	**		
SITE:LP	4			2.54	*			2.91	ns		
SPECIES:SITE:LP	4			4.25	**			2.90	ns		
SPECIES:SITE:STATE:LP	4				ns				ns	2.88	*
Residuals	101			89.	00	10	)1	9	6	7.	2

*Table 7.5:* Leaf physiological parameters, differences between sun and shade (STATE), position in the canopy (LP) and species.

ANOVA: Leaf physiology variables

*Table 7.6:* Effects of Leaf state (SUN SHADE, species and interactions. No effect of canopy position therefore data not shown (see glossary for parameter units).

ANOVA:Morphological leaf trait SLA									
Response:	SLA								
	Df	F	р	sig					
SPECIES	1	192	0.00	***					
STATE	1	28.3	0.00	***					
SPECIES:STATE	1	8.94	0.003	**					
Residuals	104								

Table 7.7: Canopy leaf N and C. Leaf state, canopy position and site effects for each species.

ANOVA table						
Leaf trait		Ν	leaf	$C_{leaf}$		
	Df	F	Р	F	Р	
SPECIES	1	883.85	0.00***	18.23	0.00***	
SITE	4	20.18	0.00***	10.79	0.00***	
STATE	1	10.11	0.00***	58.18	0.00***	
SPECIES:LP	1	4.31	0.04*			
SPECIES:SITE:LP	4	3.19	0.02*			
Residuals	89					

## Comparison of estimated biomass in Q. robur and Q. ilex.

Biomass was significantly higher in *Q. robur* than *Q. ilex* ( $F_{(1,13)} = 7.67$ , p<0.05) and an interaction between species and site ( $F_{(2,13)} = 6.18$ , p<0.05) showed that *Q. ilex* biomass was higher at the upland site than *Q. robur* (829.3 ± 476.75  $m^3$  compared to 567.17 ± 328.45  $m^3$ ).

Descriptive Statistics for Canopy Biomass									
SPECIES	SITE	No.	mean HD ²	StDev					
Q. ilex	RURAL	3	548	556					
	UPLAND	3	829	477					
	URBAN	3	503	50.1					
Q. robur	RURAL	3	1790	328					
	UPLAND	3	567	329					
	URBAN	3	991	192					

*Table 7.8:* Estimated biomass shown as volume estimate in  $m^3$  (means and standard deviations shown).

#### 7.4 Discussion.

#### 7.4.1 Differences between species and environment.

Deciduous oaks have higher maximum assimilation rates than their evergreen counterparts. However, the evergreen Q. *ilex* achieves maximum photosynthetic rates during spring and autumn when air temperatures are mild and water not limiting (Baldocchi et al. 2010; Vaz et al. 2010). This is also true during the winter months when deciduous oaks are not in leaf. In fact, my research shows that Q. *ilex* winter assimilation rates can increase to rates comparable with Q. *robur* summer rates. In contrast, Q. *robur* has been shown to increase assimilation rates quickly with increasing light (Baldocchi et al. 2010), however, increasing temperatures also allow Q. *robur* to increase assimilation rates irrespective of increasing light (as evidenced here by using the light unit). Q. *robur* has to make efficient use of the photosynthetic season and will increase its assimilation rates during higher temperatures even when short term drought occurs (Drobyshev et al. 2008). This anisohydric' can interfere with its water saving strategy of stomatal control and can leave it vulnerable to further drought episodes (Cater and Batic 2006; Sade et al. 2012).

So, evergreen broadleaved trees show a restrained response to temperature increases compared with deciduous trees (Way and Oren 2010), which for Q. *ilex* may be intrinsic adaption to drought stress in the Mediterranean environment. My research showed Q. *robur* had greater *iWUE* than Q. *ilex* and in general it maintained higher photosynthetic rates despite lowered Gs, this has been discussed by previous authors for Q. *robur* (e.g., Epron and Dreyer 1993*a*;

Kazda et al. 2000). However, despite increasing assimilation rates giving the deciduous oak an advantage over the summer months, the evergreen oaks tighter stomatal control, means it can recover from drought stress more readily (Chaves et al. 2002).

Reduced *iWUE* at the urban site may be a combination of restricted rooting depth and water availability, a direct result of the urban environment, controlling stomatal conductance and interacting with the effects of particulate pollution at that site. Research suggests benefits for planting evergreen oaks in the urban environment; in general, broadleaved species with rough surfaces are better at collecting particles, and indeed *Q. robur* was able to absorb significant amounts of pollutant before showing sign of stress (Beckett et al. 2000). However, in another study, the evergreen *Q. ilex* was better at reducing pollution and air temperatures in Rome than the deciduous *Q. pubescens*, the leaves not only absorbing pollutants but also reducing air temperatures and humidity in the City (Gratani, Crescente and Petruzzi 2000). There were also benefits for being photosynthetically active and maintaining a canopy all year round (Gratani and Varone 2006) although of the mature oaks in my research, *Q. ilex* trees were showing signs of stress via stomatal control (but see also Ugolini et al. 2012). Despite this the urban environment does not seem to have affected photosynthetic efficiency in either species despite the stomatal control and lowered water use efficiency of both species growing in Plymouth City Centre.

In terms of photosynthetic efficiency, Q. *ilex* also seems better able to recover from stress than Q. *robur* and PI remains lower in Q. *robur* at all sites while  $F_v/F_m$ , is lower in Q. *robur* than Q. *ilex* at the upland site (Castle Drogo, East Dartmoor). This suggests that during the summer months when trade-offs between stress responses and assimilation rates occur in Q. *ilex* electron transport and photochemical quenching remain efficient. However, winter photosynthetic efficiency in evergreen broadleaves show signs of photo-inhibition in the Mediterranean environment (Martinez-Ferri et al. 2004) as here evidenced by the reduced  $F_v/F_m$  and PI values during March for Q. *ilex*. In contrast Q. *robur* showed signs of stress in the upland environment. Indeed this oak is showing growth decline in many parts of its European range and this decline has been directly associated with regional climate change (e.g. Drobyshev et al. 2008).

#### 7.4.2 Differences between sun and shade leaves and their position in the canopy.

#### Water use efficiency and photosynthetic parameters

Assimilation rates of sun leaves tend to be higher than shade leaves in deciduous tree species, as was found in *Q. crispula*, a deciduous oak found in the Far East, which had assimilation rates of 13.4 and 6.7  $\mu mol CO_2 m^{-2} s^{-1}$  in sun and shade leaves respectively comparable to those found here (Koike et al. 2004). However, the same study found no difference between sun and shade in the upper canopy (Koike et al. 2004). In my study, *Q. robur* experienced reduced photosynthesis in the upper canopy despite no reduction in photosynthetic efficiency at this height (e.g., by limiting light levels by using a light unit), an indication that although there may be an intrinsic response to high light and temperatures in assimilation rate at the upper canopy any reduction in photochemical quenching at this height is likely to be reversible for this species. Other research has shown that upper canopy leaves from a deciduous red oak (*Q. rubra*) and *Q. ilex* experienced reduced  $F_v/F_m$  in upper canopy sun leaves compared to the lower canopy (Manes et al. 1997; Cavender-Bares and Bazzaz 2004).

In contrast, in Q. *ilex*, there is an indication of reduced photosynthetic efficiency, implying photo-inhibition in the upper canopy, although the signature is small which again may be due to the use of the light unit where light levels remained at saturation levels but not higher, thus not inducing photo-protective down regulation of the photosynthetic apparatus and indication that reversible photoinhibition is more evident at rural and upland sites but that more permanent damage has occurred at the urban site. Molchanov (2005) found that for shade and sun leaves in the upper canopy, photosynthetic efficiency depended on solar radiation, water availability and air temperatures. The reduced performance and vitality shown by a reduction in these parameters during March for Q. *ilex* indicate that this species is less tolerant of high light during lower temperatures and indeed it is has been suggested that Q. *ilex* experiences photo-inhibition during periods of cold stress (Ogaya and Penuelas 2003; Valladares et al. 2005) This has also been shown in other evergreen broadleaved trees such as *Ilex aquifolium* growing in a Mediterranean environment (Valladares et al. 2005) (see appendix H for data showing differences between dates).

#### Leaf level C and N and SLA.

 $N_{leaf}$ , measured on a mass basis, was lower in sun leaves than shade in *Q. ilex* and this trend was more obvious in the upper canopy and in the urban environment. In *Q. robur* only the upland site had lower  $N_{leaf}$  in sun versus shade leaves. In fact, for *Q. robur*, without taking into account leaf state and position in the canopy, the urban environment had higher  $N_{leaf}$  than the other two sites. Although  $N_{leaf}$  measured on a mass basis seems to have less predictive power for assimilation rates than area based measurements, the inverse of *SLA*, *LMA* is a good predictor of net assimilation. Therefore *SLA* decreases with increasing irradiance. Upper canopy leaves had greatest  $N_{area}$  because of their high *LMA* in *Q. robur* although  $N_{mass}$  was lower, the high  $N_{area}$  reflected high assimilation rates (10.7+  $\mu mol CO_2 m^{-2} s^{-1}$ ) as  $N_{leaf}$  is distributed through the canopy in order to maximise photosynthesis (Kazda et al. 2000).

#### 7.4.3 Overall biomass estimates and conclusions.

Although the biomass estimates for mature trees in this study are based on volume estimates and are only comparable between species at each site due to age related differences, the results indicate that Q. robur is better at accumulating carbon over time even in the urban environment. This suggests that perhaps increased N deposition to the leaf enables faster utilisation and consequently greater productivity for the deciduous oaks. The evergreen may not be able to utilise the N deposited on the waxy cuticle as efficiently, however Q. ilex leaves have been shown to be efficient bioaccumulators of N in urban areas of Italy (Alfani et al. 2000). However the Q. robur trees at Castle Drogo are under stress which is indicated by reduced photosynthetic efficiency and relative assimilation rates in this species at this site. Mature oaks growing in the oceanic environment within the Southwest, are subjected to increases in pathogen attacks such as the Honey fungus and various Phytophorum species, as well as increased defoliation from insects and caterpillars. Therefore, summer defoliation, combined with poor soils and soil water availability is perhaps the reason for reduced growth in this species at the upland site. Indeed, at 280m above sea level, the Castle Drogo trees experience extended frosts, which have caused cavitation and loss of limbs from many of the mature Q. ilex there (pers. obs). Efficiency is lower during the winter months despite milder temperatures increasing assimilation rates.

However, the competitive ability of *Q. ilex* to persist on such soils in its Mediterranean range along with its resilience to drought and temperature stress may give it an ecological advantage in future climate scenarios in the UK. Indeed, *Q. ilex* has already become a "pest" species for wildlife managers along the south coast of Britain but may become more prominent within the next 100 years.



*Figure 7.1:* Differences in carbon assimilation rate  $(A_n)$  ( $\mu$  mols  $s^{-1}$   $m^{-2}$ ) for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means  $\pm$  95 % CI).



*Figure 7.2:* Differences in canopy stomatal conductance (Gs)  $(m \ mols \ s^{-1}m^{-2})$  for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means  $\pm 95 \%$  CI).



Figure 7.3: Differences in canopy intrinsic water use efficiency (iWUE) (measured as Mean assimilation rate and stomatal conductance ratio) for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means ± 95 % CI).



*Figure 7.4:* Differences in canopy  $F_v/F_m$  as a measure of maximum photochemical quenching of electrons in photosystem II, for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means  $\pm$  95 % CI).



*Figure 7.5:* Differences in canopy performance index PI (a measure of sapling vitality and consisting of three different measurements of chlorophyll fluorescence emissions taken over the fluorescence rise see Strasser et al. (2000) for full explanation of PI) for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means  $\pm$  95 % CI).



*Figure 7.6:* Differences in canopy specific leaf area (SLA), effect of Species and leaf state for each site; A upland, B rural and C urban (means ± 95 % CI). (means ± 95 % CI).



*Figure 7.7:* Differences in canopy  $C_{leaf}$  for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means  $\pm$  95 % CI).



*Figure 7.8:* Differences in canopy  $N_{leaf}$  for sun and shade leaves in lower, middle and upper canopy for each site; A upland, B rural and C urban (means  $\pm$  95 % CI).

# Chapter 8

# **General Discussion.**

This research set out to investigate photosynthetic efficiency and carbon sequestration potential of the deciduous Pedunculate oak (Q. robur L.) and the evergreen Holm oak (Q. ilex subsp. ilex L.) in relation to their environment within the temperate, 'oceanic' climate of the Southwest. The former species, a UK native and common in lowland Southwest England, is drought sensitive (Epron and Dreyer 1993a) but copes well with waterlogged soils (Parelle et al. 2007). In comparison the evergreen has a natural distribution within the sub humid Mediterranean regions, western and central Mediterranean and North Africa, however it can also be found in temperate humid climates but only colonizes more arid areas where temperatures are much colder (Barbero et al. 1992). This suggests that the oceanic, temperate climate of the southwest could become more favourable towards *O.ilex* with future climate scenarios; however, in this type of environment competition from deciduous oaks is also more likely (Barbero et al. 1992). Further, Q. ilex has become more common in South and Southeast UK, and is known to regenerate freely in some areas along the South coast (Plantlife 2010). Q. ilex has also been widely planted (historically) within the Southwest and has become a dominant species in some woodlands such as Mt Edgcumbe in SE Cornwall, while also found as a planted tree throughout Plymouth city centre (pers. obs.).

Alternatively, increased productivity in *Q. robur*, in lowland Southwest, may mean that this species becomes the dominant planted species for forestry purposes (Broadmeadow 2004). However, this too may be questionable, as increased attack from pathogens and consequent 'oak decline' in this species (for more information on oak decline see Brasier 1999) may give a non native species such as *Q. ilex* an advantage, both ecologically and as a forestry species in the future. Although it should be noted that both within its Mediterranean range and in coastal

areas of the UK, *Q. ilex* has itself suffered from attack of oak pathogens such as *Phytophthora* spp.¹ Indeed *Q. ilex* shows morphological plasticity within a wide thermic range (Barbero et al. 1992; Gimeno et al. 2009) and is not only able to decrease *SLA* in response to stress but also exhibits a quick stomatal response to water stress and high temperatures whereby it down regulates photosynthesis until temperatures become more favourable (Gratani, Pesoli, Crescente, Aichner and Larcher 2000).

More specifically the results of the experiments (and observations) detailed here, were aimed at evaluating whether *Q*. *ilex* had greater carbon sequestration potential and photosynthetic efficiency than *Q*. *robur* and whether this was also true in each rural, urban and upland environment.

Through the course of the experiments it was found that the environmental effects of the sapling sites was very much site specific and although no direct correlations between climate and edaphic variables were made, those sites with greater macro nutrients and greater soil moisture values had the greatest photosynthetic efficiency and growth in both species although soil fertility was more important for *Q. robur* than *Q. ilex* as suggested by the literature (e.g. Litton et al. 2007).

#### 8.0.4 Summary of main results: saplings and mature trees.

#### Photosynthetic parameters and leaf traits.

In saplings, *Q. robur* had significantly higher  $A_n$  than *Q. ilex* during the summer, and again when all months were taken into account ( $F_{(1,495)} = 25.4$ , p<0.05)(7.9  $\pm$  3.5 *Q.robur*, N=247;  $6.6 \pm 2.1$ , *Q. ilex*, N=250 ( $\mu$ mol  $m^{-2} s^{-1}$ )). However, *Q. ilex*  $A_n$  was higher at urban *L* in field grown saplings where soils were drier, had the lowest nutrient status (e.g. soil *N* was 1.93  $kg^{-1}$ compared to the next lowest value of 2.79 at DPF) and saplings were more exposed. This was despite photochemical quenching being lower in *Q. ilex* at this site. *PI* was significantly lower in *Q. robur* than *Q. ilex* at rural site B, situated on Dartmoor fringe, having the second lowest air temperatures after the upland site and second driest site after L.

 $N_{leaf}$  and SLA were generally higher in Q. robur than Q. ilex in saplings. Highest SLA and

¹In particular, *Phytophtora cinnamomi* (Brasier 1999).

lowest  $N_{leaf}$  values were at the two coldest sites (DPF and B). In contrast the urban site L had the lowest *SLA* in both species.

In mature trees,  $A_n$ , *SLA* and  $N_{leaf}$  were all higher in *Q. robur* although  $A_n$  was only significantly higher in *Q. robur* at the urban site in contrast with sapling data. However, when March data was taken into account there were no differences between species ( $5.2 \pm 0.05 \ Q. \ ilex n=1140$ ;  $5.5 \pm 0.04 \ Q. \ robur$ , n=900 ( $\mu mol \ m^{-2} \ s^{-1}$ )) although *Q.ilex*  $A_n$  was significantly higher at the upland site compared with other sites ( $6.8 \pm 0.05$ ) and was also higher than *Q. robur* at this site, again this is in contrast to sapling data.  $N_{leaf}$  was higher in *Q. robur* at the urban site but this difference was not significant, however,  $C_{leaf}$  was lower at the urban site (this trend was only significant in *Q. ilex* leaves).

In mature trees,  $F_v/F_m$  was higher in *Q. ilex* than *Q. robur* at the rural site, the same at the upland site and lower at the urban site.  $F_v/F_m$  was also lower at the urban site compared to the other sites in *Q. ilex. PI* however was higher in *Q. ilex* than *Q. robur* at all sites. Within species differences meant that *Q. robur PI* was lower at the upland compared to the other sites. When different dates were compared,  $F_v/F_m$  and *PI* was significantly lower at the urban site in the drier June 2010 but increased in June 2011 (accounting for the difference in sample size for each year). For *Q. ilex* mean  $A_n$  was low at all sites in June 2010 and remained low at the urban site in June 2011, however March 2011  $A_n$  was significantly higher at all sites for this species. Despite this  $F_v/F_m$  was lower during March although  $F_v/F_m$  was not affected by drought as June 2010 and June 2011 had similarly high  $F_v/F_m$  values. However, *PI* was significantly lower during June 2010 and March 2011 for this species.

#### Intrinsic Water-use Efficiency (iWUE).

Mean intrinsic water use efficiency *iWUE* was lower in mature trees than in saplings, although maximum *iWUE* was higher in mature trees. In saplings *iWUE* was higher at the urban site, L, situated on the top of a south facing slope above Plymouth railway line at Laira in Plymouth. This site had the lowest percentage water content of all sites, (just 19 % in wet soil). L was also low in macronutrients (see chapter 3). The most productive sites were urban D and rural DT both had the highest *N*, *K* and relatively high *P* in soils at these sites. *iWUE* was greater in pots

than field grown saplings for all sites and species except at L where *iWUE* was lower in pots, in *Q. ilex* this was accompanied by a down regulation of  $A_n$ .

#### The effects of sun and shade.

For saplings, assimilation rates were higher in sun than shade leaves in both species, while  $F_v/F_m$  and *PI* were lower in sun than shade. Specific leaf area (*SLA*), tended to be lower in sun leaves of both species at the urban sites. However,  $N_{leaf}$  was greater in sun leaves at urban site D in both species, possibly an affect of atmospheric N disposition from nearby traffic at this site. In general there was less distinction between upper and lower foliage in young *Q. robur* saplings.

For mature trees, differences in leaf position in the canopy were more variable than differences between sun and shade, this was mainly due to site specific differences such as their position at the woodland edge (rural), and as parkland trees (urban and upland). Canopy density could also affect the variability within the results i.e. In the urban Q. *ilex* the upper canopy was more open and therefore there was less distinction between sun and shade within middle and upper canopy levels. Despite inconsistencies in sampling design, sun leaves had lower *SLA* and lower  $N_{leaf}$  than shade in both species although these differences were more pronounced in Q. *robur*. The same leaf traits were also lower in upper canopy leaves although the distinction was greater in shade leaves at this level of the canopy, possibly due to higher light availability in the upper canopy as a whole.

## Differences in Biomass for each species.

In saplings, *Q. ilex* had greater total biomass than *Q. robur* at all sites except the upland site. The largest differences between species were found at rural site B and urban site L.

In mature trees *Q. robur* had the largest biomass at rural and urban sites but contrary to sapling data, *Q. ilex* biomass was greater than *Q. robur* at the upland site. However, the variability in age between sites makes it difficult to make accurate comparisons between sites for individual species.

Q. ilex saplings allocated more biomass to foliage and branches, while Q. robur allocated more

to stem and roots.

# 8.1 Comparisons with the literature.

#### 8.1.1 Ontogenetic Differences.

Sapling  $A_n$  was higher than mature trees (4.36  $\pm$  2.5, 5.47  $\pm$  3.53 ( $\mu$ mol  $m^{-2} s^{-1}$ ) mature) compared with  $(7.37 \pm 4.29, 9.15 \pm 4.56 \ (\mu mol \ m^{-2} \ s^{-1})$  sapling) maximum  $A_n$  was higher in mature trees 9.22 ( $\mu$ mol m⁻² s⁻¹) Q. ilex, 21.68 ( $\mu$ mol m⁻² s⁻¹) Q. robur compared with 17 Q. ilex and 20.5 Q. robur in saplings. Maximum photosynthesis in Q. rubra was between three and six times higher in mature trees compared to young saplings depending on annual precipitation (as sapling photosynthesis declined during drought) (Cavender-Bares and Bazzaz 2000). Intrinsic water use, *iWUE* was higher in Q. *ilex* saplings but lower in Q. *ilex* as mature trees. However the urban environment had the effect of lowering *iWUE* in both life stages in both species. N content and LMA were also higher in mature tree leaves compared with saplings, similar results have previously been found in Q. ilex and deciduous oaks (Cavender-Bares and Bazzaz 2000; Mediavilla and Escudero 2003). In contrast to my results, comparisons between evergreen and deciduous oaks growing in the Mediterranean showed that mean  $A_n$ was not much different between Q. *ilex* seedlings and mature trees whereas in deciduous oaks, seedling  $A_n$  was much lower than in adult trees (Mediavilla and Escudero 2003). Deciduous oaks also had higher leaf  $N_{mass}$  in adult trees compared with seedlings, whereas there was no difference in leaf N_{mass} between ontogenetic stages in Q. ilex. iWUE was also lower in seedlings than adults in both evergreen and deciduous oaks (Mediavilla and Escudero 2003).

Previous research has also shown how seedling assimilation rates in the deciduous *Q. rubra* were a third of the level of mature trees during a wet photosynthetic season, additionally, water use efficiency (WUE) increased during drought as assimilation rates were lowered. Mature trees had a faster response to drought increasing WUE while maintaining assimilation rates, thus showing that mature trees are better able to withstand drought than seedlings due to deeper rooting depths (Cavender-Bares and Bazzaz 2000).

In southern California during drought, evergreen and deciduous oak seedlings aged 1-4 years

had lower  $F_v/F_m$ , lower assimilation rates and lower *iWUE* than mature trees situated nearby, in addition the evergreen *Q. agrifolia* had the lowest  $F_v/F_m$  which suggested the evergreen was under more stress than the deciduous *Q. lobata* (Mahall et al. 2009). However, analyses of my own data suggests that recovery from drought is much better if conditions improve the following year. Contradictions in the literature may be due to differences in sampling dates, but also many comparisons between evergreen and deciduous oaks within the literature are conducted in Mediterranean environments where seedlings are more likely to compete under dense evergreen canopies (Sanchez-Gomez et al. 2006).

#### Environmental effects on leaf traits.

Oak seedlings are especially vulnerable to drought during the first two years of establishment (e.g. Gomez-Aparicio et al. 2008). In my research, this effect was exacerbated by transplanting stress and soil fertility; therefore *Q. robur* saplings were affected by losses at the drier less fertile sites. Leaf  $N_{area}$  and *SLA* increased as rainfall decreased (Wright et al. 2004) while *SLA* can also decrease with decreasing mean annual temperatures (MAT) in evergreen broadleaved trees compared to a more conservative response in deciduous species (Wright et al. 2005). Low  $N_{leaf}$  and low *SLA* corresponded with low soil fertility and vice versa and when measured on a mass basis was related to MAT (Ordpnez et al. 2009). Therefore response in leaf traits of the saplings measured within my research show evidence of complex intrinsic and extrinsic responses in both species, the Mediterranean evergreen increases  $N_{leaf}$  at urban locations however this is accented when nutrient supply is good despite this species in generally having a more conservative response to nutrient supply (Valledares et al. 2000).

#### Differences in Biomass between species.

Q. *ilex* was only stressed at the upland site and reduced allocation to LWFt was evidence of this. Allocation to root was higher in Q. *robur* but interestingly as LWFt increased RWFt decreased suggesting trade-offs in allocation patterns. However, ontogeny can be a factor, when small, seedlings have a greater LMFt in order to improve carbon assimilation and RGR. As they grow allocation to LWFt decreases, with faster growing species having greater LWFt (Poorter et al. 2012). In general tree seedlings and young trees tend to allocate more of their biomass to leaves (Villar et al. 2006). The percentage that tree saplings may allocate to foliage has been estimated at up to 46%, dropping to just 6% in mature trees (Poorter and Evans 1998). Increased allocation to foliage in young trees can also be a reponse to stress factors (e.g. Niinemets 2010). A young tree may be more susceptible to short periods of stress than a mature tree that has more non structural carbon reserves to draw upon during such events (Niinemets 2010).

In theory plants should allocate carbon in order to increase productivity, therefore allocating to the fraction that will enable greater acquisition of a limiting resource (Weiner 2004). It follows that under low nutrient conditions root biomass should increase (Weiner 2004). However on-togenetic differences show that saplings will allocate more to roots and invest more resources into shoots as they mature (Weiner 2004). Shoot growth is more affected by drought than root growth in oaks (Arend et al. 2011). Young oaks adapt to drought by restricting canopy water loss through stomatal closure and as they mature increasing root depth, therefore ontogeny is important and this may change with age and provenance (Arend et al. 2011). Drought sensitivity was greater in more northerly provenance's of the same species in *Q. petrea*, *Q. robur* and *Q. pubescens*) (Arend et al. 2011). *Betula pendula* seedlings have been shown to alter allocation in roots and shoots in response to nutrients (Ericsson et al. 1996). To conclude, a review analysing data from the literature found that although *LMF* increased with nutrients, *SLA* was affected more, decreasing with increasing shade by 170% more than the corresponding change in *LMF* suggesting that plants are better able to change their morphology in response to environmental change than they are their allocation (Poorter et al. 2012).

#### Photo-inhibition.

Diurnal rhythms using the light unit suggested that down regulation of photosynthesis during the summer was driven by temperature. Similar results were found by Epron et al. (1992) in *Q. petrea* while another study showed *Q. robur* and *Q. petrea* were able to maintain assimilation rates and photochemical efficiency despite stomatal closure during summer drought (Epron and Dreyer 1993*a*). In contrast, *Q. ilex* winter data showed increased winter assimilation rates coupled with reduced photochemical quenching ( $F_v/F_m$ ). Although increased photosynthesis during mild winter temperatures is common in *Q. ilex*, leaves can become stressed at temperatures below 15 and above  $35^{\circ}C$  and combinations of high light and low temperatures can induce photo-inhibition (Larcher 2000).

Mature *Q. ilex*, growing naturally in Spain experienced reduced  $F_v/F_m$  during winter, a reduction in  $F_v/F_m$  can be attributed to a reduction in photochemical quenching in PSII and in this paper was attributed to photo inhibitory processes in the xanthophyll cycle activated without water stress (Martinez-Ferri et al. 2004). Increased evidence of photo inhibitory processes and resulting lowered  $F_v/F_m$  has also been shown in *Q. suber* growing in Southern Portugal where frosts are rare and average winter temperatures rarely dip below zero (Garcia-Plazaola et al. 1997). A similar response was also found in *Q. ilex* growing in Spain, (where average winter temperatures where around  $3.5^{\circ}C$ ) (GarcIa-Plazaola et al. 1999). In agreement with my research, maximum assimilation rates were not down regulated in response to the milder winter temperatures (between 5-10°*C*) despite reduced photochemical efficiency, evidence of a photo protective response (GarcIa-Plazaola et al. 1999). However, in my research, sapling data did show that at the upland site at Princetown, Dartmoor, and at the control site, monthly assimilation was reduced when mean temperature fell below 5 °|*C* (Figures 4.6, 4.9 and 4.9a and Figure 3.3 d for upland climate variables).

Low temperatures reduce assimilation in *Q. ilex* (Larcher 2000). In this research mean  $A_n$  was reduced in *Q. ilex* during winter at the upland site and (to a lesser extent) at the Dartmoor fringe site Broadley in saplings. Whereas, in mature trees, mean  $A_n$  was higher in March at all sites even though mean temperatures for that month were in fact below the optimum for *Q. ilex* assimilation at all sites  $(6.2 - 8.7^{\circ}C)$ . This suggests that drought and higher temperatures exert more control over *Q. ilex* than cold stress, or that *Q. ilex* is able to withstand temperatures as low as 5°C before photosynthesis is down regulated. As thylakoid membranes are sensitive to changes in the environment, changes in photochemistry are early indicators of temperature stress (Larcher 2000) and in mature trees  $F_v/F_m$  is reduced more so at the upland site, high light in conjunction with low temperatures can cause photo protective mechanisms thus reducing photosynthetic efficiency (Larcher 2000). On cold days, light harvesting is maintained to en-

sure high concentrations of caratenoids involved in the xanthophyll cycle (as a photo-protective measure), this in turn causes a consequent reduction in PSII efficiency. However, drought may also modify the effect of cold on growth (Crescente et al. 2002).

According to Larcher (2000), *Q. ilex* has a cold frost tolerance of  $-15^{\circ}C$  before visual damage occurs, compared to other Mediterranean vegetation. Interestingly in my research the June 2010 data (the drier June) showed no reduction in  $F_v/F_m$  at rural and upland sites despite a small reduction at the urban site, whereas *PI* is reduced at all sites during this drier June compared to June 2011 which may be evidence that the incorporation of different stages of the photochemical process in PSII recorded in the performance index *PI* (see chapter 2 section 2.3.6 for full interpretation of *PI*) is better at revealing signs of drought stress in plants.

In fact, despite  $F_v/F_m$ , as a measure of photosystem II efficiency being widely cited in the literature to quantify the effects of drought, the less often used, performance index (*PI*) (as a measure of vitality) has been shown as more susceptible to drought stress than maximum photochemical quenching ( $F_v/F_m$ ) a phenomenon that has been shown in wheat (Zivcak et al. 2008). Further, *PI* is a useful but little used plant stress parameter, but has been useful in quantifying decreased vitality in Beech leaves subjected to ozone pollution (Clark et al. 2000) and has also been used to indicate urban tree health (Hermans et al. 2003) although it was less useful than direct chlorophyll measurements at assessing health of natural oak stands suffering from defoliation (Rossini et al. 2006).

This suggests that use of both  $F_v/F_m$  and PI in determining the complex interactions between temperature and water stress would be useful addition to tree productivity research.

# 8.2 Conclusion.

Despite the ability of our native Q. *robur* to reach maximum assimilations rates in response to high summer temperatures, the trade off in water use efficiency leaves it vulnerable to drought. In contrast the evergreen Q. *ilex* is able to increase productivity in the mild oceanic climate of Southwest UK and the trade offs in maximum assimilation rates incurred from summer down regulation of the photosynthetic process mean Q. *ilex* is able to absorb more atmospheric carbon

over time than *Q. robur* at least at the sapling stage when it invests in greater foliage allocation. Greater productivity in *Q. ilex* at Castle Drogo, 300masl, is further evidence of the environmental plasticity of this species although for now at least *Q. robur* is still at an advantage on the high moor where instead it has to compete with *Q. petrea* on less fertile soils.

Finally, in my research the control site polytunnel saplings Q. *ilex* grew more quickly than Q. *robur*, Q. *robur* saplings suffered stress in the from *Erysiphe alphitoides* (powdery mildew) which almost halved max  $A_n$  in affected leaves. A reduction in maximum assimilation rates of around 30-40% in infected oak leaves was also found by (Hajji et al. 2009). In fact, Q. *robur* and in the more oceanic/humid areas of Q. *ilex* Mediterranean range, both these species are affected by pathogen attack (as discussed above). In mature trees, *Quercs ilex* did surprisingly well at castle Drogo upland site, where Q. *robur* was suffering from defoliation. The prescence of thin, nutrient poor soils seem to be reducing Q. *robur* resilience to attack at this site.

Further, *Q. ilex* is extending its range further north into central Europe and *Q. ilex* (sub sp. *ilex*) is already found along the Atlantic coast of France as far as Brittany (Barbero et al. 1992). However, caution should be taken when planting non native species, due to the competitive ability of a plant to colonise areas and out compete native vegetation for space, authors such as Peterken (2001) suggest *Q. ilex* is already a potential invasive species and the UK organisation, 'Plantlife' would like to see this species added to Schedule 9 of the Wildlife and Countryside Act in England and Wales (Plantlife 2010).

# 8.3 Limitations of this research

*Q. ilex* always had a nutritional advantage due to being pot grown compared to *Q. robur* bare root stock. However, in terms of cold hardening, *Q. robur* had the advantage. In turn, potted *Q. ilex* had greater productivity at sites where soil macro nutrients were scarce which would suggest *Q. ilex* can tolerate less fertile soils.

Planting one year saplings without watering in for the first season caused survival rates to be low at the more stressful sites for both species. With hindsight it may have been better to water in for the first Spring before leaving to take measurements the following growing season. In the mature tree experiment, although each site interspecies differences in age were minimum, the variability in age between sites and multi stem Q. *ilex* compared to single stemmed mature Q. *robur* at the upland site made biomass estimates more difficult for this life stage. Additionally, lack of mature Q. *ilex* at Princetown, Dartmoor meant the mature tree upland site was 100m lower in altitude than the sapling upland site, therefore comparisons between ontogenetic stage were also difficult.

## 8.4 Further Work.

In order to fully evaluate productivity and invasiveness potential of *Querus ilex*, germination success in relation to environment studies would need to be completed alongside more intensive work to draw out the effects of pollution from the effects of water stress. Closer analyses of pollution concentrations would make the distinction between the rural and urban environment easier to analyse. Although monthly climatic variables and soil information was gathered from sapling data. The number of different variables made it difficult to find meaningful relationships using multivariate analysis. It would be interesting to conduct a similar experiment using open top chambers in conjunction with field analyses. Additionally, there is evidence from this thesis to suggest that despite *Q. robur* having a wider European distribution, *Q. ilex* drought tolerance combined with its ability to withstand colder temperatures, warrants more research on this species growing within the UK as the majority of literature on this subject is mainly conducted within this species Mediterranean range.

Further work on biodiversity related projects is also advisable, analyses of biodiversity in the English versus Holm oak currently found in the literature found just five phytaphagus insect on *Q. ilex* compared to 243 on *Q. robur* (Kennedy and Southward 1984). The same authors reanalysed 20 years later and although the number of insects found on *Q. robur* had increased by 40, the number found on *Q. ilex* had increased to 89 (Southward et al. 2004) a nearly 20 fold increase.

# **Appendix A**

# **Soils Schedules**

[all schedules are taken direct from Geography lab schedules (Kev Sloman, 2009)].

# A.1 TOTAL N

# A.1.1 Total Nitrogen determination of soils.

The soil is digested under reflux with concentrated sulphuric acid in the presence of sodium sulphate (to raise the temperature of digestion) and copper catalyst (to promote oxidation of organic matter) to convert nitrogen compounds present to ammonium sulphate. The ammonia of the digest solution is then determined by continuous air-segmented flow colorimetry in which the sample is reacted with salicylate and dichloro-isocyanuric acid to produce a blue complex measured at 660nm. Nitroprusside is used as a catalyst.

**Reagents** Sulphuric acid (d20 1.84) Catalyst mixture. Mix together 200*g* of sodium sulphate, anhydrous and 6.0*g* of copper sulphate, pentahydrate.

Kjeldahl copper catalyst tablets. Each tablet contains 1g sodium sulphate and the equivalent of 0.1g copper II sulphate.

**Method** Allow to soak overnight a series of digestion tubes containing 10% sulphuric acid. There should be one tube per sample and two for blanks per batch.

Rinse the tubes out with distilled water and allow to dry.

Carefully weigh about 0.50g of air-dried, < 1.0mm soil into a dry digestion tube. Record the exact weight of the soil to the nearest 0.0001g.

Add  $1.0 \pm 0.05g$  of catalyst mixture (or one Kjeldahl copper catalyst tablet).

Cautiously add  $10.0 \pm 0.5 ml$  of sulphuric acid (d20 1.84).

Swirl gently to mix and insert the tube into the digestion block.

Repeat this procedure for all the samples and add two blanks (ie tubes with reagents only).

Programme the digestion block to gently heat to  $200^{\circ}C$  to remove any water present.

Increase the temperature to allow the contents of the tubes to boil under reflux. Maintain this temperature for 4 hours.

When digestion has completed allow the tubes to cool before removing them.

Carefully add 50*ml* distilled water to the tube. Ensure the tube is pointing away from the operator and anyone else nearby as the addition of water may cause the contents of the tube to react violently.

Transfer the contents to a 100ml volumetric flask and dilute to volume with water.

This diluted sample should be filtered or centrifuged prior to storage. Use a Whatman No.542 filter paper. Store in a polyethylene bottle in a fridge if not analysing immediately.

Analyse the ammonia on the Bran and Luebbe Autoanalyser.

Dilute as described in note 1 below.

When preparing the standard solutions, ensure they are acidified to the same concentration as the samples. The base matrix should contain  $10.0ml H_2SO_4$  per 100ml. This must be diluted to the same extent as the samples prior to making up the standards.

**Note 1:** Sodium salicylate precipitates in an acid medium. Do not try to neutralise the extracts with sodium hydroxide as this interferes with ammonia chemistry. To overcome the problems with acidified samples the autoanalyser should be run on the low sample range  $(0 - 5mgl^{-1})$  for salicylate chemistry and samples should be diluted 1:20. ie. For a standard 3ml sample, add 0.15ml sample to 2.85ml water.

Example digest programme (should be varied according to the nature of the sediment)

30 minutes @  $100^{\circ}C$ 

30 minutes @ 150°C

1 hour @  $200^{\circ}C$ 

2 hour @ 250°*C* 1 hour @ 360°*C* 1 hour @ 375°*C* Cooling time = 2 hours

**Calculation** The initial dry soil weight = *agrams* 

Volume of digest = 100ml

Concentration of Ammonia-*N* from AA3 = bmgl - 1

therefore in 100ml extract there is 100b??µgNH4+?N (this was present in agrams soil)

1Kg soil contains *mgtotal* -N (divide all this by 1000 to obtain grams TN)

For 1.0g soil and 100ml solution this reduces to;

1Kg soil contains bx0.10gtotal - N

# A.2 MINERAL N

#### A.2.1 Soluble Inorganic Nitrate (Mineral N)

Soils should be sampled moist from the field and extraction started within a few hours. Gently crumble the soil and omit any stones. Alternatively store at low temperature (20 - 40C) without drying or freezing (eg. stored in a sealed plastic bag in a refrigerator). This will restrict mineralization.

Potassium chloride is used to extract soluble and exchangeable nitrate and ammonium ions. Once in the potassium chloride solution, extracts can be stored for up to 2 months in a refrigerator.

**Reagent** Potassium chloride, 1*M*. Dissolve 74.5*g* of *KCl* in deionised water and make up to 1 litre.

Method Carefully weigh about 2.5g of the moist soil into a 125ml plastic bottle.

Record the exact weight of the soil.

Add 25ml of 1MKCl.

Cap the bottle securely and place on a rotary mixer for 1hour.

Filter the suspension through a Whatman 541 or Whatman no.1 filter paper. Do not use a membrane filter. If the soil contains a high percentage of clays it may save time if the suspension is first centrifuged for 5*minutes* at 3500*rpm*.

Store the filtrate for nitrate analysis and discard the sediment fraction.

Analyse the filtrate using the Bran and Luebbe Autoanalyser 3. Ensure the correct programme is run for the soil nitrate extract.

All standards, wash solutions and dilution waters must be made of the same extractant solution - ie. 1M KCl.

Determine the moisture content of the original soil using the standard method and use the corrected dry soil weight for the calculations.

**Calculation** The initial dry soil weight = a?grams Volume of KCl extractant = 25ml Concentration of Nitrate from AA3 = bmgl - 1

therefore in 25ml extract there is 25b??gN

This was present in *a*?grams soil 1Kg soil mg N (divide all this by 1000 to obtain grams N)

For 2.5g soil and 25ml KCl this reduces to; 1Kg soil contains b x 0.01g N (Kev Sloman, 2003)

# A.3 AVAILABLE PHOSPHATES

#### A.3.1 Available Phosphate-P extractions from soils

(Brays No.1 extractant)

This method is ideally suited to acidic soils. The selectivity of the extractant is designed to remove the easily acid-soluble P and a portion of the phosphates of Al and Fe. For calcareous soils follow the method using sodium bicarbonate as extractant. The method uses Brays No.1

extractant to remove the available phosphates. This method is sensitive to the soil/extractant ratio, shaking rate and time. It is therefore important that these are kept the same for all samples.

**Reagents** Bray No.1 extractant. Dissolve 2.22*g* ammonium fluoride,  $NH_4F$ , in about 1*litre* of deionised water and add 50*ml* of 1.0*NHCl*. Dilute to 2*litres*. The solution *pH* should be 2.65 ± 0.5. Store in a polyethylene bottle.

This solution comprises 0.025NHCl and  $0.03NNH_4F$ .

**Reagent A.** Dissolve 60*g* ammonium molybdate tetrahydrate in about 200ml distilled water. Add 1.455*g* antimony potassium tartrate and mix to dissolve. Slowly add 700*ml* concentrated sulphuric acid.

Carefully dilute to 1*litre* with distilled water when cooled. This reagent should be dark blue in colour and is stored in an amber glass bottle in a fridge.

Ascorbic acid reagent. Dissolve 33g L(+) ascorbic acid in 250ml distilled water.

**Working Reagent B** Pipette 25*ml* of Reagent A into a 1*litre* volumetric flask containing about 800ml deionised water and add 10ml ascorbic acid reagent. Dilute to 1*litre* with deionised water. This solution must be prepared daily.

**Method** Carefully weigh about 1.0g of air-dried, < 2mm soil into a 50ml plastic bottle. Record the exact weight of the soil. Add 10ml of Bray extractant reagent. Cap the bottle securely and place on a rotary mixer for 15 minutes. Leave to stand for 15 minutes. Filter the suspension through a Whatman *No*.540 or 541 filter paper. Alternatively centrifuge the mixture at 3500rpm for 5minutes and decant off the supernatant fraction for analysis. Measure a 2ml aliquot of the filtered fraction into a glass test tube and add 8ml of the working reagent B. Leave to stand for 20minutes for the colour to develop. Analyse the prepared sample on a spectrophotometer at 880nm wavelength.

**Calibration** Prepare a set of standard solutions in the range 0 - 8mgl - 1P using Bray's reagent as the base matrix. Measure a 2ml aliquot of the known standard into a glass test tube
and add 8ml of the working reagent B. Leave to stand for 20minutes for the colour to develop. Analyse the prepared standard on a spectrophotometer at 880nm wavelength. A blank solution can be prepared by pipetting 2ml of the extractant into a test tube and adding 8ml of the working reagent B. Plot a linear calibration curve of concentration of phosphate against absorbance.

**Calculation** The initial dry soil weight = ??grams Volume of extractant = 10ml Concentration of Phosphate from spectrophotometer = ?mgl - 1 therefore in 10ml extract there is 10???gP This is present in ??grams soil 1Kg soil contains ?mgP (divide all this by 1000 to obtain grams P)

For 1.0g soil and 10ml extractant this reduces to; 1Kg soil contains 2x0.01gP

#### A.4 POTASSIUM

#### A.4.1 The measurement of exchangeable potassium in soils

**Reagents** Use one of the following acetate preparations;

Ammonium acetate solution, 1*M*. Add 115*ml* glacial acetic acid to about 500ml deionised water. In another vessel add 110ml ammonia solution  $(35\% NH_3)$  to 500*ml* deionised water. Carefully mix the solutions together in a 2*litre* volumetric flask and adjust to *pH*7.0 with acetic acid or ammonia. Dilute to 2*litres* with deionised water.

Ammonium acetate solution, 1M. Dissolve 77.08g ammonium acetate in 800ml deionised water and dilute to 1*litre*.

Stock standard solution, 1000mgl - 1 potassium. Dissolve 1.907g potassium chloride in approximately 500ml1M ammonium acetate and dilute to 1litre with 1M ammonium acetate.

Working standard solution, (10mgl - 1K+). Dilute the stock standard 1 : 100 with 1*M* ammonium acetate.

**Preparation of extracts** Weigh  $2.5g(\pm 0.01g)$  of <2mm air-dried soil into a 125*ml* HDPE plastic bottle and record the weight.

Add 100ml ammonium acetate and leave for 4 hours or overnight.

Place on a rotary mixer for 30 minutes.

Filter through a Whatman No.540 or 542 filter paper. Use vacuum filtration if necessary.

Pour the filtrate into a 100ml volumetric flask and top up to the mark with 1M ammonium acetate. Store in the cleaned plastic bottle.

This extract is used for the determination of exchangeable potassium.

Potassium can be analysed using standard flame photometry or ICP instrumentation.

**Notes** Store prepared samples in a fridge at  $4^{\circ}C$ .

Never leave samples unattended for more than 2 days at room temperature as the extractant can act as a nutrient base for mould growth.

Use the standard extractant as a blank.

If samples require diluting, ensure the matrix in the final sample corresponds with the original extractant.

(Kev Sloman, version 1.3)

### **Appendix B**

### **Residual Plot Information**

### **B.1 Biomass data: Residual Information**

# B.1.1 Biomass data; Information on residuals to show assumptions for Linear model affectiveness met

The following figures are residual plots from the biomass fractions: Analysis of residuals can be used to justify the assumptions that need to be met to ensure a linear model is adequate analysis.



Figure B.1: Residual plot for leaf weight fraction minus control site.



Figure B.2: Residual plot for root weight fraction.



*Figure B.3:* Residual plot for stem weight fraction.



*Figure B.4:* Residual plot for branch weight fraction.

Appendix C

# **Interaction Plots.**

C.1 Final Above Ground and Total Biomass and fractions Interaction Plots



Figure C.1: Interaction plots for biomass fractions; differences between pots and planted saplings.



Figure C.2: Above ground biomass, interactions between species and condition.



Figure C.3: Above ground biomass interactions with site and condition.



Figure C.4: Total Biomass interactions with species and condition



Figure C.5: Total biomass, Interaction plot



*Figure C.6:* SPECIES and SITE Interactions for Biomass Fractions LWF, BWF, SWF, RWF, Analysed as percentage of TB.



Figure C.7: AGB, interactions between SPECIES and SITE



Figure C.8: TB interactions with SPECIES, SITE and CONDITION



Figure C.9: Percentage allocation of BWFt on logged scale, full interaction plot



Figure C.10: Percentage allocation of LWFt on logged scale, SPECIES and SITE interactions



Figure C.11: Percentage allocation of RWFt on logged scale, SPECIES and SITE interactions.



Figure C.12: Percentage allocation of SWFt on logged scale, full interaction plot.

Appendix D

## **Leaf Trait Interactions with Factors**

D.1 Box and Whiskerplots for Leaf traits against factors





*Figure D.1:* Leaf carbon data for species, site and environment (SPECIES, SITE and CHAR-ACTER) (means  $\pm$  CI)



(a) SLA

#### D.1.1 Leaf trait interaction plots



*Figure D.4:* Leaf N (Nresult in % concentration). Showing Species, site and LP (UP-PER/LOWER) interactions (see glossary for site coding) (means ±95%CI)



*Figure D.5:* Leaf C(CResult in % concentration) Showing Species, site and LP (UPPER /LOWER) interaction(means ±95%CI)



*Figure D.6:*  $SLA(cm^2g^{-1})$ . Showing Species, site and LP (UPPER /LOWER) interaction(means  $\pm 95\% CI$ )

### **Appendix E**

### **Biomass: Descriptive statistics and extra graphs**

### E.1 Actual growth rates.

The following graphs are larger individual growth/time plots for each site without homogeneous y axes, therefore enabling a clearer interpretation of the results shown in chapter 6 Figure 6.1.



*Figure E.1:* Monthly growth for control site saplings over the 33 month growth period (means  $\pm$  95 % CI).



*Figure E.2:* Monthly growth for upland site saplings over the 33 month growth period (means  $\pm$  95 % CI).



*Figure E.3:* Monthly growth for rural site B saplings over the 33 month growth period (means  $\pm$  95 % CI).



*Figure E.4:* Monthly growth for rural site DT saplings over the 33 month growth period (means  $\pm$  95 % CI).



*Figure E.5:* Monthly growth for urban site D saplings over the 33 month growth period (means  $\pm$  95 % CI).



*Figure E.6:* Monthly growth for urban site L saplings over the 33 month growth period (means  $\pm$  95 % CI).

#### E.2 Relative growth rates (RGR).

The following figures are monthly and season relative growth rates for each species.



*Figure E.7:* Relative growth rates in control site potted saplings (means  $\pm$  95 % CI). Data shown is approximately two monthly but see axis for exact increments.


*Figure E.8:* Relative growth rates in upland site field grown saplings (means  $\pm$  95 % CI). Data shown is approximately two monthly but see axis for exact increments.



*Figure E.9:* Relative growth rates in rural site B, field grown saplings (means  $\pm$  95 % CI). Data shown is approximately two monthly but see axis for exact increments.



*Figure E.10:* Relative growth rates in rural site DT, field grown saplings (means  $\pm$  95 % CI). Data shown is approximately two monthly but see axis for exact increments.



*Figure E.11:* Relative growth rates in upland site field grown saplings (means  $\pm$  95 % CI). Data shown is approximately two monthly but see axis for exact increments.



*Figure E.12:* Relative growth rates in urban site L field grown saplings (means  $\pm$  95 % CI). Data shown is approximately two monthly but see axis for exact increments.

### E.3 Biomass descriptive statistics

The following two tables (tables E.1 and E.2) give descriptive statistics for the biomass variables given in chapter 6.

Sapling type: GROUND					Q	. ilex					Q. 1	robur		
Growth parameter	Character	SITE	z	Mean	StDev	Min	Max	Skew	z	Mean	StDev	Min	Max	Skew
Leaf Biomass (g)	RURAL	в	10	58.33	21.24	18.6	95.85	-0.04	9	7.58	3.78	1.88	12.72	-0.3
		DT	10	139.7	72.7	34	247.1	0.01	10	61.6	34.4	21.3	129.7	0.86
	URBAN	D	8	78.9	31.4	40.7	127.9	0.21	10	31.03	18.69	4.12	60.78	-0.05
		L	11	55.4	36.6	20.8	128.9	1	ω	12.77	8.59	4.6	21.72	0.41
	UPLAND	DPF	9	4.46	3.69	0.48	12.18	1.2	17	2.704	1.403	0.027	5.472	0.04
Branch Biomass (g)	RURAL	в	10	40.46	25.3	8.17	85.49	0.9	9	4.88	4.69	0.48	15.62	1.69
		DT	10	97.8	61	30.8	209	0.56	10	08	64.5	22.3	243.6	2.12
	URBAN	D	8	75.6	35	26.8	139.7	0.54	10	33.08	20.86	2.85	51.57	-0.73
		L	11	59.7	57.9	11.1	184.4	1.22	ω	8.93	6.01	5.13	15.86	1.71
	UPLAND	DPF	9	1.964	1.263	0.118	4.156	0.37	17	0.6204	0.3406	0.174	1.222	0.44
Root Biomass (g)	RURAL	в	10	78.23	29.75	20.45	121.11	-0.54	9	22.1	13.69	7.73	42.96	0.44
		DT	0	111.2	38.9	62.9	161.6	-0.18	10	95.5	4	41	179.5	1.06
	UKBAN	- נ	± ∝	111.0 71 0	32.b	00.I	175.3	-0.68	3 IO	31.I	0 00 2 C	10.22	191.9 26 12	0.72
	UPLAND	DPF	9	8.83	3.53	3.63	13.05	-0.53	17	7.342	3.69	3.049	16.559	1.02
Stem Biomass (g)	RURAL	в	10	47.91	23.69	15.52	87.24	0.5	9	7.61	5.01	2.59	15.47	0.67
		DT	10	132.2	59.3	51.6	212.5	0.21	10	105.5	67	35.6	248.6	1.36
	URBAN	- D	:∞	50.3	20.9	1 o 1	164.7	-0.52	3	18.4	53.1	6 8 8	167.9 20 77	0.4
	UPLAND	DPF	9	3.912	1.616	1.817	7.01	0.56	17	3.296	1.707	1.017	6.813	0.65
		J	5	0 / 00	7 70	Г С Э	1 026	001	0	10 10	07 10	10 15	1 6 02	FO 0
Q		DT	10	480.9	183	187.9	814.8	0.09	10	327.8	186.5	116.1	722.2	1.34
	URBAN	D	8	363.1	155.9	138.5	624.6	-0.03	10	223.6	138.8	33.8	468	0.11
		L	11	239.4	171.5	57.4	567	0.77	ω	66.6	37	37	108.1	1.29
	UPLAND	DPF	9	19.17	8.61	7.31	31.44	-0.19	17	13.96	5.83	5.56	26.73	0.34
Aboveground Biomass (g)	RURAL	в	10	146.7	67.8	42.3	268.6	0.58	9	20.08	12.14	4.95	40.59	0.41
		DT	10	369.6	154.2	116.4	668.6	0.2	10	232.4	146.4	75.1	542.8	1.31
	URBAN	Ð	~ ~	251.5	124.8	78.4	468.3	0.18	, 10	142.5	90	15.1	276.1	-0.08
			Ì	169.8	124.7	52.9	431.1	1.05	ťω	39.4	28.8	17.2	91.9	1.41
	UPLAND	DPF	9	10.33	5.59	2.41	19.93	0.34	17	6.621	3.03	2.514	11.102	0.13
mean $HD^2$ (cm ³ )	RURAL	в	10	2526	1280	893	4997	0.73	9	502	368	117	1120	0.61
		DT	10	9413	4583	3216	17846	0.59	10	8023	7410	1628	26664	2.06
	URBAN	D	~ ~	6616	3119	1863	11983	0.39	<u>,</u> 10	8237	6598	464	21139	0.79
			0 [	3/69	2844 010	63 1062	2002 2002	1.01 2 07	17 5	194 y 186 y	11/ 8	575	3463	0 77
	OFLAIND	DEL	2	405	910	70	7067	2.31	1	100.0	114.0	ر، ر	402.4	0.77

DEMELZA CARNE

÷.
tree
<u>в</u> )
012
st 2)
ngn
), A
DTS
<u></u>
ngs
sapli
МN
gro
pot
for
ters
ame
para
vth
grov
and
ass a
om
l bi
Finê
7
Ē
ple
Ta

apling type: POTS		CITE	Z	Moon	C4Dair	). ilex	Moi	C1	Z	Moon	Q.	robur	More	Close
Variable		NIE	z	Mean	SUDEV	MIN	Max	Skew	z	Mean	SUDEV	MIN	Max	Skew
Leaf Biomass (g)	RURAL	шŗ	6 1	85.7	57.5	18.9 15 1	176 257 2	0.28	10	15.74	7.72	3.13 10.1	28.87	-0.25
		11	- `	C.16	101.2	1.61	5.107	0./1	, <i>ح</i>	77.14	70.C2	10.1	90.45	7 7
	URBAN	D	9	129.1	70.3	39.4	228.7	0.11	10	27.48	14.13	9.13	56.6	0.92
		Γ	4	78.9	65.2	16	166.8	0.96	10	14.36	6.72	6.42	27.12	0.76
	UPLAND	DPF	×	10.63	19.97	1.06	59.64	2.73	12	13.62	15.12	0.39	53.07	2.09
	CONTROL site	C	8	52.17	12.34	36.63	70.2	-0.03	٢	12.86	6.29	5.08	21.27	0.15
Branch Biomass (g)		þ	c	с Г <u>ч</u>	7		0 2 1 1	30.0	0		00 C		10.05	
	KUKAL	μ	ות	5./C	41.4	10./	7.011	C7.U	10	1.14	5.88	00.7	12.90	-0.24
		DT	-	102.4	123	8.8	323.9	1.08	6	38.5	32.2	10.6	114.4	1.93
	URBAN	D	9	116.5	71.1	19.8	212.9	0.02	10	31.81	28.69	9.6	99.93	1.88
		Γ	4	87.4	93.2	15.5	215.9	1.21	10	10.41	3.76	5.43	17.49	0.97
	UPLAND	DPF	×	7.69	12.66	0.88	38.4	2.62	12	9.25	12.41	0.17	40.08	2.01
Root Biomass (g)	CONTROL site	C	×	33.07	8.09	18.69	44.54	-0.62	2	17.7	33	1.2	92	2.58
	RURAL	В	6	128.2	78.5	47.3	234.7	0.27	10	37.71	14.39	11.89	56.43	-0.96
		DT	٢	108.2	109.3	20	283.5	1.19	6	78.1	74.2	33.4	267.4	2.56
	URBAN	D	9	161.7	93.1	40.5	246.1	-0.61	10	98.5	52.6	58.6	232.5	2.17
		Γ	4	157.4	7.66	39.2	279	0.09	10	60.18	29.62	37.15	128.02	1.63
Stem Biomass (g)	UPLAND	DPF	×	25.3	31.8	5.5	100.9	2.43	12	36.6	37.6	1.7	138.3	2.08
	CONTROL site	C	×	91.04	20.77	62.31	119.79	-0.28	Г	35.01	13.17	17.35	50.34	-0.58
		ç	c		с <u>у</u> с		1 201	01 0	0	02 01		ī	00 20	
	NUNAL	۹ ¦	ות	0. <del>1</del> .	C.UC	C.U1	100.4	0.49	21	12.30	00.1	1. 't	66.07	17.0
		DT	-	106.9	9.111	13.8	278.1	0.64	6	62.2	63.9	7.1	208.8	1.71
	URBAN	D	9	114.3	52.2	33.2	157.8	-1.06	10	48.47	23.19	14.8	93.03	0.5
Total Biomass (g)		Γ	4	85.9	77.4	14.5	173.4	0.26	10	16.92	11.04	7.4	35.32	0.88
	UPLAND	DPF	×	9.12	14.17	1.88	43.97	2.76	12	16.69	17.84	1.11	65.2	2.16
	CONTROL site	U	×	43.28	10.45	30.93	59.25	0.22	2	7.08	3.11	2.98	12.3	0.68
	RURAL	в	6	320.4	172.7	108.9	606.6	0.13	10	73.78	29.96	21.74	115.77	-0.99
		DT	٢	406	391	59	862	0.38	6	205.5	185.7	54.9	606.9	2.28
boveground Biomass (g)	URBAN	D	9	519	277	134	801	-0.54	10	203.6	114.2	93.1	482.1	1.8
		Γ	4	426	357	85	900	0.87	10	102.4	37.4	65.7	164	0.68
	UPLAND	DPF	×	53.4	78	12	242.9	2.63	12	76.1	80.5	3.4	296.6	2.16
	CONTROL site	C	×	218.7	46.5	151.1	269	-0.61	٢	72.6	41.2	29.5	150.5	1.04
	RURAL	В	6	192.2	130.7	49.8	373.4	0.4	10	36.07	18.22	7.71	63.92	-0.27
mean $HD^2$ $(cm^3)$		DT	٢	298	318	39	800	0.68	6	127.4	116.6	21.5	399.6	1.84
	URBAN	D	9	357.6	187.3	78.7	555.1	-0.56	10	105.1	63	34.6	249.6	1.43
		Γ	4	268	261	46	621	1.03	10	42.2	19.08	26.29	70.65	0.84
	UPLAND	DPF	×	28.1	46.5	6.3	142	2.73	12	39.6	43.9	1.7	158.3	2.17
	<b>CONTROL</b> site	U	8	127.7	29.9	88.8	167	-0.31	2	37.6	37.2	12	117.9	2.17

### Appendix F

## **Scattergraphs**

#### F.1 Scatter graphs for biomass and stem volume relationships.

The following graphs give scatter relationships between biomass and stem volume for all sites and combinations and clearly show the linear relationship between the two variables. Sites where saplings have experienced stress such as *Q. robur* at the least fertile and drier urban site L and *Q. ilex* growing at the upland site DPF have weaker relationship than those saplings growing at the more fertile sites where water availability is greater (see figure F.2).



Figure F.1: Relationship between total biomass TB and stem volume  $HD^2$ , initial harvest₁.



*Figure F.2:* Relationship between total biomass *TB* and stem volume  $HD^2$ , grouping for pots (P) versus field grown (G) saplings ('I' = *Q. ilex* and 'R' = *Q. robur*, sites: 'C' = Control Site, 'DPF' = Upland, rural sites, 'B' = Broadley, 'DT' = Dartington, and urban sites, 'D' = Digginit and 'L' = laira)



*Figure F.3:* Relationship between aboveground biomass *AGB* and stem volume  $HD^2$ , grouping for pots (P) versus field grown (G) saplings ('I' = *Q. ilex* and 'R' = *Q. robur*, sites: 'C' = Control Site, 'DPF' = Upland, rural sites, 'B' = Broadley, 'DT' = Dartington, and urban sites, 'D' = Digginit ant D = 1



*Figure F.4:* Relationship between total biomass TB and stem volume  $HD^2$  on ln scale. Grouping for pots (P) versus field grown (G) saplings ('1' = Q. *ilex* and 'R' = Q. *robur*, sites: 'C' = Control Site, 'DPF' = Upland, rural sites, 'B' = Broadley, 'DT' = Dartington, and urban sites, 'D' = Digginit and 'L' = laira)



*Figure F.5:* Aboveground biomass *AGB* and stem volume  $HD^2$  on a *ln* scale. Grouping for pots (P) versus field grown (G) saplings ('I' = *Q. ilex* and 'R' = *Q. robur*, sites: 'C' = Control Site, 'DPF' = Upland, rural sites, 'B' = Broadley, 'DT' = Dartington, and urban sites, 'D' = Digginit and 'L' = laira)



*Figure F.6:* Relationship between leaf weight fraction LWFt and root weight fraction RWFt allocation patterns for *Q. ilex* 'I' and *Q. robur* 'R' at different sites

### **Appendix G**

# Information on Light levels and filter densities for Light Response Curves

Light-unit + 0.1 + 0.2 + 0.3 ND Filters: P.A.R.  $\approx$  100 mols m²/s.

Light-unit + 0.1 + 0.2 ND Filters: P.A.R.  $\approx$  180 mols m²/s.

Light-unit + 0.1 + 0.3 ND Filters: P.A.R.  $\approx 225$  mols m²/s.

Light-unit + 0.3 ND Filters: P.A.R.  $\approx 250$  mols m²/s.

Light-unit + 0.2 ND Filters: P.A.R.  $\approx$  340 mols m²/s.

Light-unit + 0.1 ND Filters: P.A.R.  $\approx$  410 mols m²/s.

Light Unit without ND filter: P.A.R.  $\approx 500$  P.A.R. mols m²/s.

Natural Light: 800-1800 P.A.R. (depending on weather).

## Appendix H

# Mature trees: extra graphs.

The following figures show differences between dates for both Q. robur and Q. ilex for photosynthetic variables, no data for leaf trait SLA,  $N_{leaf}$  and  $C_{leaf}$  shown.



(a) Q. robur iWUE, Date



(b) Q. ilex iWUE, Date

*Figure H.1:* Differences in canopy level intrinsic water use efficiency (iWUE) between species and dates (means  $\pm$  95 % CI).



(a) Q. robur Gs, Date



(b) Q. ilex Gs, Date

*Figure H.2:* Canopy *Gs*:Differences between dates (means  $\pm$  95 % CI).



(a) Q. robur  $A_n$ , Date



(b) Q. ilex  $A_n$ , Date

*Figure H.3:* Canopy  $A_n$ :Differences between dates (means  $\pm$  95 % CI).



(a) Q. robur  $F_v/F_m$ , Date



(b) Q. ilex  $F_v/F_m$ , Date

*Figure H.4:* Canopy  $F_v/F_m$ :Differences between dates (means  $\pm$  95 % CI).



(a) Q. robur PI, Date



(b) Q. ilex PI, Date

*Figure H.5:* Canopy *PI*:Differences between dates (means  $\pm$  95 % CI).

## Glossary.

 $\mu g m^3$  micrograms per cubic metre

- $A_n$  Net photosynthetic rate measured in  $\mu$  mol  $m^{-2}$   $s^{-1}$
- $A_{max}$  maximum photosynthetic rate,  $\mu mol \ m^{-2} \ s^{-1}$ .
- C atmospheric Carbon
- *CF* Correction factor for the back transformation of a logged power function
- *GHG* Greenhouse gas; a gas that can absorb and re-emit infrared radiation, thus adding to the greenhouse effect (warming of the lower atmosphere as *GHG* re-emit infrared radiation back towards the earths surface
- Gs stomatal conductance of  $CO_2$  (Gs) measured in mol  $m^{-2} s^{-1}$
- *iWUE* Intrinsic water use efficiency, measured as  $A_n/Gs$  in  $\mu mol \ CO_2 / mol \ H_2O^{-1}$
- *LAR* Leaf Area Ratio,  $(m^2g^{-1})$
- *LMA* leaf mass per unit leaf area, measured as  $1/SLA mg cm^{-2}$
- $Log_e$  or Ln Natural log of a number is the power to which e would have to be raised to get that number where  $e \approx 2.7182$
- *LWR* Leaf Weight Ratio,  $gg^{-1}$
- mg One milligram me = 0.001g.
- *RGR* mean relative growth rate, modeled as the difference between mean of weight at harvest at time 2 ( $_2t$ ) minus mean of weight of harvest at time 1 ( $_1t$ ) divided by the difference between  $_2t$  and  $_1t$  and measured in  $g \, day^{-1}$
- AGB above ground biomass
- AK available potassium in soils
- Anisohydric behaviour. The'risk taking' action of keeping stomata open to increase assimilation rates despite lowered water availability and leaf water potential (see (e.g., Sade et al. 2012))
- Aoil Auger. Special hand drill for removal of soil layers

- AP available phosphate in soils
- B Factor SITE: Broadley rural allotment, near South Brent Devon
- BEF Biomass expansion factor: additive allometric equations that use the relationship between SW and different biomass fractions (e.g. branch, root, foliage) to model TB and AGB from DBH and H parameters

BRP A combination of factors: SITE, SPECIES and CONDITION

- CS Factor SITE: Control site, poly-tunnel situated in SE Cornwall
- CYM Current year mature leaf
- D Factor SITE: Digginit Community allotment, urban site located in Plymouth
- DPF Factor SITE: Dartmoor Prison farm, Upland site
- DT Factor SITE: Dartington, rural allotment near Totnes, Devon
- E Mean transpiration rate (*E* measured in *mmol*  $m^{-2} s^{-1}$

Flow-rate Rate at which CO₂ is passed through the leaf chamber of the IRGA

- G Factor CONDITION; field grown saplings
- g grammes
- GLM General Linear Model, using least squares regression and allowing multiple factors to be analysed against a dependent variable and assuming homogeneity of variances and normal distribution
- GtC Gigatonnes of atmospheric Carbon
- I Factor SPECIES *Quercus ilex* subsp. *ilex*, common name: Holly or Holm oak
- IRGA Infra-red Gas Analyser
- L Factor SITE: Laira, urban allotment, located in Plymouth
- LA Leaf area,  $cm^2$
- LAR Leaf area ratio: Leaf area/ dry plant mass
- MAT Mean Annual Temperatures.
- mg Milligrammes in weight
- N mineral nitrogen, insoluble nitrogen in soils

- NL Newly formed, immature current year leaf
- Oceanic warm temperate, fully humid with warm (not hot) summers, mild winters and evenly distributed precipitation

Ontogenetic Relating to growth and developmental stages in an individuals life

- P Factor CONDITION; Saplings planted in pots
- P.A.R. Photosynthetically active radiation; the light that plants can absorb in order to carry out the functions of photosynthesis, from the electro magnetic spectrum, visible light consisting of wavelengths between 400-700 nanometers (nm)
- pH Acidity/alkalinity of soils
- Phytaphagus e.g. A phytaphagus insect feeds off plants
- Plasticity The ability of a species to cope with changes in its normal environmental conditions; a more plastic species being able to tolerate changes better than one with less plasticity (this may or may not be subject to selection)
- PNUE Photosynthetic Nitrogen Use Efficiency, An / Nleaf ratio
- ppm Parts per million
- R Factor SPECIES Quercus robur, common name: pedunculate or English oak

Root/Shoot ratio. Ratio of root mass divided by stem and/or total aboveground biomass

Sapling. Young tree. In this research between one and four years in age

- SLA Specific leaf area; the ratio of leaf area to leaf dry weight  $cm^2g^{-1}$
- Specific gravity. Specific wood gravity determines how much of a trees structural growth is attributed towards strength and support (Williamson and Wiemann 2010) and is determined by dividing oven dry mass (ODM) by water displaced by the same sample when green, or measured as density, as ODM divided by green volume  $(g^{-1}cm^3)$  (Muller-Landau 2004)
- TB total biomass
- TLA total leaf area
- TN Total nitrogen content in soils

## List of references.

- Abrams, M. D. (2011), 'Adaptatons of forest ecosystyems to air pollution and climate change', *Tree Physiology* **31**, 258–261.
- ADC Bioscientific (2007), 'Lci portable infra-red gas analyser (irga)'.
- Alcamo, J., Kreileman, G. J. J., Krol, M., Leemans, R., Bollen, J., Van Minnen, J., Schaefer, F., Toet, S. and De Vries, B. (1998), 'Global modelling of environmental change: An overview of image 2.1', *Global change scenarios of the 21st century Results from the IMAGE 21 model* pp. 3 – 94.
- Alcamo, J., Moreno, J. M., Nováky, B., Bindi, M., Corobov, R., devoy, R. J. N., Giannakopoulos, C., Martin, E., Olesen, J. E. and Shividenko, A. (2007), Europe. climate change 2007: Impacts, adaptation and vulnerability. contribution of working group ii to the fourth assessment report of the intergovernmetal panel on climate change, Technical report, Cambridge University Press, Cambridge, UK.
- Alfani, A., Baldantoni, D., Maisto, G., Bartoli, G. and Virzo De Santo, A. (2000), 'Temporal and spatial variation in c, n s and trace element contents in the leaves of *Quercus ilex* within the urban area of naples', *Environmental Pollution* **109**, 119–129.
- Alves, P., aes, A. C. N. M. and Barja, P. R. (2002), 'The phenomenon of photoinhibition of photosynthesis and its importance in reforestation', *The Botanical Review* 68, 193–208.
- Amthor, J. S. (1995), Higher plant respiration and its relationship to photosynthesis, *in* E. D. Schulza and M. M. Caldwell, eds, 'Ecophysiology of Photosynthesis', Springer-Verlag, Berlin, chapter 4.
- Amthor, J. S. (1999), Increasing atmospheric co₂ concentration, water use, and water stress: Scaling up fro the plant to the landscape, *in* Y. Luo and H. A. Mooney, eds, 'Carbon Dioxide and Environmnetal Stress', Academic Press, London.
- Amthor, J. S. (2000), 'The mccree de wit penning de vries-thornley respiration paradigms: 30 years later', *Annals of Botany* **86**, 1–20.
- Anon (2003), *Good Practice Guidance for Land Use, Land-Use Change and Forestry*, Special Report of the Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge.
- Anon (2008), 'Air quality progress report for plymouth april 2008', Plymouth City Council. URL: *http://www.plymouth.gov.uk*

- Antunez, I., Retamosa, E. C. and Villar, R. (2001), 'Relative growth rate in phylogenetically related deciduous and evergreen woody species', *Oecologia* **128**, 172–180.
- Arend, M., Kuster, T., Gunthardt-Goerg, M. S. and Dobbertin, M. (2011), 'Provenance-specific growth responses to drought and air warming in three european oak species (*Quercus robur*, *Q. petraea* and *Q. pubescens*)', *Tree Physiology* **31**(3), 287–297.
- Baldocchi, D. D., Ma, S., Rambal, S., Mission, L., Ourcival, J.-M., Limousin, J.-M., Pereira, J. and Papale, D. (2010), 'On the differential advantages of evergreenness and deciduousness in mediterranean oak woodland, a flux perspective', *Ecological Applications* 20(6), 1583–1597.
- Barabaroux, C. and Beda, N. (2002), 'Contrasting distribution and seasonal dynamics of carbohydrate reserves in stem wood of adult ring-porous sessile oak and diffuse-porous beech trees', *Tree Physiology* 22, 1201–1210.
- Barbero, M., Loisel, R. and Quézel, P. (1992), 'Biography, ecology and history of mediterranean quercus ilex ecosystems', *Vegetatio* **99-100**, 19–34.
- Baskerville, G. (1972), 'Use of logarithmic regression in the estimation of plant biomass.', *Canadian Journal of Forst Research* **2**, 49 53.
- Bassow, S. L. and Bazzaz, F. A. (1998), 'How environmental conditions affect leaf-level photosynthesis in four deciduous tree species', *Ecology* **79**(8), 2660–2675.
- Bateman, I. J. and Lovett, A. A., eds (2000), Modelling and Valuing Carbon Sequestration in Softwood and Hardwood Trees, Timber Products and Forest Soils, Vol. 2000 of CSERGE Working Paper GEC, University of East Anglia, Norwich.
- Becker, M., Nieminen, T. M. and Geremia, F. (1994), 'Short-term variations and long-term changes in oak productivity in northeastern france. the role of climate and atmospheric  $co_2$ ', *Annals of Forest Science* **51**(5), 477 492.
- Beckett, K. P., Freer-Smith, P. H. and Taylor, G. (2000), 'Particulate pollution capture by urban trees: effect of species and windspeed', *Global Change Biology* **6**(8), 995–1003.
- Berger, S., Salke, G., Walther, G.-R. and Pott, R. (2007), 'Bioclimatic range shifts of cold-hardy evergreen species at their northern distributional limit', *Phytocoenologia* **37**(3-4), 523–539.
- Bergh, J., Linder, S., Lundmark, T. and Elfving, B. (1999), 'The effect of water and nutrient availability on the productivity of norway spruce in northern and southern sweden', *Forest Ecology and Management* **119**(1-3), 51–62.
- Betts, R. A. (2000), 'Offset of the potential carbon sink from boreal forestation by decreases in surface albedo', *Nature* **408**, 187 190.

- Bobinac, M., Batos, B., Miljkovic, D. and Radulovic, S. (2012), 'Polycyclism and phenological variability in the common oak (*Quercus robur* 1.)', *Arch. Biol. Sci., Belgrade* **64**, 97–105.
- Boisvenue, C. and Running, S. W. (2006), 'Impacts of climate change on natural forest productivity - evidence since the middle of the 20th century', *Global Change Biology* 12(5), 862– 882.
- Bolstad, P. V., Reich, P. and Lee, T. (2003), 'Rapid temperature acclimation of leaf respiration rates in *Quercusalba* and *Quercus rubra*', *Tree Physiology* **23**, 969–976.
- Both, H. and Bruggemann, W. (2009), 'Photosynthetic studies on european evergreen and deciduous oaks grown under central european climate conditions. i: a case study of leaf development and seasonal variation of photosynthetic capacity in *Quercus robur* (1.), *Q. ilex* (1.) and their semideciduous hybrid, *Q x turneri* (willd.)', *Trees* 23, 1081–1090.
- Brasier, C. (1999), Phytophthora pathogens of trees: Their rising profile in europe, Information note, Forestry Commission, 231 Corstorphine Road, Edinburgh.URL: www.forestry.gov.uk
- Breda, N., Cochard, H., Dreyer, E. and Granier, A. (1993), 'Field comparison of transpiration, stomatal conductance and vulnerability to cavitation of quercus petraea and quercus robur under water stress', *Annals Forestry Science* **50**(6), 571–582.
- Broadmeadow, M. (2004), The potential effects of climate change for trees and woodland in the south west: A report prepared for the south west conservancy of the forestry commission,funded by the sustainable forestry group of the forestry commission, Technical report, Forestry Commission.
- Broadmeadow, M. J. and Matthews, R. (2004), UK Emissions by Sources and Removals by Sinks due to Land Use, Land Use Change and Forestry Activities, Report, Centre for Hydrology and Hydrology, Queen's University Belfast, Aberdeen University, Rothamsted Research, Forest Research Alice Holt, Centre for Terrestrial Carbon Dynamics (University of Edinburgh, University College London, University of Sheffield)., chapter Section 8: Survey Methods for Kyoto Protocol Monitoring and Verification of UK Carbon Stocks, pp. 1–25.
- Broadmeadow, M. and Matthews, R. (2003), *Forests Carbon and Climate Change: the UK Contribution*, Information Note, Forestry Commission, Edinburgh.
- Broadmeadow, M., Ray, M. and Samuel, C. (2005), 'Climate change and the future for broadleaved tree species in britain', *Forestry* 78, 145–161.
- Brown, S. (2002), 'Changes in the use and management of forests for abating carbon emissions: issues and challenges under the kyoto protocol', *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences* **360**(1797), 1593–1605.

- Calatayud, V., Cervero, J., Calvo, E. and Garcia-Breijo, F.-J. (2011), 'Responses of evergreen and deciduous *Quercus* species to enhanced ozone levels', *Environmental Pollution* 159, 55– 63.
- Camarero, J. J., Olano, J. M., Alfaro, S. J. A., Fernandez-Marin, B., Becerril, J. M. and Garcia-Plazola, J. I. (2012), 'Photoprotection mechanisms in *Quercus ilex* under contrasting climatic conditions', *Flora* 207, 557–564.
- Canadell, J., Riba, M. and Andrés, P. (1988), 'Biomass equations for quercus ilex l. in the montseny massif, northeastern spain', *Forestry* **61**(2), 137–147.
- Cannell, M. G. R. (2003), 'Carbon sequestration and biomass energy offset: Theoretical, potential and achievable capacities globally, in europe and the uk', *Biomass and Bioenergy* 24, 97 – 116.
- Cannell, M. G. R. and Dewar, R. C. (1995), 'The carbon sink provided by plantation forests and their products in britain', *Forestry* **68**(1), 35–48.
- Cannell, M. G. R. and Thornley, J. H. M. (2000), 'Modelling the components of plant respiration: Some guiding principles', *Annals of Botany* 85(1), 45–54.
- Cater, M. and Batic, F. (2006), 'Groundwater and light conditions as factors in the survival of pendunculate oak (quercus robur l.) seedlings', *125* pp. 419–426.
- Catovsky, S. and Bazzaz, F. A. (2000), 'Contributions of coniferous and broad-leaved species to temperate forest carbon uptake: a bottom-up approach', *Canadian Journal of Forest Research* 30, 100–111.
- Catovsky, S., Holbrook, N. M. and Bazzaz, F. A. (2002), 'Coupling whole-tree transpiration and canopy photosynthesis in coniferous and broad-leaved tree species', *Canadian Journal* of Forest Research **32**, 295–309.
- Cavender-Bares, J. and Bazzaz, F. A. (2000), 'Changes in drought response strategies with ontogeny in quercus rubra: implications for scaling from seedlings to mature trees', *Oecologia* 124(1), 8–18.
- Cavender-Bares, J. and Bazzaz, F. A. (2004), From leaves to ecosystems: Using chlorophyll fluorescence to assess photosynthesis and plant function in ecological studies, *in* 'Chlorophyll a fluorescence a signature of photosynthesis', Vol. 19 (612), Kluwer Academic Publishers, Norwell, USA, chapter 29, pp. 737–755.
- Chappelka, A. H. and Freer-Smith, P. H. (1995), 'Predisposition of trees by air pollutants to low temperatures and moisture stress', *Environmental Pollution* **87**, 105–117.

- Chavana-Bryant, C. (2006), Long-term regeneration patterns and conservation status of a remnant old-growth oak forest: Wistman Wood, Dartmoor, Thesis, University of Plymouth, Plymouth.
- Chave, J., Andalo, C., Brown, S., Cairns, M., Chambers, J., Eamus, D., Folster, H., Fromard, F., Higuchi, N., Kira, T., Lescure, J.-P., Nelson, B., Ogawa, H., Puig, H., Riera, B. and Yamakura, T. (2005), 'Tree allometry and improved estimation of carbon stocks and balance in tropical forests', *Oecologia* 145, 87–99. URL: http://dx.doi.org/10.1007/s00442-005-0100-x
- Chaves, M. M., Pereira, J. S., Maroco, J., Rodrigues, M. L., Ricardo, C. P. P., Osorio, M. L., Carvalho, I., Faria, T. and Pinheiro, C. (2002), 'How plants cope with water stress in field. photosynthesis and growth', *Annals of Botany* 89, 907–916.
- Clark, A. J., Landolt, W., Bucher, J. B. and Strasser, R. J. (2000), 'Beech (fagus sylvatica) response to ozone exposure assessed with a chlorophyll a fluorescence performance index', *Environmental Pollution* **109**(3), 501–507. doi: 10.1016/S0269-7491(00)00053-1.
- Clark, D. A., Brown, S., Kicklighter, D. W., Chambers, J. Q., Thomlinson, J. R. and Ni, J. (2001), 'Measuring net primary production in forests: Concepts and field methods', *Ecological Applications* **11**(2), 356–370.
- Clark III, A. and Schroeder, J. G. (1986), 'Weight, volume, and physical properties of major hardwood species in the southern appalachian mountains', *Southeastern Forest Experiment Station, Research Paper, SE-253* United States Department of Agriculture Forest Service.
- Cornelissen, J. H. C., Lavorel, S. garnier, E., Diaz, S., bucahmann, N., Gurviich, D. E., Reich, P. B., ter Steege, H., Morgan, H. D., vsn der Heijden, M. G. A., Pausas, J. G. and Poorter, H. (2003), 'A handbook of protocols for easy measurement of plant functional traits worldwide', *Australian Journal of Botany* 51, 335–380.
- Crescente, M. F., Gratani, L. and Larcher, W. (2002), 'Shoot growth efficiency and production of quercus ilex l. in different climates', *Flora Morphology, Distribution, Functional Ecology of Plants* **197**(1), 2–9.
- Davis, M. and Shaw, R. (2001), 'Range shifts and adaptive responses to quaternary climate change', *Science* **292**, 673–9.
- De Nicola, F., Alfani, A. and D'Ambrosio, N. (2011), 'Impact of the mediterranean urban environment on photosynthetic efficiency of *Quercus ilex* leaves', *Water, Air, & Soil Pollution* 220(1), 151–160.
- Delfine, S., Di Marco, G. and Loreto, F. (1999), 'Estimation of photorespiratory carbon dioxide recycling during photosynthesis', *Functional Plant Biology* **26**(8), 733–736.

- DeLucia, E. H., Drake, J. E., Thomas, R. B. and Gonzalez-Meler, M. (2007), 'Forest carbon use efficiency: is respiration a constant fraction of gross primary production?', *Global Change Biology* 13, 1157–1167.
- Dickson, R. E. and Tomlinson, P. T. (1996), 'Oak growth, development and carbon metabolism in response to water stress', *Annales des Sciences Forestieres* **53**, 181–196.
- Digimap (2009), 'Location of study sites (plymouth & dartmoor)', University of Edinburgh.
- Dlugokencky, E. and Tans, P. (2013), 'Recent global monthly *co*₂ [webpage] url:www.esrl.noaa.gov/gmd/ccgg/trends/. accessed on [14/03/2013].', online.
- Drobyshev, I., Niklasson, M., Eggertsson, O., Linderson, H. and Sonesson, K. (2008), 'Influence of annual weather on growth of pedunculate oak in southern sweden', *Annals of Forest Science* **65**(5), 512.
- Dungan, R. J. and Whitehead, D. (2006), 'Modelling environmental limits to light use efficiency for a canopy of two broad-leaved tree species with contrasting leaf habit', *New Zealand Journal of Ecology* **30**(2), 251–260.
- Enquist, B. J., Allen, A. P., Brown, J. H., Gillooly, J. F., Kerkhoff, A. J., Niklas, K. J., Price, C. A. and West, G. B. (2007), 'Biological scaling: Does the exception prove the rule?', *Nature* 445(7127), E9–E10.
- Enquist, B. J., Brown, J. H. and West, G. B. (1998), 'Allometric scaling of plant energetics and population density', *NATURE* **395**(6698), 163–165.
- Enquist, B. J., Tiffney, B. H. and Niklas, K. J. (2007), 'Metabolic scaling and the evolutionary dynamics of plant size, form, and diversity: Toward a synthesis of ecology, evolution, and paleontology', *International Journal of Plant Sciences* 168(5), 729–749.
- Environmental Protection and Monitoring Unit (2009), 2009 detailed assessment of nitrogen dioxide for plymouth city council. in fulfilment of part iv of the environment act 1995, Report Plym/DA/2009, Plymouth City Council, Plymouth City Council Floor 6 The Civic Centre Armada Way Plymouth Devon PL1 2AA.
- Environmental Protection and Monitoring Unit (2012), 2012 air quality updating and screening assessment for plymouth city council. in fulfillment of part iv of the environment act 1995 local air quality management, Report PLYM/LAQM/USA/2012, Plymouth City Council, Civic Centre, Armada Way, Plymouth.
- Epron, D. and Dreyer, E. (1993*a*), 'Long-term effects of drought on photosynthesis of adult oak trees quercus petraea (matt.) liebl. and quercus robur l. in a natural stand', *New Phytologist* 125(2), 381–389. 10.1111/j.1469-8137.1993.tb03890.x.

- Epron, D. and Dreyer, E. (1993*b*), 'Photosynthesis of oak leaves under water stress: maintenance of high photochemical efficiency of photosystem ii and occurrence of non-uniform *co*₂ assimilation', *Tree Physiology* **13**(2), 107–117.
- Epron, D., Dreyer, E. and Breda, N. (1992), 'Photosynthesis of oak trees [*Quercus petrea* (matt.) liebl.] during drought under field conditions: diurnal course of net *co*₂ assimilation and photochemical efficiency of photosystem ii', *Plant, Cell and Environment* **15**, 809–820.
- Ericsson, O., Rytter, L. and Vapaavuori, E. (1996), 'Physiology of carbon allocation in trees', Biomass and Bioenergy 11(2Ű3), 115–127.
  URL: http://www.sciencedirect.com/science/article/pii/0961953496000323
- EUFORGEN (2009), 'Distribution map of pedunculate oak (Quercus robur)'.
- Evans, J. R. and Poorter, H. (2001), 'Photosynthetic acclimation of plants to growth irradiance: the relative importance of specific leaf area and nitrogen partitioning in maximising carbon gain', *Plant, Cell and Environment* **24**(8), 755–767.
- Ferretti, M., Innes, J. L., Jalkanen, R., Saurer, M., Schaffer, J., Spiecker, H. and von Wilpert, K. (2002), 'Air pollution and environmental chemistry L'ÛÒ what role for tree-ring studies?', *Dendrochronologia* 20(1-2), 159 – 174.
- Filella, I., Llusià, J., Piñol, J. and Peñuelas, J. (1998), 'Leaf gas exchange and fluorescence of *Phillyrea latifolia*, *Pistacia lentiscus* and *Quercus ilex* saplings in severe drought and high temperature conditions', *Environmental and Experimental Botany* **39**(3), 213–220.
- Filippo, B. (2008), 'Functional leaf traits, plant communities and acclimation processes in relation to oxidative stress in trees: a critical overview', *Global Change Biology* 14(11), 2727– 2739.
- Fitter, A. and Hay, R. (2002), *Environmental Physiology of Plants*, 3 edn, Academic Press, New York.
- Forestry Commission (2012*a*), 'Does flushing date vary with seed origin? forestry research phenology 2004 data [online] . accessed on 10th september 2012 [url] http://www.forestry.gov.uk/fr//infd-5zymmx'.
- Forestry Commission (2012*b*), 'National forest inventory information', *Web page* p. [online] http://www.forestry.gov.uk/inventory (accessed on 12 November 2012).
- Franco, M. (1985), A modular approach to tree production. pp. 257-272., *in* J. White, ed., 'Studies on Plant Demography: a Festschrift for John L. Harper', Academic Press, London.
- Franco, M. (2008), Carbon absorption and storage, Technical report, The Stationary Office Ltd, Crown Copyright, London.
- Franco, M. and Kelly, C. (1998), 'The interspecific mass-density relationship and plant geometry.', *Proceedings of the National Academy of Sciences of the USA* **95**, 7830–7835.
- Friedrichs, D. A., Buntgen, U., Frank, D. C., Esper, J., Neuwirth, B. and Loffler, J. (2009), 'Complex climate controls on 20th century oak growth in central-west germany', *Tree Physiol* 29(1), 39–51.
- Futai, K., Takeshi, T. and Ryota, K. (2008), *Mycorrhizae: Sustainable Agriculture and Forestry*, Springer Science, chapter Ectomycorrhizae and their importance in forest ecosystems., pp. 241–285.
- Garcia-Plazaola, J., Faria, T., Abadia, J.and Abadia, A., Chaves, M. and Pereira, J. S. (1997), 'Seasonal changes in xanthophyll composition and photosynthesis of cork oak (*Quercus suber* 1.) leaves under mediterranean climate', *Journal of Experimental Botany* 48, 1667– 1674.
- GarcIa-Plazaola, J. I., Artetxe, U. and Becerril, J. M. (1999), 'Diurnal changes in antioxidant and carotenoid composition in the mediterranean schlerophyll tree *Quercus ilex* (1) during winter', *Plant Science* **143**(2), 125–133.
- Gieger, T. and Thomas, F. M. (2002), 'Effects of defoliation and drought stress on biomass partitioning and water relations of *Quercus robur* and *Quercus petrea*', *Basic and Applied Ecology* **3**, 171–181.
- Gimeno, T. E., Pias, B., Lemos-Filho, J. P. and Valladares, F. (2009), 'Plasticity and stress tolerance override local adaptation in the responses of mediterranean holm oak seedlings to drought and cold', *Tree Physiol* **29**(1), 87–98.
- Givnish, T. J. (1988), 'Adation to sun and shade: A whole plant perspective', *Australian Journal* of *Plant Physiology* **15**, 63–92.
- Givnish, T. J. (2002), 'Adaptive significance of evergreen vs. deciduous leaves: Solving the triple paradox', *Silva Fennica* **36**(3), 703–743.
- Gomez-Aparicio, L., Parez-Ramos, I. M., Mendoza, I., Matías, L., Quero, J. L., Castro, J., Zamora, R. and Maron, T. (2008), 'Oak seedling survival and growth along resource gradients in mediterranean forests: implications for regeneration in current and future environmental scenarios', *Oikos* 117(11), 1683–1699.
- Gomez-Aparicio, L., valladares, F. and Zamora, R. (2006), 'Differential light responses of mediterranean tree saplings: linking ecophysiology with regeneration nihe u four co-occuring species', *Tree Physiology* 26, 947–958.

- Gough, C. M., Vogel, C. S., Schmid, H. P. and Curtis, P. S. (2008), 'Controls on annual forest carbon storage: Lessons from the past and predictions for the future', *BioScience* 58(7), 609– 622.
- Gracia, C. and Sabate, S. (2002), 'Aboveground biomass expansion factors and biomass equations of forests in catalonia', *CREAF expert meeting on biomass expansion factors (BEF); Valencia* p. 5.
- Grant, R. F., Margolis, H. A., Barr, A. G., Black, T. A., Dunn, A. L., Bernier, P. Y. and Bergeron,
  O. (2008), 'Changes in net ecosystem productivity of boreal black spruce stands in response to changes in temperature at diurnal and seasonal time scales', *Tree Physiol* 29(1), 1–17.
- Gratani, L. and Bonmbelli, A. (2001), 'Differences in leaf traits among mediterranean broadleaved evergreen shrubs', *Annals of Botany Fennici* **38**, 15–24.
- Gratani, L., Crescente, M. F. and Petruzzi, M. (2000), 'Relationship between leaf life-span and photosynthetic activity of quercus ilex in polluted urban areas (rome)', *Environmental Pollution* **110**, 19–28.
- Gratani, L., Pesoli, P., Crescente, M. F., Aichner, K. and Larcher, W. (2000), 'Photosynthesis as a temperature indicator in *Quercus ilex* 1', *Global and Planetary Change* **24**(2), 153–163.
- Gratani, L. and Varone, L. (2006), 'Carbon sequestration by quercus ilex l. and quercus pubescens willd. and their contribution to decreasing air temperature in rome', *Urban Ecosystems* **9**, 27–37.
- Gratani, L., Varone, L. and Catoni, R. (2008), 'Relationship between net photosynthesis and leaf respiration in mediterranean evergreen species', *Photosynthetica* **46**(4), 567–573.
- Gratini, L. and Varone, L. (2006), 'Carbon sequestration by quercus ilex l. and quercus pubescens willd. and their contribution to decreasing air temperature in rome', *Urban Ecosystems* **9**, 27–37.
- Grime, J. P. (2002), *Plant Strategies, Vegetation Processes, and Ecosystem Properties*, Wiley, Chichester.
- Hajji, M., Dreyer, E. and Marcais, B. (2009), 'Impact of erysiphe alphitoides on transpiration and photosynthesis in quercus robur leaves', *European Journal of Plant Pathology* **125**, 63– 72.

URL: http://dx.doi.org/10.1007/s10658-009-9458-7

Hamerlynck, E. and Knapp, A. (1996), 'Photosynthetic and stomatal responses to high temperature and light in two oaks at the western limit of their range', *Tree Physiology* **16**, 557–565.

- Hansatech Instruments Ltd (2007), 'Pocket pea; portable continuous excitation chlorophyll fluorescence meter'.
- Hansatech Instruments Ltd (2010), 'Chorophyll flourescence measured parameters'.

Harper, J. L. (1989), 'The value of a leaf', Oecologia 80(1), 53-58. 10.1007/BF00789931.

- Hegerl, G. C., Zwiers, F. W., Braconnot, P., Gillett, N. P., Luo, Y., Marengo, J., Nicholls, N., Penner, J. E. and Stott, P. A. (2007), 'Understanding and attributing climate change', *Climate Change 2007: The Physical Science Basis. Working Group 1 Contribution to the Fourth Assessment Report of the Intergovernmental Panell on Climate Change (IPCC)* Chapter 9, 663 746.
- Heimann, M. and Reichstein, M. (2008), 'Terrestrial ecosystem carbon dynamics and climate feedbacks', *Nature* 451(7176), 289–292.
- Hermans, C., Smeyers, M., Rodriguez, R., Eyletters, M., Strasser, R. and Delhaye, J.-P. (2003), 'Quality assessment of urban trees: A comparative study of physiological characterisation, airborne imaging and on site fluorescence monitoring by the ojip-test', *Journal of Plant Physiology* **160**(1), 81–90.
- Hodges, J. D. (1967), 'Patterns of photosynthesis under natural environmental conditions', *Ecology* 48(2), 234–242.
- Hoffman, W. A. and Poorter, H. (2002), 'Avoiding bias in calculations of relative growth rate', *Annals of Botany* **80**, 37–42.
- Hollinger, D. (1992), 'Leaf and simulated whole-canopy photosynthesis in two co-occurring tree species', *Ecology* 73, 1Ű14.
- Hunt, R. (1982), *Plant growth Curves: The Functional Approach to Plant Growth Anaylsis*, 1 edn, Edward Arnold, London.
- IPCC (2000), International panel on climate change, special report on land use, land-use change and forestry, Technical report, Cambridge University Press, Cambridge, UK.
- IPCC (2003), 'Good practice guidance for land use, land-use change and forestry', Institute forGlobal Environmnetal Strategies (IGES).
- IPCC (2007), Climate change 2007: Synthesis report. contribution of working groups i, ii and iii to the fourth assessment report of the intergovernmental panel on climate change [core writing team, pachauri, r.k and reisinger, a. (eds.)]. ipcc, geneva, switzerland, 104 pp., Technical report, IPCC, Geneva, Switzerland.

- Jarvis, P. and Linder, S. (2007), *Forestry and Climate Change*, Cab International, chapter 9: Forests Remove Carbon Dioxide from the Atmosphere: Spruce Forest Tales, pp. 60–72.
- Jebson, S. (2007), National meteorological library and archive: Fact sheet no. 4 climate of the united kingdom, Met Office, Crown Copyright.
- Jump, A. S., Hunt, J. M. and Peñuelas, J. (2006), 'Rapid climate change-related growth decline at the southern range edge of *Fagus sylvatica*', *Global Change Biology* **12**(11), 1365–2486.
- Kazda, M., Salzer, J. and Reiter, I. (2000), 'Photosynthetic capacity in relation to nitrogen in the canopy of a quercus robur, fraxinus angustifolia and tilia cordata flood plain forest', *Tree Physiology* **20**(15), 1029–1037.
- Kennedy, C. E. J. and Southward, T. R. E. (1984), 'The number of species of insects associated with british trees: A re-analysis', *Journal of Animal Ecology* 53, 455–478.
- Kennedy, F. M. (2003), Forest Research Annual Report and Accounts 2001-2002, Stationery Office, Edinburgh, chapter How extensive are the impacts of nitrogen pollution in Great Britain'sforests?, pp. 66–75.

**URL:** http://www.forestresearch.gov.uk/pdf/FR0102nitro.pdf/ FILE/FR0102nitro.pdf [Accessed January 2010].

- Kennedy, M., Anderson, C., O'Hagan, A., Lomas, M., Woodward, I., Gosling, J. P. and Heinemeyer, A. (2008), 'Quantifying uncertainty in the biospheric carbon flux for england and wales', *Journal of the Royal Statistical Society: Series A (Statistics in Society)* 171, 109–135.
- Kern, Z. and Popa, I. (2007), 'Climate-growth relationship of tree species from mixed stand of apuseni mts., romania', *Dendrochronolgia* 24, 109–115.
- Ketterings, Q. M., Coe, R., va Noordwijk, M., Ambagu, Y. and Palm, C. A. (2001), 'Reducing uncertainty in the use of allometric biomass equations for predicting avbove-ground tree biomass in mixed secondary forests', *Forest Ecology and Management* pp. 199–209.
- Kleinschmidt, J. (1993), 'Intraspecific variation of growth and adaptive traits in european oak species', *Annals of Forest Science* **50**(1), 1665–1855.
- Koike, T., Kitaoka, S., Ichie, T., Lei, T. T. and Kitao, M. (2004), *Global Environmental Change* in the Ocean and on Land, TERRAPUB, Tokyo, chapter Photosynthetic Characteristics of mixed Deciduous-Broadleaf Forests form Leaf to Stand, pp. 453–472.
- Korner, C. (2003), 'Carbon limitation in trees', *Journal of Ecology* **91**(1), 4–17. 10.1046/j.1365-2745.2003.00742.x.
- Kottek, M., Grieser, M. J., Beck, C., Rudolf, B. and Rubel, F. (2006), 'World map of the köppen-geiger climate classification updated', *Meteorologische Zeitschrift* 15(3), 259–263.

- Kozlowski, T. T. (1971), 'Growth and development of trees', *Seed Germination, Ontogeny, and Shoot Growth* 1(Physiological Ecology: A Series of Monographs, Texts and Treatises).
- Kuster, T. M., Arend, M., Beleuler, P., Gunthadrt-Goerg, M. and Schulin, R. (2012), 'Water regime and growth of young oak stands subjected to air-warming and drought on two different forest soils in a model ecosystem experiment', *Plant Biology* Early View, 1–10. Article first published online: 30 JAN 2012. DOI: 10.1111/j.1438-8677.2011.00552.x.
- Lambers, H., III, F. S. C. and Pons, T. L. (2008), *Plant Physiological Ecology*, 2 edn, Springer, New York.
- Larcher, W. (1969), 'The effect of environmental and physiological variables on the carbon dioxide gas exchange of trees.', *Photosynthetica* **3**, 167–198.
- Larcher, W. (2000), 'Temperature stress and survival ability of mediterranean sclerophyllous plants', *Plant Biosystems* 134, 279–295.
- Laureano, R. G., Lazo, Y. O., Linares, J. C., Luque, A., Martinez, F., Seco, J. I. and Merino, J. (2008), 'The cost of stress resistance: construction and maintenance costs of leaves and roots in two populations of quercus ilex', *Tree physiology* 28(11), 1721–1728.
- Lichtenthaler, H. K., Ac, A., Marek, M. V., Kalina, J. and Urban, O. (2007), 'Differences in pigment composition, photosynthetic rates and chlorophyll fluorescence images of sun and shade leaves of four tree species', *Plant Physiology and Biochemistry* 45(8), 577–588.
- Linderson, M. L., Mikkelsen, T. N., Ibrom, A., Lindroth, A., Ro-Poulsen, H. and Pilegaard, K. (2012), 'Up-scaling of water use efficiency from leaf to canopy as based on leaf gas exchange relationships and the modeled in-canopy light distribution', *Agricultural and Forest Meteorology* 152(0), 201–211.
- Lindroth, A., Lagergren, F., Aurela, M., Bjarnadottir, B., Christensen, T., Dellwik, E., Grelle, A., Ibrom, A., Johansson, T., Lankreijer, H., Launiainen, S., Laurila, T., der, M. M., Nikinmaa, E., Pilegaard, K. I. M., Sigurdsson, B. D. and Vesala, T. (2008), 'Leaf area index is the principal scaling parameter for both gross photosynthesis and ecosystem respiration of northern deciduous and coniferous forests', *Tellus B* **60**(2), 129–142. 10.1111/j.1600-0889.2007.00330.x.
- Litton, C. M., Raich, J. W. and Ryan, M. G. (2007), 'Carbon allocation in forest ecosystems', *Global Change Biology* **13**(10), 2089–2109. 10.1111/j.1365-2486.2007.01420.x.
- Lorenz, K. and Lal, R. (2010), Carbon Sequestration in forest Ecosystems, Springer.
- Lunt, P. H. and Hedger, J. N. (2003), 'Effects of organic enrichment of mine spoil on growth and nutrient uptake in oak seedlings inoculated with selected ectomycorrhizal fungi', *Restoration Ecology* **11**(2), 125–130.

- Luyssaert, S., Inglima, I., Jung, M., Richardson, A. D., Reichsteins, M., Papale, D., Piao, S. L., Schulzes, E. D., Wingate, L., Matteucci, G., Aragao, L., Aubinet, M., Beers, C., Bernhoffer, C., Black, K. G., Bonal, D., Bonnefond, J. M., Chambers, J., Ciais, P., Cook, B., Davis, K. J., Dolman, A. J., Gielen, B., Goulden, M., Grace, J., Granier, A., Grelle, A., Griffis, T., Grunwald, T., Guidolotti, G., Hanson, P. J., Harding, R., Hollinger, D. Y., Hutyra, L. R., Kolar, P., Kruijt, B., Kutsch, W., Lagergren, F., Laurila, T., Law, B. E., Le Maire, G., Lindroth, A., Loustau, D., Malhi, Y., Mateus, J., Migliavacca, M., Misson, L., Montagnani, L., Moncrieff, J., Moors, E., Munger, J. W., Nikinmaa, E., Ollinger, S. V., Pita, G., Rebmann, C., Roupsard, O., Saigusa, N., Sanz, M. J., Seufert, G., Sierra, C., Smith, M. L., Tang, J., Valentini, R., Vesala, T. and Janssens, I. A. (2007), 'Co 2 balance of boreal, temperate, and tropical forests derived from a global database', *Global Change Biology* 13(12), 2509–2537.
- M., W. J., Reich, P. B., Oleksyn, J. and Eissenstat, D. M. (2006), 'Comparisons of stucture and life span in roots and leaves among temperate trees', *Ecological Monographs* 76(3), 381–397.
- Mahall, B. E., Tyler, C. M., Cole, E. S. and Mata, C. (2009), 'A comparative study of oak (quercus, fagaceae) seedling physiology during summer drought in southern california.', *American journal of botany* 96(4), 751–761.
- Manes, F., Seufert, G. and Vitale, M. (1997), 'Ecophysiological studies of mediterranean plant species at the castelporziano estate', *Atmospheric Environment* **31**, **Supplement 1**(0), 51–60.
- Martinez-Ferri, E., Manrique, E., Valladares, F. and Balaguer, L. (2004), 'Winter photoinhibition in the field involves different processes in four co-occurring mediterranean tree species', *Tree Physiol* 24(9), 981–990.
- Masarovicova, E. (1991), 'Leaf shape, stomatal density and photosynthetic rate of the common oak leaves', *Biologia Plantarum* **33**(6), 495–500.
- Mascaro, J., Litton, C. M., Huges, R. F., Uowolo, A. and Schnitzer, S. A. (2011), 'Minimizing bias in biomass allometry: Model selection and log-transformation of data', *Biotropica* 43(6), 649–653.
- Matthews, R. W. and Mackie, D. (2006), *Forest mensuration: A handbook for Practitioners*, Forestry Commission, Edinburgh.
- McCarthy, H. R., Oren, R., Finzi, A. C. and Johnsen, K. H. (2006), 'Canopy leaf area constrains [co2]-induced enhancement of productivity and partitioning among aboveground carbon pools', *PNAS* 103, 19356–19361.
- Mediavilla, S. and Escudero, A. (2003), 'Mature trees versus seedlings: Differences in leaf traits and gas exchange patterns in three co-occurring mediterranean oaks', *Annals of Forest Science* 60(5), 455–460.

- Meir, P., Kruut, B., Broadmeadow, M., Barbosa, E., Kull, O., Carswell, F., Nobre, A. and Jarvis, P. G. (2002), 'Acclimation of photosynthetic capacity to irradiance in tree canopies in relation to leaf nitrogen concentration and leaf mass per unit area', *Plant, Cell and Environment* 25, 343–357.
- Metnet (2009), 'Rain data from university of plymouth 'metnet'. university of plymouth, drakes circus, pl4 8aa', online archives.
- Miller, P. C. and Stoner, W. A. (1979), Canopy structure and environmental interactions, *in* O. T. Solbrig, S. Jain, G. B. Johnson and P. H. Raven, eds, 'Topics in Plant Population Biology', Columbia University Press, New York, pp. 428–458.
- Minitab ® (2010), 'Minitab 16 ® statistical software'.URL: [Computer software]. State College, PA: Minitab, Inc. (www.minitab.com)
- Molchanov, A. G. (2005), 'Photosynthetic productivity of oak stand under variable environmental conditions and water availability', *Russian Journal of Plant Physiology* **52**, 463–472.
- Morecroft, M. D., Bealey, C. E., Beaumont, D. A., Benham, S., Brooks, D. R., Burt, T. P., Critchley, C. N. R., Dick, J., Littlewood, N. A., Monteith, D. T., Scott, W. A., Smith, R. I., Walmsley, C. and Watson, H. (2009), 'The uk environmental change network: Emerging trends in the composition of plant and animal communities and the physical environment', *Biological Conservation* 142(12), 2814–2832.
- Morecroft, M., Stokes, V. and Morison, J. I. L. (2003), 'Seasonal changes in the photosynthetic capacity of canopy oak (quercus robur) leaves: the impact of slow development on annual carbon uptake', *International Journal of Bio-meteorology* **47**, 221–226.
- Morin, X., Ameglio, T., Ahas, R., Kurz-Besson, C., Lanta, V., Lebourgeois, F., Miglietta, F. and Chuine, I. (2007), 'Variation in cold hardiness and carbohydrate concentration from dormancy induction to bud burst among provenances of three european oak species', *Tree Physiol* 27(6), 817–825.
- Morin, X., Roy, J., Sonie, L. and Chuine, I. (2010), 'Changes in leaf phenology of three european oak species in response to experimental climate change', *New Phytologist* 186, 900–910.
- Muller-Landau, H. C. (2004), 'Interspecific and inter-site variation in wood specific gravity of tropical trees', *Biotropica* **36**(1), 20–32.
- Muller-Landau, H. C., Condit, R. S., Chave, J., Thomas, S. C., Bohlman, S. A., Bunyavejchewin, S., Davies, S., Foster, R., Gunatilleke, S., Gunatilleke, N., Harms, K. E., Hart, T., Hubbell, S. P., Itoh, A., Kassim, A. R., LaFrankie, J. V., Lee, H. S., Losos, E., Makana, J.-R.,

Ohkubo, T., Sukumar, R., Sun, I.-F., Nur Supardi, M. N., Tan, S., Thompson, J., Valencia, R., Muñoz, G. V., Wills, C., Yamakura, T., Chuyong, G., Dattaraja, H. S., Esufali, S., Hall, P., Hernandez, C., Kenfack, D., Kiratiprayoon, S., Suresh, H. S., Thomas, D., Vallejo, M. I. and Ashton, P. (2006), 'Testing metabolic ecology theory for allometric scaling of tree size, growth and mortality in tropical forests', *Ecology Letters* **9**(5), 575–588.

- Muukkonen, P. (2007), 'Generalized allometric volume and biomass equations for some tree species in europe', *EUROPEAN JOURNAL OF FOREST RESEARCH* **126**(2), 157–166.
- Nabuurs, G.-J., Van de Werf, D. C., Heidema, A. H. and van den Wyngaert, I. J. J. (2007), *Forestry and Climate Change*, Cabi International, chapter Chapter 13: Towards a High Resolution Foest Carbon Balance for Europe Based on Inventory Data, pp. 105–111.
- Nardini, A., Salleo, S., Lo Gullo, M. A. and Pitt, F. (2000), 'Different responses to drought and freeze stress of quercus ilex l. growing along a latitudinal gradient', *Vegetatio* 148(2), 139– 147.
- National Soil Resources Institute (2009), 'Soilscape soil descriptions for study sites.', online. **URL:** *www.landis.org.uk/soilscapes/*
- Niinemets, U. (2010), 'Responses of forest trees to single and multiple environmental stresses from seedlings to mature plants: Past stress history, stress interactions, tolerance and acclimation', *Forest Ecology and Management* 260(10), 1623–1639.
- Niinemets, U. and Tenhunen, J. D. (1997), 'A model separating leaf structural and physiological effects on carbon gain along light gradients for the shade-tolerant species *Acer saccharum*', *Plant Cell and Environment* **20**, 845–866.
- Nobel, P. S. (2005), *Physicochemical and Environmental Plant Physiology*, Academic Press, London.
- Nowak, D. J. and Crane, D. E. (2002), 'Carbon storage and sequestration by urban trees in the usa', *Environmental Pollution* **116**, 381–389.
- Ogaya, R. and Penuelas, J. (2003), 'Comparative seasonal gas exchange and chlorophyll fluorescence of two dominant woody species in a holm oak forest', *Flora - Morphology, Distribution, Functional Ecology of Plants* **198**(2), 132–141.
- Ogaya, R. and Penuelas, J. (2006), 'Contrasting foliar responses to drought in *Quercus ilex* and *Phillyrea latifolia*', *Biologia Plantarum* **50**, 3.
- Ogaya, R. and Penuelas, J. (2007*a*), 'Leaf mass per area ratio in quercus ilex leaves under a wide range of climatic conditions. the importance of low temperatures', *Acta Oecologica* **31**(2), 168–173.

- Ogaya, R. and Penuelas, J. (2007*b*), 'Tree growth, mortality, and above-ground biomass accumulation in a holm oak forest under a five-year experimental field drought', *Plant Ecology* **189**(2), 291–299.
- Ohtsuka, T., Saigusa, N. and Koizumi, H. (2009), 'On linking multiyear biometric measurements of tree growth with eddy covariance-based net ecosystem production', *Global Change Biology* 15(4), 1015–1024.
- Ordpnez, J. C., van Bodegom, P. M., Witte, J.-P. M., Wright, I. J., Reich, P. B. and Aerts, R. (2009), 'A global study of relationships between leaf traits, climate and soil measures of nutrient fertility', *Global Ecology and Biogeography* 18, 137–149.
- Osborne, C. P. and Beerling, D. J. (2003), 'The penalty of a long hot summer. photosynthetic acclimation to high co2 and continuous light in "living fossil" conifers', *Plant Physiology* **133**, 803–812.
- Pallardy, S. G. (2008), Physiology of Woody Plants, 3 edn, Elsevier.
- Parelle, J., Brendel, O., Jolivet, Y. and Dreyer, E. (2007), 'Intra- and interspecific diversity in the response to waterlogging of two co-occurring white oak species (quercus robur and q. petraea)', *Tree Physiol* 27(7), 1027–1034.
- Peñuelas, J., Filella, I., Llusià, J., Siscart, D. and Pinol, J. (1998), 'Comparative field study of spring and summer leaf gas exchange and photobiology of the mediterranean trees *Quercus ilex* and *Phyllyrea latifolia*', *J. Exp. Bot.* **49**(319), 229–238.
- Penuelas, J. and Ogaya, R. (2003), 'Comparative field study of *Quercus ilex* and *Phillyrea latifolia*: photosynthetic response to experimental drought conditions', *Environmental and Experimental Botany* 50(2), 137–148.
- Peterken, G. F. (2001), 'Ecological effects of introduced tree species in britain', *Forest Ecology* and Management **141**(1-2), 31–42.
- Phillips, I. D. and Denning, H. (2007), 'Winter daily precipitation variability over the south west peninsula of england', *Theoretical and Applied Climatology* **87**, 103–122.
- Phillips, I. D. and Mcgregor, G. R. (2002), 'The relationship between monthly and seasonal south-west england rainfall anomalies and concurrent north atlantic sea surface temperatures', *International Journal of Climatology* 22, 197–217.
- Plantlife (2010), 'Plantlife: Invasive non-native plants and the law', Online. URL: http://www.plantlife.org.uk/uploads/documents/Invasivesandthelaw.pdf

- Ponton, S., Dupouey, J.-L., Breda, N. and Dreyer, E. (2002), 'Comparison of water-use efficiency of seedllings from two sympatric oak species: genotype x environment interactions', *Tree Physiology* 22, 413–422.
- Poorter, H. and Evans, J. (1998), 'Photosynthetic nitrogen-use efficiency of species that differ inherently in specific leaf area', *Oecologia* 116, 26–37.
- Poorter, H., Niklas, K. J., Reich, P. B., Oleksyn, J., Poot, P. and Mommer, L. (2012), 'Biomass allocation to leaves, stems and roots: meta-analyses of interspecific variation and environmental control', *New Phytologist* 193(1), 30–50.
- R (2009), 'R version 2.10.0 (2009-10-26) ; r foundation for statistical computing'. URL: *http://www.r-project.org*
- Rackham, O. (2008), 'Ancient woodlands: modern threats', New Phytologist 180(3), 571-586.
- Ravindranath, N. and Ostwald, M. (2008), Carbon Inventory Methods Handbook for Greenhouse Gas Inventory, Carbon Mitigation and Roundwood Production Projects, Vol. 29 of Advances in Global Change Research, Springer, Netherlands.
- Read, D., Freer-Smith, P., Morison, J., Hanley, N., West, C. and Snowdon, P. (2009), Combating climate change – a role for uk forests. an assessment of the potential of the uk's trees and woodlands to mitigate and adapt to climate change. the synthesis report, *in* N. A. of UK Forestry and C. C. S. G. 2009, eds, 'The synthesis report', The Stationery Office, Edinburgh.
- Reich, P. B., Ellsworth, D. S., Walters, M. B., Vose, J. M., Gresham, C., Volin, J. C. and Bowman, W. D. (1999), 'Generality of leaf trait relationships: A test across six biomes', *Ecology* 80(6), 1955–1969.
- Reille, M. and Pones, A. (1992), 'The ecological significance of schlerophyllous oak forests in the western part of the mediterranean basin: a note on pollen analytical data', *Vegetatio* 99-100, 13–17.
- Rohner, B., Bigler, C., Wunder, J., Brang, P. and Bugmann, H. (2012), 'Fifty years of natural succession in swiss forest reserves: changes in stand structure and mortality rates of oak and beech', *Journal of Vegetation Science* 23(5), 892–905.
- Rossini, M., Panigada, C., Meroni, M. and Colombo, R. (2006), 'Assessment of oak forest condition based on leaf biochemical variables and chlorophyll fluorescence', *Tree Physiol* 26(11), 1487–1496.
- Royer, D. L., Osborne, C. P. and Beerling, D. J. (2003), 'Carbon loss by deciduous trees in a co2-rich ancient polar environment', *Nature* **424**, 60–62.

- Royer, D. L., Osborne, C. P. and Beerling, D. J. (2005), 'Contrasting seasonal patterns of carbon gain in evergreen and deciduous trees of ancient polar forests', *Paleobiology* **31**(1), 141–150.
- Ruiz-Roberto, J. and Villar, R. (2005), 'Relative growth rate and biomass allocation in ten woody species with different leaf longevity using phylogenetic independent contrasts (pics)', *Plant Biology* 7, 484–494.
- Sabaté, S., Gracia, C. A. and Sánchez, A. (2002), 'Likely effects of climate change on growth of quercus ilex, pinus halepensis, pinus pinaster, pinus sylvestris and fagus sylvatica forests in the mediterranean region', *Forest Ecology and Management* **162**(1), 23–37.
- Sade, N., Gebremedhin, A. and Moshelion, M. (2012), 'Risk-taking plants: anisohydric behaviour as a strss-resistance trait.', *Plant Signal Behaviour* **7**, 767–770.
- Sanchez-Gomez, D., Valledares, F. and Zavala, M. A. (2006), 'Performance of seedlings of mediterranean woody species under experimental gradients of irradiance and water availability: trade-offs and evidence for niche differentiation', *New Phytologist* **170**, 795–806.
- Scott, P. (2008), Physiology and Behaviour of Plants, John Wiley & Sons Ltd, Chichester.
- Searle, S. Y., Turnbull, M. H., Boelman, N. T., Schuster, W. S., Yakir, D. and Griffin, K. L. (2012), 'Urban environment of new york city promotes growth in northern red oak seedlings', *Tree Physiology*.
- Simmons, I. (1965), 'Dartmoor oak copses: observations and speculations', *Field Studies* **2**, 225–235.
- Solomon, S., Plattner, G.-K., Knutti, R. and Friedlingstein, P. (2009), 'Irreversible climate change due to carbon dioxide emissions', *PNAS* **106**(6), 1704–1709.
- Solomon, S., Qin, D., Manning, M., Alley, R., Berntsen, T., Bindoff, N., Chen, Z., Chidthaisong, A., Gregory, J., Hegerl, G., Heimann, M., Hewitson, B., Hoskins, B., Joos, F., Jouzel, J., Kattsov, V., Lohmann, U., Matsuno, T., Molina, M., Nicholls, N., Overpeck, J., Raga, G., Ramaswamy, V., Ren, J., Rusticucci, M. and. Somerville, R., Stocker, T., Whetton, P., Wood, R. and Wratt, D. (2007), Technical summary. in: Climate change 2007: The physical science basis. contribution of working group i to the fourth assessment report of the intergovernmental panel on climate change [solomon, s., d. qin, m. manning, z. chen, m. marquis, k.b. averyt, m. tignor and h.l. miller (eds.)]. cambridge university press, cambridge, united kingdom and new york, ny, usa., Technical report, IPCC.
- Southward, T. R. E., Wint, G. W., Kennedy, C. E. J. and Greenwood, S. R. (2004), 'Seasonality, abundance, species richness and specificity of the phytophagous guild of insects on oak (*Quercus*) canopies', *European Journal of entomology* 101, 43–50.

Sparks, J. P. (2009), '2009', Oecologia 159, 1–12.

- Stace, C. (1997), *New Flora of the British Isles (Second Edition)*, Cambridge University Press, Cambridge.
- Stokes, V., Morecroft, M. and Morison, J. (2010), 'Comparison of leaf water use efficiency of oak and sycamore in the canopy over two growing seasons', *Trees - Structure and Function* 24(2), 297–306.
- Strasser, R. J., Srivastava, A. and Tsimilli-Michael, M. (2000), The fluorescence transient as a tool to characterise and screen photosynthetic samples, *in* M. Yunus, U. Pathre and P. Mohanty, eds, 'Probing Photosynthesis: Mechanisms, Regulation and Adaptation', Taylor and Francis, London, chapter 25.
- Struve, D., Ferini, F., Fini, A. and Pennati, L. (2009), 'Relative growth and water use of seedlings from three italian *Quercus* species', *Arboriculture and Urban Forestry* 35(3), 113– 121.
- Takashima, T., Hikosaka, K. and Hirose, T. (2004), 'Photosynthesis or persistence: nitrogen allocation in leaves of evergreen and deciduous quercus species', *Plant, Cell and Environment* 27(8), 1047–1054.
- Tan, K. H. (2005), *Soil Sampling, Preparation, and Analysis*, Books in Soils, Plants, and the Environment, CRC Press, New York.
- Tognetti, R., Cherubini, P., Marchi, S. and Raschi, A. (2007), 'Leaf traits and tree rings suggest different water-use and carbon assimilation strategies by two co-occurring quercus species in a mediterranean mixed-forest stand in tuscany, italy', *Tree Physiology* **27**(12), 1741–1751.
- Ugolini, F., Bussotti, F., Lanini, G. M., Raschi, A., Tani, C. and Tognetti, R. (2012), 'Leaf gas exchange and photosystem efficiency of the holm oak in urban green areas of florence, italy', *Urban Forestry and Urban Greening* **11**, 313–319.
- UKCP (2009), 'Uk climate projections [online] url: http://ukclimateprojections.defra.gov.uk/23122
  [accessed on] 15/03/2013', Online.
  URL: http://ukclimateprojections.defra.gov.uk/23122
- United Nations (1998), 'Kyoto protocol to the united nations framework convention on climate change.', [online]. URL: http://unfcc.int/resource/docs/convkp/kpeng.pdf
- United Nations (2006), Manual on methods and criteria for harmonized sampling, assessment, monitoring and analysis of the effects of air pollution on forests part iiia sampling and analysis of soil, Technical report, International Co-operative Programme on Assessment and Monitoring of Air Pollution Effects on Forests.

- Valentini, R., Matteucci, G., Dolman, A. J., Schulze, E. D., Rebmann, C., Moors, E. J., Granier, A., Gross, P., Jensen, N. O., Pilegaard, K., Lindroth, A., Grelle, A., Bernhofer, C., Grunwald, T., Aubinet, M., Ceulemans, R., Kowalski, A. S., Vesala, T., Rannik, U., Berbigier, P., Loustau, D., Gumundsson, J., Thorgeirsson, H., Ibrom, A., Morgenstern, K., Clement, R., Moncrieff, J., Montagnani, L., Minerbi, S. and Jarvis, P. G. (2000), 'Respiration as the main determinant of carbon balance in european forests', *NATURE* 404(6780), 861–865.
- Valladares, F., Arrieta, S., Aranda, I., Lorenzo, D., Sanchez-Gomez, D., Tena, D., Suarez, F. and Pardos, J. A. (2005), 'Shade tolerance, photoinhibition sensitivity and phenotypic plasticity of ilex aquifolium in continental mediterranean sites', *Tree Physiology* 25, 1041–1052.
- Valledares, F., Martinez-Ferri, E., Balaguer, L. and Perez-corona, E. and Manrique, E. (2000), 'Low leaf-level response to light and nutrients in mediterranean evergreen oaks: a conservative resource-use strategy?', *New Phytologist* 148, 79–91.
- van Minnen, J., Strengers, B., Eickhout, B. and Klein Goldewijk, K. (2006), 'Simulating carbon exchange between the terrestrial biopshere and atmosphere', *Integrated modelling of global environmental change. An overview of IMAGE 24* pp. 113 130.
- van Minnen, J., Strengers, B., Eickhout, B., Swart, R. and Leemans, R. (2008), 'Quantifying the effectiveness of climate change mitigation through forest plantations and carbon seques-tration with an integrated land-use model', *Carbon Balance and Management* **3**(1), 3.
- VanOijen, M., Schapendonk, A. and Hoglind, M. (2010), 'On the relative magnitudes of photosynthesis, respiration, growth and carbon storage in vegetation', *Annals of Botany* 105, 793– 797.
- Vaz, M., Pereira, J., Gazarini, L., David, T., David, J., Rodrigues, A., Maroco, J. and Chaves, M. (2010), 'Drought induced photosynthetic inhibition and autumn recovery in two mediterranean oak species (*Quercus ilex* and *Quercus suber*)', *Tree Physiology* **30**(8), 946–956.
- Villar, R., Held, A. A. and Merino, J. (1995), 'Dark leaf respiration in light and darkness of an evergreen and a deciduous plant species', *Plant Physiology* **107**, 421–427.
- Villar, R., Robleto, J. R., Jong, Y. D. and Poorter, H. (2006), 'Differences in construction costs and chemical composition between deciduous and evergreen woody species are small as compared to differences among families', *Plant, Cell and Environment* 29(8), 1629–1643.
- Walters, M. B., Kruger, E. L. and Reich, P. B. (1993), 'Growth, biomass distribution and co2 exchange of northern hardwood seedlings in high and low light, relationships with successional status and shade tolerance', *Oecologia* **94**(1), 7–16.

- Waring, R., Nordmeyer, A., Whitehead, D., Hunt, J., Newton, M., Thomas, C. and Irvine, J. (2008), 'Why is the productivity of douglas-fir higher in new zealand than in its native range in the pacific northwest, usa?', *Forest Ecology and Management* 255(12), 4040–4046.
- Watson, R. T., Noble, I. R., Bolin, B., Ravindranth, N. H., Verado, D., Dokken, D. J. and Eds (2000), 'Ipcc special report on land use, land-use change, and forestry', p. 377.
- Way, D. A. and Oren, R. (2010), 'Differential responses to changes in growth temperature between trees from different functional groups and biomes: a review and synthesis of data', *Tree Physiology* 30(6), 669–688.
  URL: http://treephys.oxfordjournals.org/content/30/6/669.abstract
- Weiner, J. (2004), 'Allocation, plasticity and allometry in plants', *Perspectives in Plant Ecology, Evolution and Systematics* **6**(4), 207–215.
- Welk, E., Katterfield, C. and Seidler, G. (2013), "'climatic modelling of distribution ranges of plant species", research run by the martin-luther-university hallewittenberg, scientific faculty: Dep. of biology, institute of geobotany and botanical garden: Natural distribution map for *Q. ilex.*', [online] url: www2.biologie.unihalle.de/bot/ag_chorologie/areale/IMPRESSUM.pp?sprache=E Accessed 3rd January 2013.
- West, C. and Gawith, M. (2005), Measuring progress: Preparing for climate change through the uk climate impacts programme, Technical report, UKCIP, Oxford.
- West, P. W. (2009), Tree and Forest Measurement, 2 edn, Springer, Berlin.
- Westphal, L. M. (2003), 'Social aspects of urban forestry: Urban greening and social benefits: a study of empowerment outcomes', *Journal of Arboriculture* **29**, 3.
- Whitehead, D., Griffin, K. L., Turnbull, M. H., Tissue, D. T., Engel, V. C., Brown, K. J., Schuster, W. S. F. and Walcroft, A. S. (2004), 'Response of total night-time respiration to differences in total daily photosynthesis for leaves in a *Quercus rubra* l. canopy: implications for modelling canopy *co*₂ exchange', *Global Change Biology* **10**(6), 925–938.
- Whitmarsh, J. and Govindjee (1999), *Concepts in Photobiology: Photosynthesis and Photomorphogenesis*, Kluwer Academic, chapter The Photosynthetic Process, pp. 11–51.
- WHO (2011), Air Quality Guidelines for Europe, WHO Regional Publications, European Series, No. 91, 2 edn, World Health Organisation, Copenhagen [online], chapter Chapter 11: Effects of nitrogen containing air pollutants: critical levels, pp. 230–233.

**URL:** *http://www.euro.who.int/en/what-we-publish/abstracts/air-quality-guidelines-for-europe* 

- Williams, R. A. and McClenahen, J. R. (1984), 'Biomass prediction equations for seedlings, sprouts, and saplings of ten central hardwood species', *Forest Science* **30**(2), 523–527.
- Williamson, G. B. and Wiemann, M. (2010), 'Measuring wood specific gravity', American Journal of Botany 97(3), 519–524.
- Willoughby, I., Stokes, V., Poole, J., White, J. E. J. and Hodge, S. J. (2007), 'The potential of 44 native and non-native tree species for woodland creation on a range of contrasting sites in lowland britain', *Forestry* 80(5), 531–553.
  URL: http://forestry.oxfordjournals.org/cgi/content/abstract/80/5/531
- Wood, N. L. H. (2004), 'Regional climate trends in south-west england and the north atlantic oscillation', *Weather* **59**(2), 39–41.
- Wright, I. J., Reich, P. B., Cornelissen, H. C., Falster, D. S., Groom, P. K., Hikosaka, K., Lee, W., Lusk, C. H., Niinemets, U., Oleksyn, J., Osada, N., poorter, H., Warton, D. I. and Westoby, M. (2005), 'Modulation of leaf economic traits and trait relationships by climate', *Global Ecology and Biogeography* 14, 411–421.
- Wright, I. J., Reich, P. B., Westoby, M., Ackerly, D. D., Baruch, Z., Bongers, F., Cavender-Bares, J., Chapin, T., Cornelissen, J. H. C., Diemer, M., Flexas, J., Garnier, E., Groom, P. K., Gulias, J., Hikosaka, K., Lamont, B. B., Lee, T., Lusk, C., Midgley, J. J., Navas, M.-L., Niinemets, U., Oleksyn, J., Osada, N., Poorter, H., Poot, P., Prior, L., Pyankov, V. I., Roumet, C., Thomas, S. C., Tjoelker, M. G., Veneklaas, E. J. and Villar, R. (2004), 'The worldwide leaf economics spectrum', *Nature* **428**, 821–827.
- Yunus, M., Pathre, U. and Mohanty, P. (2000), *Probing photosynthesis : mechanisms, regulation and adaptation*, Taylor and Francis, London.
- Zianis, D. and Mencuccini, M. (2004), 'On simplifying allometric analyses of forest biomass', *Forest Ecology and Management* 187(2-3), 311–332. doi: DOI: 10.1016/j.foreco.2003.07.007.
  - URL: http://www.sciencedirect.com/science/article/B6T6X-49S6WXT-1/2/1252c0b67941f7e9a53d7f9bafe33b99
- Zivcak, M., Brestic, M., Olsovska, K. and Slamka, P. (2008), 'Performance index as a sensitive indicator of water stress in *Triticum aestivum* 1.', *Plant Soil and Environment* 54, 133–139.